

UNLV Theses, Dissertations, Professional Papers, and Capstones

8-2011

Enhancing the cultural competence of women's health nurses via online continuing education

Ella T. Heitzler University of Nevada, Las Vegas

Follow this and additional works at: https://digitalscholarship.unlv.edu/thesesdissertations

Part of the Adult and Continuing Education and Teaching Commons, Curriculum and Instruction Commons, Curriculum and Social Inquiry Commons, Educational Assessment, Evaluation, and Research Commons, Instructional Media Design Commons, and the Maternal, Child Health and Neonatal Nursing Commons

Repository Citation

Heitzler, Ella T., "Enhancing the cultural competence of women's health nurses via online continuing education" (2011). *UNLV Theses, Dissertations, Professional Papers, and Capstones*. 913. https://digitalscholarship.unlv.edu/thesesdissertations/913

This Dissertation is protected by copyright and/or related rights. It has been brought to you by Digital Scholarship@UNLV with permission from the rights-holder(s). You are free to use this Dissertation in any way that is permitted by the copyright and related rights legislation that applies to your use. For other uses you need to obtain permission from the rights-holder(s) directly, unless additional rights are indicated by a Creative Commons license in the record and/or on the work itself.

This Dissertation has been accepted for inclusion in UNLV Theses, Dissertations, Professional Papers, and Capstones by an authorized administrator of Digital Scholarship@UNLV. For more information, please contact digitalscholarship@unlv.edu.

ENHANCING THE CULTURAL COMPETENCE OF WOMEN'S HEALTH NURSES

VIA ONLINE CONTINUING EDUCATION

By

Ella Thomas Heitzler

A dissertation submitted in partial fulfillment of the requirements for the

Doctor of Philosophy in Nursing

School of Nursing Division of Health Science The Graduate College

University of Nevada, Las Vegas August 2011 Copyright by Ella Thomas Heitzler, 2011 All Rights Reserved



THE GRADUATE COLLEGE

We recommend the dissertation prepared under our supervision by

Ella Thomas Heitzler

entitled

Enhancing the Cultural Competence of Women's Health Nurses via Online Continuing Education

be accepted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy in Nursing

School of Nursing

Lori Candela, Committee Chair

Cheryl Bowles, Committee Member

Patricia Alpert, Committee Member

LeAnn Putney, Graduate College Representative

Ronald Smith, Ph. D., Vice President for Research and Graduate Studies and Dean of the Graduate College

August 2011

ABSTRACT

Enhancing the Cultural Competence of Women's Health Nurses via Online Continuing Education

by

Ella Thomas Heitzler

Dr. Lori Candela, Examination Committee Chair Associate Professor of Nursing & Psychosocial Department Chair University of Nevada, Las Vegas

By 2050, current minority groups will comprise almost half of the US population further challenging healthcare providers and nurses to deliver culturally competent care. Numerous organizations have published documents supporting cultural competence and its incorporation into nursing curricula has been encouraged since 1986. However, practicing nurses, specifically those providing care to childbearing women and families, continue to acknowledge their lack of cultural competence. This is concerning as large health disparities exist between culturally diverse women and cultural competence can lead to greater health equality and better client care. Studies have shown face-to-face education increases the cultural competence of healthcare providers; few studies have explored the impact of online education on cultural competence levels and no studies of online socially interactive continuing education (CE) have been conducted with nurses. Therefore, the purpose of this study was to evaluate the effect of two different online CE interventions on the cultural competence level of nurses who care for childbearing women and newborns in the US. The study also explored: the relationship between social desirability and self-reported level of cultural competence, the relationship between level of educational attainment and cultural competence level, and the

iii

relationship between having previous cultural diversity training and level of cultural competence. The framework for the study included two existing models: the 3dimensional puzzle model of culturally congruent care and the instructional strategy framework for online learning environments. The study had an experimental pre- and post-test design using the Cultural Competence Assessment instrument. Three groups of RNs who care for childbearing women and newborns in the US were used: (a) control, (b) socially interactive online CE intervention, and (c) socially isolated online CE intervention. The study began with 249 registered participants, 190 completed the informed consent process, 132 completed the pre-test, and 93 completed the study. Data was analyzed with a RM-ANOVA with a between-groups variable, ANCOVA, and correlation statistics. Major findings indicated socially isolated online cultural competence CE is significantly more effective than not having online cultural competence CE. However, socially interactive online cultural competence CE is not more effective than socially isolated, nor is it significantly more effective than not having online cultural competence CE. Findings also indicated MCSDS scores and the number of previous types of cultural diversity training are positively correlated with CCA scores.

ACKNOWLEDGEMENTS

Numerous individuals have assisted me throughout my doctoral education and the dissertation process. Co-workers, friends, family, and faculty have given me strength, guidance, and support during some of the most challenging experiences of my life. At times the concurrent roles of student, faculty member, wife, and family member seemed to be too much. I am grateful that someone was always there when I needed help.

First and foremost I would like to thank my husband, Carter Paul Heitzler, III. Although he has likely sacrificed more than anyone else as I have undertaken my doctoral studies, he has always supported my unyielding need for furthering my education. Paul has consistently believed in me and tolerated my busy schedule, "work mode," and moodiness caused by stress. He has given up vacation time, holidays, and his own free time to be with me when needed and to keep our home up and running. I will never be able to thank him enough for being my friend, cheerleader, and partner.

I would also like to thank my family. Even my grandmother, who passed away many years ago, has contributed to my journey by encouraging me to get as much education as possible and by teaching me at a young age that anything worth doing is worth doing well. My parents, S. Glenn and Leta Thomas, have given me the gift of time by serving as exceptional editors and helping me in any way possible. My mother has been all a mother could be: friend, housecleaner, cook, and chauffeur; my father has served as a mechanic, handyman, and advisor. Although I have not been able to spend much time in the past few years with my brother, Stephen Thomas, he has shared his time during my life crises and afforded me time to do work in the car by going on several road

V

trips. I would also like to thank my in-laws for feeding our cat, Madison, during the numerous trips to campus.

Several of my co-workers have also played an integral role in my success. Each of them has provided me with emotional support and guidance. Dorothy Crowder has been a friend and mentor, providing words of wisdom, taking on additional responsibilities to free up some of my time, and even delaying her other life plans to see me through this program. Mary Hancock has helped me keep my sense of humor and balance, kept me connected to the outside world, helped me navigate through AWHONN, and took on additional workload when I needed it most. Finally, Dr. Barbara Laird has been flexible, understood my journey, and supported my educational and emotional needs.

Last, but certainly not least, I would like to thank the members of my dissertation committee: Drs. Lori Candela, Cheryl Bowles, Patricia Alpert, and LeAnn Putney. Gratitude is also owed to Dr. Chad Cross, previous UNLV Associate Professor of Biostatistics, for his assistance with methodology and the statistical analysis portion of this work. A very special thank you is owed to my chairperson, Dr. Candela. She has understood my needs as both a student and person since the very first time I spoke with her. Not only has she provided me with her feedback and time, but also with much needed words of encouragement. She has truly "had my back" throughout this process.

vi

| ABSTRACT | iii |
|---|------------|
| ACKNOWLEDGEMENTS | v |
| CHAPTER 1 INTRODUCTION | 1 |
| Background and Significance | 1 |
| Problem Statement | 5 |
| Purpose Statement | 7 |
| Summary | 7 |
| CHAPTER 2 REVIEW OF RELATED LITERATURE | 9 |
| Culture | 9 |
| Cultural Competence | 10 |
| Online Continuing Education | 42 |
| Summary | 64 |
| | |
| CHAPTER 3 THEORETICAL FRAMEWORKS | 66 |
| 3-Dimensional Puzzle Model of Culturally Congruent Care | 66 |
| Instructional Strategy Framework for Online Learning Environments | 70 |
| Conceptual Definitions | 74 |
| Operational Definitions | 74 |
| Summary | 75 |
| Hypotheses | 75 |
| Assumptions | 76 |
| | 77 |
| Design | <i>ו</i> ו |
| Design | |
| Ethical Considerations | |
| Drogodyrog and Data Collection | 80 |
| Procedures and Data Conection | |
| Data Analysis | 89 |
| CHAPTER 5 FINDINGS OF THE STUDY | 91 |
| Response and Attrition Rates | 91 |
| Demographic Data | 92 |
| Hypothesis One | 95 |
| Hypothesis Two | 97 |
| Hypothesis Three | |
| Attrition | |
| Summary | 102 |
| | |
| CHAPTER 6 SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS | 104 |
| Brief Overview of Study | 104 |
| Discussion of the Findings | 105 |
| Recommendations for Nursing and Nursing Education | 118 |

TABLE OF CONTENTS

| Study Lin Recomm Summary | nitations endations for Future Research | |
|--------------------------------|--|---------|
| APPENDIX A | E-MAIL TO AWHONN STATE LEADERS | 124 |
| APPENDIX B | FOLLOW-UP PHONE CALL TO AWHONN STATE LEAD | DERS125 |
| APPENDIX C | PDF AND JPEG INVITATION | 126 |
| APPENDIX D | INFORMED CONSENT FORM FOR GROUP 1 | 127 |
| APPENDIX E | INFORMED CONSENT FORM FOR GROUP 2 | |
| APPENDIX F | INFORMED CONSENT FORM FOR GROUP 3 | |
| APPENDIX G | ORIGINAL IRB APPROVAL | 136 |
| APPENDIX H | IRB MODIFICATION APPROVAL | 137 |
| APPENDIX I | RECOMMENDED ONLINE COURSE DESIGN PRINCIPL | ES138 |
| APPENDIX J | PERMISSION TO USE CCA | 140 |
| APPENDIX K | CCA, MCSDS, AND DEMOGRAPHIC ITEMS | 141 |
| APPENDIX L | COURSE EVALUATION ITEMS | 150 |
| APPENDIX M | TABLE OF COURSE EVALUATION FINDINGS BY CO | URSE151 |
| APPENDIX N | PERMISSION TO USE ISFOLE | 152 |
| APPENDIX O | PERMISSION TO USE 3-D MODEL FIGURE | |
| APPENDIX P | PERMISSION TO USE ISFOLE FIGURE | 154 |
| REFERENCES | | 158 |
| VITA | | 174 |

CHAPTER 1

INTRODUCTION

Cultural competence is of paramount importance as the United States (US) continues to become more culturally diverse. To utilize a common analogy, the US was once called a melting pot of different peoples who mixed together into one. Today, the US has been described as a stew or salad with differing ingredients that come together but keep their distinct flavors (Schim, Doorenbos, Benkert, & Miller, 2007). This diversity impacts healthcare providers and the client care they provide.

Numerous studies have focused on cultural competence in healthcare. However, many of these works have centered on promoting cultural competence in students or healthcare providers other than registered nurses (RNs). While a limited number of studies have been conducted with licensed nurses, there is no consensus regarding the best way to enhance the cultural competence of practicing nurses.

Background and Significance

By 2050, current minority groups (Hispanic, Asian, Pacific Islander, African American, and American Indian) will comprise approximately 48% of the US population (Gardner, 2010). In 2002 immigrant mothers accounted for 23% of all births, up from 15% in 1990 (Camarota, 2005), and in the past 20 years immigration has accounted for almost all of the enrollment increase within US public schools (Camarota, 2007). In fact, the US may be the most culturally diverse nation in the world (Cooper, Grywalski, Lamp, Newhouse, & Studlien, 2007).

This demographic shift challenges healthcare providers to deliver culturally competent care to improve health equality. However, equality of health is not currently a

reality. One major goal of the *Healthy People 2010* initiative was to elevate minority health into public view and eliminate health disparities (Office of Minority Health, 2008). One of the overarching goals of the *Healthy People 2020* framework continues to address healthy equality, the elimination of health disparities, and enhancement of health for all peoples (Office of Disease Prevention and Health Promotion, 2009). Furthermore, as part of the *Initiative to Eliminate Racial and Ethnic Disparities in Health*, six major areas of racial and ethnic disparity have been targeted by the US President because they impact many minority groups: cancer screening and management; cardiovascular disease; diabetes; HIV and AIDS; immunizations; and infant mortality (Office of Minority Health, 2008).

Large health disparities also exist between women of varying cultural, ethnic, and racial groups within the US (Callister, 2005; Sarto, 2005). For example, although the US maternal mortality rate has declined, maternal mortality among African American women is four times that of Caucasian women, comprising one of the biggest racial disparities in public health indicators (Sarto, 2005). Additionally, American Indians and Alaska Natives are twice as likely as Caucasian Americans to lack first trimester prenatal care (Agency for Healthcare Research and Quality [AHRQ], 2007). This is unfortunate, as the link between prenatal care and birth weight has been evident since 1977 (Showstack, Budetti, and Minkler, 1984) and early prenatal care can potentially decrease infant mortality rates, infant injury rates, and maternal postpartum depression (Alexander & Kotelchuck, 2001). Further minority disparities exist in the incidence of preterm birth (Moore, Moos, & Callister, 2010), infant mortality (Moore et al., 2010; Sarto, 2005), and very low birth-weight newborns (Callister, 2005).

Cultural competence can ultimately lead to greater health equality (Doorenbos, Schim, Benkert, & Borse, 2005; Edwards, 2003; Jackson, 2007; Lipson & Desantis, 2007; McHenry, 2007; Paez, Allen, Carson, & Cooper, 2008) and has numerous other benefits as well. Cultural competence increases client satisfaction (Bussema & Nemec, 2006), enhances communication between providers and clients (Kelly & Papadopoulos, 2009), and lessens societal economic strain (Schultz, 2004). Within women's health, cultural competence has been said to increase the incidence of breastfeeding (Hernandez, 2006; Riordan & Gill-Hopple, 2001), which could potentially decrease the incidence of childhood obesity in Hispanic Americans (Hernandez, 2006). Additionally, culturally sensitive healthcare providers increase the willingness of Hispanic women to access prenatal care (Shaffer, 2002) and culturally competent patient-centered care directly impacts the amount of prenatal care received by Hispanic women and their ability to learn at prenatal appointments (Tandon, Parillo, & Keefer, 2005). Finally, Latina women are more satisfied with providers who have undergone cultural competence training and are more likely to be compliant with recommended treatments (Castro & Ruiz, 2009). Clearly cultural competence impacts women's healthcare.

In recent years numerous groups and organizations have published standards, statements, or other documentation supporting the importance of cultural competence within healthcare. In 2001 the Office of Minority Health, a department of the US Department of Health and Human Services, published the national standards for culturally and linguistically appropriate services in health care, or the CLAS standards. The CLAS standards address the importance of cultural competence in healthcare and provide 14 standards of care for healthcare organizations and providers (Office of

Minority Health, 2001). Standard 1 states organizations should ensure all clients receive respectful care provided in a manner which is consistent with their cultural beliefs. Standard 3 then urges organizations to ensure all staff receives training regarding culturally appropriate service (Office of Minority Health, 2001). The Office of Minority Health has published a document entitled *Teaching Cultural Competence in Health Care:* A Review of Current Concepts, Policies and Practices prepared by the American Institutes for Research (2002). Additionally, the Liaison Committee on Medical Education, the National Committee for Quality Assurance, the Accreditation Council for Graduate Medical Education, and the Joint Commission on Accreditation of Healthcare Organizations (JCAHO) have all actively supported cultural competence in one or more ways (American Institutes for Research, 2002). Both the American Association of Colleges of Nursing (AACN) and the American Nurses Association (ANA) have encouraged the incorporation of cultural training into nursing curricula (AACN, 2008; Campinha-Bacote, 2006). The need to embrace cultural competence in healthcare and nursing is articulated across these organizations.

Although guidelines for incorporating cultural material into nursing programs have existed since 1986 (Campinha-Bacote, 2006), a meta-synthesis completed by Coffman (2004) reveals many practicing nurses perceive they do not have the needed training or skills to appropriately care for clients of various cultures and therefore express the desire for additional cultural training. Specifically, women's health nurses, or RNs caring for childbearing women and/or newborns in the US, acknowledge their inability to provide culturally competent care (Cooper et al., 2007). While the exact cultural competence level of women's health nurses is unknown, a study by Noble, Noble, Geiss,

and Hand (2006) found 77% of providers working with breastfeeding mothers in the US were not culturally competent. This is unfortunate as views of childbearing often depend on cultural beliefs related to health, healthcare, reproduction, and the role of women.

Many cultures have customs that guide behaviors during pregnancy, labor, delivery, and the postpartal period (Dean, 2010; Mattson, 2000). Specific customs, beliefs, and needs related to childbearing have been documented for: Native American women (Cesario, 2001; Galanti, 2008), Russian women (Callister et al., 2007), Asian women (Davis, 2001; Galanti, 2008), Hispanic women (Higgins & Learn, 1999; Galanti, 2008), Mexican women (Galanti, 2008; Johnson, 2005; Kim-Godwin, 2003), Middle Eastern women (Galanti, 2008; Kim-Godwin, 2003), Arab women (Bowers, n.d.; Galanti, 2008), African American women (Bowers, n.d.; Galanti, 2008), Orthodox Jewish women (Galanti, 2008; Semenic, Callister, & Feldman, 2004), Sephardic Jewish women (Galanti, 2008), and Muslim women (Bowers, n.d.; Galanti, 2008) residing in the US. Cultural groups have unique beliefs and behaviors related to childbearing; therefore, cultural competence is important to women's health nursing. In fact, the Association of Women's Health, Obstetric, and Neonatal Nurses (AWHONN) incorporates cultural competence into its vision, mission, and values, stating the association has a responsibility to increase the cultural competence knowledge and skills of nurses who serve childbearing women and newborns (AWHONN, n.d.).

Problem Statement

The diverse childbearing population of the US needs culturally competent nurses to provide them with adequate care and reduce health disparities. While nursing programs have been encouraged to incorporate transcultural training into their curricula,

practicing nurses, specifically women's health nurses, continue to acknowledge their lack of cultural competence. No specific work has measured or addressed the cultural competence of nurses practicing in the specialty area of women's health, but studies have shown face-to-face educational interventions increase the cultural competence of healthcare providers (Pugh, 2008). However, face-to-face continuing education (CE) is limited by time and space. Furthermore, one of the key research questions identified by the AHRQ, in their publication entitled *Setting the Agenda for Research on Cultural Competence in Healthcare*, asks which educational delivery methods most effectively increase cultural competence (Fortier & Bishop, 2004).

While two pieces of literature have explored the effect of socially interactive (interaction between individuals) online education on the cultural competence of nurses enrolled in graduate coursework (Hunter, 2008; Kelly & Papadopoulos, 2009), no studies of online socially interactive education have been conducted with nurses who are not enrolled in graduate study. Furthermore, no conclusive evidence supports the effectiveness of the two identified publicly accessible socially isolated (no interaction between individuals) CE modules (Gerace & Salimbene, n.d.; Office of Minority Health, 2007). While research supports the importance of social interaction in online education (Gallien & Oomen-Early, 2008; Mancuso-Murphy, 2007; Rovai, 2002; Zhao, Lei, Lai, & Tan, 2005), no studies have actually compared the effectiveness of socially isolated online CE to socially interactive online CE. As online CE courses are growing in popularity (Cobb, 2004), further studies are needed to evaluate the impact of both socially isolated and socially interactive online cultural competence CE on the cultural competence level of nurses, particularly those specializing in women's health.

Purpose Statement

The purpose of this study was to evaluate the effect of two different online CE interventions on the cultural competence level of nurses who care for childbearing women and newborns in the US. A quantitative methodology was used to compare change in cultural competence level after completing a socially isolated online cultural competence CE course to the change in cultural competence level following completion of a socially interactive online cultural competence CE course. The study also explored: the relationship between social desirability and self-reported level of cultural competence, the relationship between level of educational attainment and cultural competence level, and the relationship between having previous cultural diversity training and level of cultural competence.

Summary and Organization of Remaining Chapters

This introductory chapter presents basic study details. The background and significance, problem, and purpose are detailed. The next chapter (Chapter 2) provides a comprehensive review of literature related to cultural competence, its measurement, previously utilized interventions designed to increase cultural competence, and research which has been conducted with practicing healthcare professionals. Chapter 2 also details existing literature related to online teaching and learning theories, best practices in online education, and previous research of online CE. Chapter 3 describes the two theoretical frameworks employed in greater detail, provides conceptual and operational definitions, and lists the hypotheses and assumptions. Chapter 4 outlines study methodology, Chapter 5 presents study findings, and the sixth and final chapter expounds

upon study findings, discusses their implications, and sets forth recommendations for nursing, nursing education, and further research.

CHAPTER 2

REVIEW OF RELATED LITERATURE

This chapter provides a comprehensive review of the literature related to cultural competence and online education. Culture will be defined and cultural competence models discussed. Instruments measuring cultural competence will be detailed as well as studies of face-to-face and online education designed to enhance the cultural competence of healthcare providers. The second half of the chapter reviews online teaching and learning, detailing theories/frameworks, pros and cons, the importance of social interaction, course evaluation, and best practices for designing online courses.

Culture

The term culture must be defined before cultural competence can be discussed. Many definitions of culture are found in the literature. Leininger (1997) defines culture as the lifeways of a person or group of peoples which relate to beliefs, values, patterns, and practices that are shared, learned, and passed from one generation to the next. Schim et al. (2007) define culture as a vast scope of social behaviors, attitudes, values, and shared symbols most individuals take for granted as the background of their life. Culture permeates everyday existence and impacts extraordinary moments. In short, culture can be described as the context in which one lives (S. M. Schim, personal communication, May, 2008).

Further exploration reveals culture is a dynamic concept which is learned, shared, and evolves in response to political realities and environmental, social, and economic changes (Schim et al., 2007). In its broadest definition, culture is not simply race or ethnicity; rather, culture encompasses a wide range of similarities and differences

between people and communities (Schim et al., 2007). In fact, people are an amalgamation of a complex set of cultures. Purnell (2002) lists nationality, color, gender, age, and religious affiliation as primary cultural characteristics and educational level, socioeconomic status, political beliefs, occupation, location of residence, marital status, physical characteristics, immigration status, and sexual orientation as secondary cultural characteristics. Schim et al. (2007) include subcultures, a concept similar to Purnell's secondary characteristics. Subcultures include communities of common interest (cancer survivors, Harley-Davidson motorcycle owners, learning disabled or similar, etc.) and communities with common needs (the homeless, mentally ill, diabetics, etc.). Culture may also include similarities and differences between socioeconomic classes, those with differing sexual orientations, or those of different genders (Schim, Doorenbos, & Borse, 2006a). In short, one's culture is the background to one's life.

Cultural Competence

Models

Many models, theories, or frameworks related to culture and cultural competence have been utilized in healthcare. Madeline Leininger is credited with developing the first of such theories in the 1970's (Welch, 2002). Numerous other models, theories, or frameworks have followed. Several commonly utilized and cited models or theories will be discussed below.

The theory of care diversity and universality.

Madeline Leininger is the founder of transcultural nursing. Her theory, the theory of cultural care diversity and universality, was derived from both nursing and anthropology (Welch, 2002). Leininger first related care to culture in the 1950's after

realizing these two significant concepts in nursing had never been linked (Leininger, 1997). The major premise of Leininger's theory is differences (diversities) and similarities (universalities or commonalities) exist which provide a body of transcultural nursing knowledge (Leininger, 1997). Leininger believes humanistic caring needs a broad cultural theoretical perspective to guide nursing and promote positive outcomes. The purpose of her theory is to find, document, decipher, and explain the predicted factors which influence and explain care as a cultural holistic practice. The theory's goal is to provide culturally congruent care to contribute to health (Leininger, 1997).

Culturally congruent nursing care involves taking appropriate actions which are related to culture care (culturally derived and supportive acts for another person), thus accommodating and restructuring care to meet client needs (Leininger, 1997). The sunrise model, which depicts the theory of cultural care diversity and universality, illustrates these concepts, showing culturally congruent care, cultural care maintenance, nursing actions, folk systems, professional healthcare systems, clients, holistic wellbeing, and numerous cultural factors are all interconnected. In short, nurses who value and practice culturally congruent care can effect positive healthcare changes for clients of various cultures (Leininger & McFarland, 2006).

The Giger and Davidhizar transcultural assessment model.

In 1988 Giger and Davidhizar developed a model to assist nursing students in providing care for culturally diverse clients (Giger & Davidhizar, 2002). The metaparadigm upon which their model is built includes: (a) culturally diverse and transcultural nursing, (b) culturally unique clients and providers, (c) culturally competent

nursing care, (d) environments which are culturally sensitive, and (e) health as defined by culturally specific behaviors (Giger & Davidhizar, 2008).

Transcultural nursing involves culturally competent nursing care centered on the client. Culturally diverse nursing care references the variance in nursing approaches which are needed to provide culturally competent care. Specifically, culturally diverse nursing care accounts for the six cultural phenomena which are evident within all cultures: communication, space, time, social organization, biological variations, and environmental control (Giger & Davidhizar, 2008). Cultural competence is a fluid and dynamic process in which meaningful and practical care-delivery strategies are activated based upon the cultural beliefs and behaviors of those receiving the care. Additionally, cultural competence implies a sophisticated high level of cognitive, psychomotor skills, and attitudes. Hence, cultural competence allows nurses to devise appropriate interventions, promoting optimal health regardless of culture (Giger & Davidhizar, 2008).

Giger and Davidhizar (2008) state all individuals, including clients and nurses are unique individuals, so nurses must avoid projecting their beliefs onto their clients. Diversity exists within and across cultural groups, so knowledge of general concepts relevant to cultural groups is a starting point for providing culturally competent care. Finally, culturally sensitive environments allow nurses to plan and use culturally relevant treatments for all. Culturally sensitive environments assess all six cultural phenomena when working with culturally diverse clients. Therefore, the Giger and Davidhizar transcultural assessment model was developed to allow for such assessment (Giger & Davidhizar, 2008).

Purnell model for cultural competence.

The Purnell model for cultural competence was first developed to help nurses and nursing students organize cultural assessment (Purnell, 2000). According to Purnell (2008) it now has many other purposes: (a) serve as a framework for all healthcare professionals to learn concepts of culture, (b) describe circumstances which affect cultural worldviews, (c) link central cultural relationships within one model, (d) facilitate the delivery of culturally competent healthcare, (e) provide a structure for the analysis of cultural data, and (f) assist in viewing individuals, families, or groups within their unique cultural environments. However, Purnell's model itself was designed for assessment, focusing mainly on 12 domains to include in cultural assessment: heritage, communication, family roles, workforce issues, biocultural ecology, high-risk behaviors, nutrition, pregnancy and childbearing, death rituals, spirituality, healthcare practices, and healthcare practitioner concepts (Purnell, 2000). The model does not actually guide cultural competence development because assessment is only one part of cultural competence; cultural competence is defined by Purnell (2000) as the adaptation of care in a manner consistent with the client's culture. Unfortunately, the model itself does not address interventions or adaptation of care.

Papadopoulos, Tilki and Taylor model for developing cultural competence.

The Papadopoulos, Tilki, and Taylor model was developed for use with students (Papadopoulos, Tilki, & Lees, 2004). According to Papadopoulos, Tilki, and Ayling (2008), cultural competence is the process in which one continuously develops and refines one's ability to provide effective healthcare taking into account people's cultural

beliefs, needs, and behaviors. The process involves blending cultural awareness, knowledge, and sensitivity and applying them to practice (Papadopoulos et al., 2008).

Cultural awareness, the first stage, focuses on self. It includes being self-aware, defining one's own culture, adhering to individual heritage, and decreasing ethnocentricity. The second stage, cultural knowledge, involves avoidance of stereotyping, having basic knowledge of health beliefs and behaviors of diverse groups, being knowledgeable of similarities and differences between groups, and understanding ethnohistory, psychological, sociological, and biological concepts. The third stage is cultural sensitivity, which involves empathy, communication skills, acceptance, trust, and respect. Finally, cultural competence is the synthesis and application of the first three stages (Papadopoulos et al., 2004). Cultural competence is reflected in assessment, diagnostic, and clinical skills (Papadopoulos et al., 2008).

The process of cultural competence in the delivery of healthcare services.

Campinha-Bacote's model, the process of cultural competence in the delivery of healthcare services, includes five constructs: cultural awareness, knowledge, skill, encounters, and desire. Campinha-Bacote (1999) defines cultural awareness as an intentional cognitive process in which providers appreciate and gain sensitivity to the values, beliefs, and practices of diverse cultures; cultural knowledge as an educational foundation of various world views which includes biocultural ecology and ethnic pharmacology; cultural skill as the ability to collect cultural information regarding health and performing a culturally specific physical assessment; cultural encounters as the way in which healthcare providers directly engage in cross-cultural encounters; and finally, cultural desire as the motivation to want to engage in the process of cultural competence.

Campinha-Bacote (2007) asserts the five constructs are related in an interdependent manner. Hence, providers must experience or address each of the constructs during their quests for cultural competence. Cultural competence therefore involves integrating cultural desire, awareness, knowledge, skills, and encounters (Campinha-Bacote, 2007). Campinha-Bacote (1999) defines cultural competence as the process by which healthcare providers continuously strive to achieve the ability to work effectively within clients' cultural contexts. One concern regarding the model is the cultural knowledge construct, which focuses on learning facts about various cultural groups, could lead to client stereotyping.

3-Dimensional puzzle model of culturally congruent care.

The 3-dimensional puzzle model of culturally congruent care, henceforth referred to as the 3-D model, has evolved from the Schim and Miller Cultural Competence Model which was developed in 1999 (Schim, Doorenbos, Miller, & Benkert, 2003). The 3-D model includes both a provider and client level which come together to form culturally congruent care (Schim et al., 2007). At this time the client level constructs have not been defined and therefore will not be discussed. Constructs of the provider level include the cultural competencies of cultural diversity, awareness, sensitivity, and competence behaviors. In short, provider cultural competence is having and using the cognitive, affective, and psychomotor skills necessary for facilitating culturally congruent care (Schim et al., 2007). This model will be further described in Chapter 3.

Cultural Instruments

A multitude of instruments which assess cultural competence have been developed and utilized. Numerous ones are specific to populations (such as nurses,

students, mental health professionals) or specialty areas of healthcare. Many cultural competence assessment instruments are described below.

Multicultural Sensitivity Scale.

The Multicultural Sensitivity Scale (MSS) was derived from a 52 item 5-point Likert-type scale which was developed by Ford in 1979 for use with teachers (Jibaja-Rusth, Kingery, Holocomb, Buckner, & Pruitt, 1994). Jibaja-Rusth et al. (1994) modified the scale for use with teachers of health professions. Face validity was confirmed by six experts and factor analysis with varimax rotation was used with a sample of 150 school nurses. Twenty-one items which loaded at a minimum of .44 on the factors were pooled to form the MSS which was then administered to 31 secondary health educators. Test-retest reliability was assessed and the internal consistency reliability (Cronbach's alpha) was calculated to be .90 (Jibaja-Rusth et al., 1994).

Multicultural Counseling Awareness Scale.

The Multicultural Counseling Awareness Scale (MCAS) was developed by Ponterotto and Casas in 1991 for use with counselors and students (Ponterotto, Gretchen, Utsey, Rieger, & Austin, 2002). The original 135 items were reduced to 70 items following focus group discussions and assessment of content validity. The remaining 70 items used a 7-point Likert-type scale and were piloted on 126 counseling professionals and students. Factor analysis was used and the scale reduced to its current form, which includes two subscales and a total of 45 items. The two subscales, knowledge/skills and awareness, have yielded test-retest reliability of .70 and .73, respectively, and coefficient alpha values of .78 to .93 and .67 to .83, respectively for the two subscales (Ponterotto et al., 2002).

Multicultural Case Conceptualization Ability.

The Multicultural Case Conceptualization Ability (MCCA) task was developed by Ladany, Inman, Constantine, and Hofheinz (1997) for use with healthcare providers. The MCCA utilizes a coding system to evaluate the extent to which racial factors are integrated into the conceptualization of the concerns with which clients present (Ladany et al., 1997). Specifically, the two conceptualizations evaluated include etiology of a presenting problem and treatment of the presenting problem. Scores range from zero to five and higher scores indicate greater differentiation (ability to form alternative interpretations of a client's problems) and integration (ability to form connections among different interpretations of problems). Only items found to have interrater agreement of at least 85% were included in the final tool (Ladany et al., 1997).

Cross-Cultural Adaptability Inventory.

The Cross-Cultural Adaptability Inventory (CCAI), a self-report inventory, was developed by Kelley and Meyers in 1995 for use in conjunction with training and research about individuals' ability to adapt when living in or traveling to other cultures (Kelley & Meyers, 2010). The CCAI responds to practical concerns expressed by culturally diverse individuals and the professionals who work with them (Kelley & Meyers, 2010). In fact, instrument development included both polling experts and a review of research (Capell, Veenstra, & Dean, 2007). These expert opinions guided identification of the most commonly cited skills and traits associated with cultural adaptability and served as a measure of construct validity (Capell et al., 2007).

The CCAI assesses four areas which are considered critical for effective crosscultural interactions: emotional resilience, flexibility/openness, perceptual acuity, and

personal autonomy (Kelley & Meyers, 2010). It includes 50 items and has yielded reliabilities ranging from .68 to .9 in early studies (Davis & Finney, 2006). Factor analysis with principle components analysis was conducted, but several items had low correlations with the components they were said to represent or were found to be highly correlated with others (Davis & Finney, 2006). Furthermore, during development, items designed for one subscale were reassigned to other subscales with little support for the change (Davis & Finney, 2006). In 2006 Davis and Finney conducted confirmatory factor analysis with a sample of over 700 college students and found the four-factor structure was a poor fit as high correlations were found between factors (Davis & Finney, 2006).

Cultural Competence Assessment- Khanna, Cheyney, and Engle.

Khanna, Cheyney, and Engle (2009) developed an instrument for use in their study of healthcare professionals. While they acknowledged the name has been commonly used in cultural competence education, they named their instrument the Cultural Competence Assessment. Their instrument included 29 items designed to reflect the concepts covered in their training program, which was guided by the Institute of Medicine (IOM) and CLAS standards. The tool included three parts: demographic information, knowledge-related statements, and statements related to skills (Khanna et al., 2009). Reliability was not discussed, nor was further information presented regarding the instrument's development.

Cultural Attitude Scale.

The Cultural Attitude Scale (CAS) was developed by Bonaparte (1979) to measure nurses' attitudes toward culturally different patients. The CAS is a 34 item scale

which includes vignettes with four different cultural groups: Jewish, Hispanic, African American, and Caucasian. Each of the four vignettes is followed by multiple items which include statements drawn from literature and personal clinical experiences (Bonaparte, 1979). Face validity was confirmed by three nurses and principle factor analysis with varimax rotation was conducted on a sample of 300 nurses. Three factors were discovered: nursing care-patient interaction, cultural attitudes and beliefs, and cultural health attitudes and beliefs. All item statements loaded at a minimum value of .4 on one of the three factors. Internal consistency reliability for the three subscales was found to be .94, .95, and .77 respectively; the entire scale's reliability was not reported (Bonaparte, 1979).

Cultural Fitness Survey.

The Cultural Fitness Survey (CFS), developed by Rooda (1993), includes three sections: the Ethnic Attitude Scale (EAS), developed by Rooda; the CAS, developed by Bonaparte and discussed above; and eight demographic questions. The EAS addresses knowledge of diversity and includes questions related to culturally specific diseases and disorders, values, and family (Rooda, 1993). Three specific minority groups are addressed with approximately an equal number of items: African Americans, Hispanics, and Asian Americans. Three multicultural expert nurse educators reviewed the instrument for clarity, level of knowledge, and content appropriateness. Twenty-two items remained following this review and were pilot-tested on 32 registered nurses. Reliability was confirmed (KR20 = .71) and the EAS was combined with the CAS and demographic questions to form the CFS for use with nurses (Rooda, 1993).

Transcultural Self-Efficacy Tool.

The Transcultural Self-Efficacy Tool (TSET) was designed to measure students' perception of their self-efficacy for performing transcultural nursing skills (Jeffreys & Smodlaka, 1998). Items were developed based upon literature review to be specific to transcultural nursing and appropriate for pre-licensure nursing students. Content validity was confirmed via expert review by six certified transcultural nurses; 13 items were deleted and numerous others changed based upon the review. Three subscales, cognitive, practical, and affective emerged. The tool was piloted on a sample of 357 nursing students. Split-half reliability was found to be between .7 and .93 for the entire TSET and each of its subscales. Internal consistency reliability was .97 for the entire scale and between .9 and .98 for the three subscales. Finally, test-retest reliability was confirmed with coefficients between .63 and .84 for the subscales (Jeffreys & Smodlaka, 1998).

Reliability studies were repeated with a sample of 1,260 nursing students. The split-half reliability was found to be .86 for the entire instrument and between .85 and .92 for its subscales while internal consistency was .98 for the entire TSET, .96 for the cognitive subscale, .97 for the practical subscale, and .96 for the affective subscale. Factor analysis was conducted via principle components analysis with a varimax rotation. A nine factor solution emerged with 70 items which loaded at a minimum value of .5 and explained 62% of the total variance. Internal consistency reliability of these nine factors ranged from .87 to .95 (Jeffreys & Smodlaka, 1998).

Cultural Self-Efficacy Scale.

The Cultural Self-Efficacy Scale (CSES) was developed by Bernal and Froman in 1987 to evaluate nurses' self-efficacy of caring for diverse cultural groups (Bernal &

Froman, 1993). Items were developed based upon transcultural nursing and anthropological literature (Bernal & Froman, 1987). Expert review was conducted by five public health nurses and a convenience sample of 190 was utilized to evaluate the original scale's 30 items (Bernal & Froman, 1987). Internal consistency reliability of the full original scale was found to be .97 (Bernal & Froman, 1987). In 1993 items were added to the scale to include 16 behavioral statements related to each of three cultural groups (African Americans, Latinos, and Southeast Asians) and 10 items to assess selfefficacy in all three groups, totaling 58 self-reported ratings. Principle factor analysis was conducted with a sample of 206 community health nurses and revealed four factors using a .4 factor loading value requirement. The four factors, general cultural skills, Black self-efficacy, Latino self-efficacy, and Southeast Asian self-efficacy, accounted for 90% of the total variance. Internal consistency reliability for the entire CSES was found to be .97 with values of .89, .99, .98, and .98 for the factors respectively (Bernal & Froman, 1993).

Lee Cultural Sensitivity Tool: Hispanic Version.

The Hispanic version of the Lee Cultural Sensitivity Tool was developed by Lee, Anderson, and Hill (2006) to measure nurses' knowledge of Hispanic health practices and beliefs. The tool contains 10 items with three answering options: agree, disagree, or do not know/no opinion. Scores range from zero to 10. While Lee et al. state content validity was established via reviewing literature related to Hispanic health beliefs and practices, no specific works are cited. The tool was piloted on a sample of seven nurses and its Cronbach's alpha was found to be .6 (Lee et al., 2006).

Cultural Awareness Scale.

The Cultural Awareness Scale (CAS) was developed by Rew, Becker, Cookston, Khosropour, and Martinez (2003) for use with nursing faculty and students. Thirty-seven items with a 7-point Likert-type scale were developed for five categories. The categories were identified via literature review to illustrate cultural awareness' multidimensional nature. The original pilot test of 72 nursing students and an unknown number of faculty yielded a CAS internal consistency reliability of .91 for the students and .82 for the faculty. Internal consistency reliability for the subscales ranged from .66 to .88 for the student subscales and from .56 to .87 for the faculty subscales (Rew et al., 2003). Expert review was then conducted by 10 nursing faculty and the context validity index was calculated to be .88; one item was removed and a few others reworded. The CAS was then completed by 118 additional students. Data from all samples were combined and factor analysis, using principle components analysis with varimax rotation, was completed using a .3 loading criteria. Five factors emerged: general educational experience, cognitive awareness, research issues, behaviors/comfort with interactions, and care/clinical issues. Internal consistency was found to be .82 for the entire scale and .85, .79, .94, .71, and .77, respectively, for the five subscales (Rew et al., 2003).

Multicultural Awareness- Knowledge-and Skills Survey.

The Multicultural Awareness- Knowledge- and Skills Survey (MAKSS) was developed by D'Andrea, Daniels, and Heck in 1990 for use with graduate level counseling students (D'Andrea, Daniels, & Heck, 1991). The survey was designed to obtain individuals' perceptions of their own multicultural counseling, knowledge, skills, and awareness level following multicultural counseling training. By matching survey

items to the training objectives for which they were designed to evaluate, content validity was established. The MAKSS was also compared to another instrument, the Multicultural Counseling Awareness Scale, to establish criterion related validity. Finally, factor analysis with principle axis factoring was completed using a loading value criterion of .3. Three factors emerged: awareness, knowledge, and skills. While not given for the full instrument, the internal consistency reliability of the three subscales was found to be .75, .9, and .96 respectively. The MAKSS includes 60 multiple choice Likert-type items and takes 20 to 25 minutes to complete (D'Andrea et al., 1991).

Cultural Competence Assessment Tool.

The Cultural Competence Assessment Tool (CCATool) is a self-assessment tool grounded in the Papadopoulos, Tilki, and Taylor model for developing cultural competence (Papadopoulos et al., 2004). The tool has four sections which have an equal number of items: cultural awareness, knowledge, sensitivity, and competent practice. The items use both a disagree/agree format and a visual analog scale from one to 10, with one indicating no awareness, knowledge, sensitivity, or competence behaviors, and 10 indicating a high level for each of these. Expert review and a pilot of both students and mental health professionals were included in its development. Test retest reliability and internal consistency reliability were assessed but not reported (Papadopoulos et al., 2004).

Multicultural Counseling Inventory.

The Multicultural Counseling Inventory (MCI) is a self-report inventory developed by Sodowsky, Taffe, Gutkin, and Wise (1994). It is based upon the crosscultural counseling competencies developed by Sue, Bernier, Durran, Feinberg, Pedersen,

Smith, and Vasques-Nuttall (as cited in Sodowsky et al., 1994). The instrument was developed to measure counselors' competencies when working with culturally diverse or minority clients. The initial pool consisted of 87 items on a 4-point Likert-type scale which were tested with 604 counselors and psychologists. Factor analysis yielded a four-factor solution which accounted for 36% of the variance. The four factors were named multicultural counseling skills, multicultural awareness, multicultural counseling relationship, and multicultural counseling knowledge respectively. Forty items were kept which had satisfactory loading values on the four factors. Internal consistency reliability was .83, .83, .65, and .79 for the four respective subscales and .88 for the full scale. Confirmatory factor analysis with a sample of 300 counselors supported the four factors and all 40 items had factor loadings of at least .30. Internal consistency reliability with this sample was .81, .80, .67, .80 for the four respective subscales and .86 for the full scale (Sodowsky et al., 1994).

Unnamed Instrument.

Polacek and Martinez (2009) developed an unnamed instrument. The instrument was designed to collect information from employees about the cultural knowledge, beliefs, attitudes, and skills of themselves and the organizations for which they worked. The instrument contains 137 items based on a detailed literature review of factors which reflect cultural awareness. The items address three areas: cultural knowledge, cultural perceptions/perspectives, and the organization's cultural competence. One hundred and fifty-six employees, including managers, clinical staff, clerical staff, physicians, and tech support personnel completed the instrument. Factor analysis was then employed with a .45 loading inclusion criteria. Five factors were extracted: language skills,

communication skills, knowledge, awareness, and relationships (Polacek & Martinez, 2009). Internal consistency reliability was not reported.

Inventory for Assessing the Process of Cultural Competence Among Healthcare Professionals- Revised.

The Inventory for Assessing the Process of Cultural Competence Among Healthcare Professionals- Revised (IAPCC-R) is a self-report instrument developed by Campinha-Bacote to specifically fit her model of cultural competence. It includes 25 items, using a self-administered 4-point Likert scale (Campinha-Bacote, 2003). Five different response sets are utilized: strongly disagree to strongly agree, not knowledgeable to very knowledgeable, not aware to very aware, not involved to very involved, and not comfortable to very comfortable. Total scores range from 25-100, indicating culturally incompetent (score of 25-50), culturally aware (score of 51-74), culturally competent (score of 75-90), or culturally proficient (score of 91-100). Content validity was established by transcultural healthcare experts and sixteen studies have found an average internal consistency reliability of .83 (Transcultural CARE Associates, 2007). There are two major criticisms for this instrument. First, Campinha-Bacote (2003) states higher scores suggest greater cultural competence, yet the highest score range aligns with what she terms cultural proficiency. Furthermore, the five response sets require rapid shifts between varying levels which likely leads to undue confusion.

Cultural Competence Assessment- Schim, Doorenbos, Miller, and Benkert.

The Cultural Competence Assessment (CCA) is a 25 item self-report questionnaire with two subscales: the eight item cultural awareness and sensitivity (CAS) subscale and the 17 item cultural competence behaviors (CCB) subscale. The CCA was

originally developed by Schim et al. (2003) based upon the constructs which comprise the provider level of the current 3-D model. The instrument uses a 7-point Likert-type scale, is written at a fifth grade reading level (Schim et al., 2003), and takes about 20-30 minutes to complete (Doorenbos et al., 2005). Total scores range from 25 to 175 and average scores from one to seven; higher scores indicate greater cultural competence. Instrument development included expert review, field testing, a pilot test, and factor analysis with principle axis factoring. A two-factor solution was discovered and 25 items loaded between .42 and .77 on the cultural competence behaviors (CCB) and the cultural awareness and sensitivity (CAS) factors. CCA internal consistency reliability was .92, and was .93 and .75 for the CCB and CAS respectively. Criterion related validity was established by comparison to the IAPCC (Schim et al., 2003). Further studies have yielded CCA reliabilities of .91 and .89 (Doorenbos & Schim, 2004; Doorenbos et al., 2005), CCB reliability of .91, and CAS reliability of .75 (Doorenbos et al., 2005).

A short form of the Marlow-Crowne Social Desirability Scale (MCSDS) was added to the CCA after its initial development in response to criticism that self-report measures are not reliable due to social desirability. The 13 item form has a reliability of .76 (Reynolds, 1982). MCSDS scores have not been shown to correlate with CCA scores (S. M. Schim, personal communication, July 10, 2010).

Summary.

A plethora of instruments exist for measuring cultural competence or related concepts. However, as noted above, many only address one component of cultural competence such as cultural sensitivity or cultural awareness. Other instruments focus on a limited number of cultural groups or were designed for use with one specific
population. Furthermore, many instruments have limitations related to scoring, reliability, or validity. Finally, only three instruments reviewed were grounded in a cultural competence theory.

Research of Cultural Competence Training for Healthcare Professionals

Numerous studies have explored the effect of face-to-face, otherwise known as traditional, education on cultural competence levels. Few studies have explored online cultural competence educational interventions. Face-to-face and online cultural competence educational interventions for healthcare providers will be described.

Face-to-face educational interventions.

Pugh.

Pugh (2008) conducted a meta-analysis of educational delivery methods used in healthcare provider cultural competence education. Thirty studies of face-to-face educational interventions with varying lengths, study designs, and participants were included in the analysis. Specifically, studies in the meta-analysis had training lengths between one hour and four years in duration and included doctors, therapists, social workers, nurses, healthcare educators, and healthcare students. The meta-analysis revealed an overall effect size of .46 with a 95% confidence interval, indicating various educational methods moderately increase healthcare provider self-reported cultural competence level. Explicitly, the 12 studies with the greatest effect sizes utilized lecture, active learning, independent or self-directed learning, problem-based learning, and demonstration (Pugh, 2008). Limitations noted by Pugh include the various tools used to measure cultural competence in the studies and the inclusion of only English literature about studies conducted in the US.

Majumdar, Browne, Roberts, and Carpio.

Majumdar, Browne, Roberts, and Carpio (2004) conducted a study to determine the effect of cultural sensitivity training on healthcare provider knowledge and to ascertain the impact trained providers have on the health and satisfaction outcomes of minority clients. A randomized controlled trial which originally consisted of 114 nursing and home health providers and 133 patients was utilized. The experimental group of providers underwent 36 hours of cultural sensitivity training and were assessed with the Self-Assessment of Cultural Awareness questionnaire and the Rokeach Dogmatism Scale. Instruments utilized by or on the patients in the study included: the Expenditures of Health Care and Social Services, the Client Satisfaction Questionnaire, Client Health Outcomes, and the Physical and Mental Health Assessment Questionnaire. Following the training there was a significant increase in provider understanding of multiculturalism, cultural awareness, understanding of cultural differences, cultural beliefs, adopting health care literature, considering social circumstances, and considering culture to be important; no differences were found in the dogmatism scale. No statistically significant differences were noted between the control and experimental patient groups in client satisfaction, mental health, physical health, or activities of daily living. However, social resources and economic resources did have significant patient differences over time. In fact, the average total social and health expenditures dropped from \$18,000 at baseline to \$11,000 one and a half years after the provider training and study commencement (Majumdar et al., 2004).

The study, while innovative, had several limitations. Majumdar et al. (2004) cite difficulty maintaining the patient population due to death and illness, the potential for

strong emotional bonds between providers and patients, and loss of providers due to moving or termination as potential limitations. In fact, the 12 month study had 44% patient and 33% provider attrition. Additionally, the study was limited by its sample of providers and patients from two home healthcare agencies and one hospital and potential bias based upon financially compensating study participants. Furthermore, Majumdar et al. do not discuss educational or cultural theories. It is therefore assumed they did not use either to guide the training sessions.

Papadopoulos, Tilki, and Lees.

Papadopoulos et al. (2004) conducted a study of 35 mental health staff members using a pre- and post-test design. The educational intervention included eight sessions, two dedicated to each of the four areas of the Papadopoulos, Tilki, and Taylor model for developing cultural competence, offered over four months. The CCATool was used to measure cultural competence before and after the intervention. Before the intervention 24 staff were found to be culturally aware, 10 culturally safe, and one culturally competent. Eighteen staff completed the CCATool after the intervention; most remained culturally aware, four increased to culturally safe, and two moved down to culturally aware (Papadopoulos et al., 2004).

Papadopoulos et al. (2004) note the low CCATool response rate following the intervention limits the study findings. They also note conflicting views, prejudice, stereotypes, racism, and difficulty deciding what to cover in the sessions as being challenges. Other limitations include using a convenience sample, not utilizing a comparison group, and not using an educational theory to guide development of the intervention.

Williams.

In 2002 Williams conducted a study to evaluate the effectiveness of an educational program which was designed to enhance the cultural competence level of practicing social workers. The quasi-experimental mixed-methods study included a convenience sample of 48 addiction and mental health care social workers. While a comparison group was used, randomization was not.

The educational intervention utilized adult educational principles and included four three-hour sessions over consecutive weeks which were guided by review of the literature and four social work textbooks. The intervention focused on four assumptions of cultural competence in social work: (a) helping relationships are affected by the culture of the client and the practitioner; (b) formal social work education forms a foundation for culturally competent practice, but does not fully develop the skills needed for all cross-cultural situations; (c) cultural competence ideas and concepts are in development and discussion is needed to form guidelines for practice; and (d) race and ethnicity are crucial when discussing mental health cultural competence, but other identity issues should also be included (Williams, 2002). In general, the course provided opportunities to enhance skills relevant to interacting with culturally diverse individuals.

The MCI and the MCCA task were used to collect data about all participants before and after the time during which the educational intervention was offered. Participants in the intervention group had a superior gain in ability to integrate cultural concepts into case vignette responses/interventions. Significant differences were found only in the awareness subscale of the MCI. Additionally, eight participants completed a follow-up interview regarding their impression of the course and how its content had

been applied within their own practice. Qualitative analysis revealed four themes: shift in awareness, practices attributed to training, anticipated practices attributed to training, and development of culturally competent practice knowledge (Williams, 2002).

Williams (2002) cites the study was limited by its convenience sample which limits generalizability. An additional limitation is the lack of true cultural competence theoretical foundation for course development.

Rooda and Gay.

Rooda and Gay (1993) conducted a study of a workshop designed to enhance incorporation of cultural sensitivity into nursing practice and care. All 28 directors of nursing from two hospitals attended the seven hour, one day workshop. The workshop addressed the impact of culture on the healthcare practices of specific ethnic groups (limited to Asian, African, and Mexican-Americans) and on developing a learning sequence which could be utilized with staff. While no instrument was used to measure outcomes, participants did make positive comments about the workshop's relevance and value (Rooda & Gay, 1993).

This study was limited in numerous ways. First, the workshop was limited by time and resources (Rooda & Gay, 1993). Additionally, the study used a convenience sample of nursing directors, did not have a control group, nor was an instrument used to measure workshop outcomes. Furthermore, no educational or cultural theory was discussed or used to develop the workshop.

Lee, Anderson, and Hill.

In 2006 Lee et al. conducted a study to evaluate the effect of an educational program about Hispanic health practices and beliefs on the cultural sensitivity of nurses

who provide care to diverse patient populations (Lee et al., 2006). A one group pre- and post-test design was used with a convenience sample of seven RNs employed in one health department. The educational intervention lasted one and a half hours and was designed to increase cultural knowledge. While no further information was given about the intervention, Lee et al. do report holding the program at the health department and using visual aids during the presentation. A significant difference (p = .018) was found in the pre- and post-test Lee Cultural Sensitivity Tool scores with the mean score increasing from 5.4 to 9.7 (Lee et al., 2006).

While not specifically addressed by Lee et al. (2006), there were several study limitations. The study used a very small convenience sample from one location and did not use a control group. Additionally, little information was given about the program and no information was given about its development and whether or not cultural or educational theories were utilized. Furthermore, the program focused only on Hispanic clients.

Khanna, Cheyney, and Engle.

Khanna et al. (2009) conducted a study to ascertain whether a cultural competence training course produces a significant change in self-assessment of the knowledge and skills needed to care for patients from diverse cultural backgrounds. Sixty doctors and nurses from one area in Oregon chose to participate in the four hour cultural competence workshop. The workshop, based mainly on the CLAS and IOM standards, addressed cultural and linguistic competency, cultural health disparities, the connection between health beliefs and culture, and how cultural competency facilitates

effective communication. Participants received four continuing education units (CEUs) for completing the course (Khanna et al., 2009).

A post-then-pre-test evaluation methodology was used. Specifically, participants completed the CCA Khanna et al. (2009) developed only at the end of the training course. Questions were designed to assess two different time frames: before and after the training. One reason Khanna et al. cite for using this methodology is that a previous study by Rockwell and Harriet suggests post-then-pre-test methodology limits the problem of ideal reporting (social desirability) which is often associated with a pre- and post-test design. Statistically significant changes were noted in both knowledge and skills related to cultural competence (Khanna et al., 2009).

While not specifically addressed by Khanna et al. (2009), the study was limited in many ways. It used a convenience sample and did not use a control group. No specific cultural competence theory or educational theory was used to guide training development. Finally, while Khanna et al. defend their choice of using a post-then-pretest design, their choice is somewhat unconventional and therefore questionable.

Schim, Doorenbos, and Borse.

Schim, Doorenbos, and Borse (2006b) examined change in cultural competence following a face-to-face educational intervention. The sample for their quasiexperimental, longitudinal, crossover designed study included 130 hospice workers (administrators, clergy, clerical support, nurses, nursing assistants, social workers, and volunteers) from eight different hospice agencies. The hospice workers were randomized into one of two groups. One group completed the CCA, followed by a one hour cultural competence training session which was modeled after the *End-of-Life Nursing Education* *Consortium Training Materials Module 5: Cultural Considerations in End-of-Life Care.* The materials were congruent with the provider level of the 3-D model and included both cultural awareness and sensitivity components (Schim et al., 2006b). Interventions which reflect cultural competence behaviors were also discussed. After three to four months they took the CCA again, followed by a one hour ethics (control) session, finishing with completing the CCA one final time. The other group completed the CCA and training at the same intervals, but completed the ethics training prior to the cultural competence session. The study, utilizing repeated-measures multivariate analysis of covariance, found a significant increase (p = .034) in cultural competence following the cultural education sessions in both groups (Schim et al., 2006b).

Limitations of the study include the short educational intervention, the small randomized convenience sample, and not using an educational theory to guide development of the training sessions. Additionally, as is common with face-to-face educational interventions, Schim et al. (2006b) noted scheduling and travel presented significant challenges.

Smith.

Smith (1998) studied the effect of an eight and a half hour face-to-face educational intervention on the cultural competence of RNs living in one county in Alabama. The Giger and Davidhizar transcultural assessment model served as the theoretical foundation of the study which had a mixed between-within groups repeated measures quasi-experimental design. The study included 94 RNs who attended either an eight and a half hour culture school or a nursing informatics session. The culture school objectives included cultural competence nursing care concepts, identifying common

health beliefs and practices, communication, implementing culturally competent care plans, identifying cultural competence within past experiences, and identifying personal goals, strengths, and areas for improvement. RNs completing the culture school experienced a significant increase in cultural self-efficacy and cultural knowledge over their counterparts who completed the informatics course (Smith, 1998). The study was limited by all participants living in the same county and not using an educational theory to guide intervention development.

Brathwaite.

In 2004 Brathwaite conducted a study in Ontario, Canada to address the absence of a course that integrated theoretical knowledge with practical knowledge regarding cultural competence for public health nurses (Brathwaite, 2004). Brathwaite's work was based on Campinha-Bacote's model and used Campinha-Bacote's instrument, the IAPCC-R, to conduct a one-group repeated measure design study. The sample consisted of 76 public health RNs who completed a five week face-to-face course based on the Campinha-Bacote model and an additional booster session one month following the course. Each week in the five week course focused on one of Campinha-Bacote's components: cultural awareness, skill, knowledge, encounters, and desire. The booster session promoted discussion of nursing experiences and application of cultural competence concepts in practice. Adult and experiential learning theories were used to guide intervention development. The IAPCC was administered to each of the 76 participants two months prior to the five week course, immediately prior to the five week course, one week following the course, and three months after the course. Findings indicated mean scores for the IAPCC immediately prior to the course and one week after,

as well as the scores one week after the course and three months following the course, were both significantly different. Overall, the results illustrated a move in participants from being culturally aware to culturally competent or proficient (Brathwaite, 2004).

Study limitations include the threat of social desirability, meaning participants may have answered the IAPCC based on social awareness, and the limited number and diversity of participants (Brathwaite, 2005). Specifically, the study was limited by its convenience sample of public health nurses from one department in Ontario, Canada, and by not using a control group.

Lal.

In 2010 Lal conducted a study to evaluate the effect of an educational workshop on the cultural competence of acute care nurses. A convenience sample of 37 nurses was invited to participate in the study. Thirty-four nurses chose to participate in the quasiexperimental one group pre- and post-test design study which included a one and a half hour cultural competence intervention. The intervention was guided by a cultural competence teaching packet, designed by Campinha-Bacote, which utilized experiential exercises. Specifically, the workshop defined cultural competence and addressed each of the components of Campinha-Bacote's model of cultural competence. Twenty-five participants were culturally aware (as defined by the IAPCC-R) and 12 were culturally competent before the workshop, while 12 participants were culturally aware and 22 were found to be culturally competent following the workshop. Analysis with a paired t-test revealed a statistically significant change in pre-test and post-test IAPCC-R scores (p <.001). The study was limited by its non-random small convenience sample (Lal, 2010).

Other potential limitations include not using an educational theory to guide development of the workshop and not using a control group.

Summary.

The above review evidences the ability of face-to-face educational interventions to enhance cultural competence or its components of cultural awareness, skills, knowledge, and sensitivity. However, most studies have been limited in several ways. First, all nine of the studies discussed used a convenience sample from one geographic area and only three had a control or comparison group. Furthermore, five studies were not guided by a cultural competence theory and only two of the nine used an educational theory to guide educational intervention development. Finally, one study did not use an instrument to measure change and the other eight used self-report instruments which can be biased based upon social desirability. Clearly further research is necessary.

Online continuing education for healthcare providers.

As face-to-face interventions are not always practical, online CE has been used to address various topics with healthcare providers. Few online cultural competence CE sources exist; some are socially interactive while others are socially isolated. Review of the literature reveals no studies which have compared socially interactive online education to socially isolated, neither related to cultural competence or other topics.

Online socially interactive cultural competence CE.

Hunter.

In 2008 Hunter conducted a quasi-experimental pre- and post-test design study of a two-credit graduate level cultural competence course at the University of Missouri. The course was offered in two formats (one per semester) over two consecutive semesters: (a)

a socially interactive online format, and (b) a control group traditional face-to-face course (Hunter & Krantz, 2010). Campinha-Bacote's cultural competence model and constructivist learning principles were used to guide course development. The course included four modules: cultural awareness, knowledge, skills, and encounters and included activities such as defining one's own culture, writing a cultural autobiography, designing and utilizing a cultural assessment tool, and creating a cultural encounter with a culture other than one's own (Hunter, 2008). Fifty-two and 24 students completed the online and face-to-face courses, respectively. Both courses significantly increased cultural competence, as measured by the IAPCC-R (Hunter, 2008). While Hunter does not specifically address limitations, potential limitations of the study include using a convenience sample from one location and using samples from two consecutive semesters.

Kelly and Papadopoulos.

In 2009 Kelly and Papadopoulos conducted a 150 hour socially interactive eight week online cultural competence course. The purpose of the course was to help healthcare practitioners develop skills and knowledge for working with ethnically diverse clients and to encourage cultural competence in healthcare. Five graduate students (one midwife in the United Kingdom, one psychology student in Ireland, and three nurses in Greece) completed the course. The course was guided by the Papadopoulos, Tilki, and Taylor model of cultural competence and included several learning activities: online lectures, e-discussions, self-assessment exercises, reflective journaling, and module evaluations/quizzes. Formative assessment included both peer and self-assessments,

while summative evaluation was in the format of reflective journaling; no cultural competence instrument was utilized (Kelly & Papadopoulos, 2009).

Kelly and Papadopoulos (2009) cite few study limitations. First, only five students were studied, all of which were female. Additionally, most students encountered technical difficulties during the course (Kelly & Papadopoulos, 2009). Furthermore, while students felt the course was beneficial, no instrument was used to measure cultural competence. Finally, an educational theory wasn't utilized.

Summary of online socially interactive cultural competence CE.

Online cultural competence education is effective. Unfortunately, online socially interactive cultural competence courses have only been evaluated with nurses in graduate courses. This limits generalizability to all nurses because level of educational attainment has been shown to directly impact cultural competence levels (Doorenbos & Schim, 2004; Schim, Doorenbos, & Borse, 2005).

Online socially isolated educational interventions.

Jones.

A study of socially isolated cultural competence training for RNs and licensed practical nurses (LPNs) was conducted by Jones (2009). The purpose of the descriptive quantitative study was to measure the degree to which cultural competence behaviors are exhibited by nurses after completing the training program. The sample consisted of 60 nurses working at one Texas hospital who were randomly assigned to either the control or treatment group. Written self-study packets about cultural competence with case studies of several culturally diverse clients and self-graded questions were either read online or in hard copy format by the 32 nurses in the treatment group. The nurses completed the

IAPCC-R after completing the self-study packet. Participants in the control group completed the IAPCC-R during the same time period as the participants in the treatment group. The self-study significantly increased cultural competence (p < .01) as measured by the IAPCC-R (Jones, 2009).

Jones (2009) cites numerous limitations to her study. First, participant responses may have been biased by social desirability or other extraneous factors which could not be controlled. Additionally, a convenience sample was utilized (Jones, 2009). Other limitations include using a post-test only design and not using an educational theory to guide intervention development. In fact, it could even be said while the intervention was accessible online, it was only online education in the most liberal sense, as a learning management system (LMS) was not used (Halstead & Billings, 2009).

Culturally competent nursing modules.

Publicly accessible, free online modules which award CEU credits upon successful post-test completion, entitled Culturally Competent Nursing Care: A Cornerstone for Caring, were developed by the Office of Minority Health (2007). The modules are organized by the CLAS standards and Campinha-Bacote's cultural competence model was also utilized (Office of Minority Health, 2007). The first module addresses principles of cultural competency, strategies for increasing cultural awareness, and skills applicable to delivering client-centered care; the second module addresses language access, effective communication strategies, and health literacy; and the third and final module supports advocacy for cultural competence within organizations (SRA International, 2009).

A two-year review of the modules was conducted in 2009 (SRA International, 2009). Analysis topics included: change in cultural competence knowledge, change in behavior, and changing practices in healthcare organizations. A control group of 27 participants completed an abbreviated version of the modules, a survey, and pre- and post-tests. Three different participant groups, with 27 participants in each group, were formed because the original plan to use control and participant groups of 60 was not feasible due to only 27 control group participants completing the modules. The report indicated nurses scored higher on the post-test in all groups, which is said to indicate the modules, whether in full or abbreviated form, significantly increased knowledge of cultural competence, knowledge of attitudes, and knowledge of cultural competence skills (SRA International, 2009). However, as the pre- and post-test evaluated knowledge, not attitudes and skills themselves, the study's findings are limited. The small sample size and the use of a control group which underwent an abbreviated form of the modules limits the study's validity and generalizability. The study was further limited by not using a valid and reliable instrument to measure cultural competence. Finally, nowhere was it stated that an educational theory was used to guide module development.

Cultural Competence for Today's Nurses.

One additional online publicly accessible socially isolated course which awards CEUs upon completion, entitled Cultural Competence for Today's Nurses, Part One: Culture and Women's Health (Gerace & Salimbene, n.d.), has been identified. However, the authors do not indicate an educational theory was used to develop the CE, no cultural

competence theory was utilized, material is mainly factual and knowledge based, and no studies have evaluated its efficacy at increasing cultural competence.

Summary.

Existing online cultural competence CE for nurses is socially isolated. While Jones' (2009) study and the Culturally Competent Nursing Modules utilized a cultural competence theory, no online socially isolated cultural competence CE has utilized an educational theory to guide intervention development. Furthermore, only Jones (2009) measured cultural competency with a valid and reliable cultural competence tool; no conclusive evidence supports the ability of the Culturally Competent Nursing Modules (Office of Minority Health, 2007) or the CE entitled Cultural Competence for Today's Nurses, Part One: Culture and Women's Health (Gerace & Salimbene, n.d.) to increase cultural competence levels.

Online Continuing Education

Much literature discusses online education. Theories or frameworks for developing online education will be discussed and the pros and cons of online education presented. The importance of social interaction in online learning will be supported followed by a review of best design practices for online courses. Finally, instruments measuring students' evaluations of online courses will be presented.

Online Educational Theories

While educational interventions show much promise for increasing the cultural competence of nurses, most studies have the major limitation of not using an educational theory to guide intervention development. Many studies have evidenced technology has minimal impact on student outcomes; rather, instructional design is the primary factor

(Johnson & Aragon, 2003). In fact, while distance "programs enhance access to continuing education for the health professional, increased access is often coupled with decreased quality in course design" (Sadera & Fisher, 2009, p. 157). Therefore, theory specific to online education should be foundational to developing online CE interventions. Four online educational frameworks or models will be described.

A framework for the use of constructivism online.

This analytical framework was first described by Bonk and Cunningham in 1998 (as cited in Weasenforth, Biesenbach-Lucas, & Meloni, 2002) and was revised and utilized by Weasenforth et al. (2002). It includes four sets of factors relevant to learning: cognitive and metacognitive, motivational and affective, developmental and social, and individual differences. The first group of factors, cognitive and metacognitive, address the nature of learning (learning is most effective when meaning is intentionally constructed from experience and information), goals of learning (learners create meaningful representations of knowledge with support and guidance), how knowledge is constructed (new information is linked with existing knowledge by learners in meaningful ways), strategic thinking (learners can use strategies to achieve complex learning), thinking about thinking (creativity and critical thinking are encouraged by higher-order strategies), and the context of learning (learning is impacted by the environmental factors of technology, instructional practices, and culture). The second set of factors, motivational and affective, address emotional and motivational influences on learning (motivation impacts what is learned and emotions impact motivation), the intrinsic motivation to learn (motivation is stimulated by novel but difficult tasks which are relevant to personal interest and choice), and how motivation effects effort (little is

learned without exertion of effort). The third set of factors, developmental and social, address developmental influences (learning is most effective when development is taken into account) and social factors influencing learning (social interactions and interpersonal relations impact learning). Finally, the fourth set of factors, individual differences, address individual differences for learning (learners have unique strategies and styles which are a function of their background), the impact of diversity on learning (learners have diverse backgrounds which must be taken into account), and assessment (high standards for learner assessment is most important).

Model of online learning.

The model of online learning is a blended model for online education (Anderson, 2008). The model itself depicts the relationship between the two major human actors of learners and teachers as well as the interaction of human actors and their relationship with course content. The model includes numerous types of interaction: student-to-student, student-to-teacher, student-to-content, teacher-to-teacher, teacher-to-content, and content-to-content. Each of these interactions will be described.

Student-to-student interaction is of paramount importance in both constructivist and connectivist theories (Anderson, 2008). Consequently, collaborative learning and student-led teams are appropriate educational strategies. Student-to-content interaction, which is depicted in the model as independent study, has long been a major part of education. Today the Internet supports both passive and active student-to-content interaction through virtual labs, computer-assisted learning modules, and immersion in micro-environments. Student-to-teacher interaction is referred to in the model as a community of inquiry. Student-to-teacher interaction can be supported either

synchronously or asynchronously through the use of audio, video, and text communications. Teacher-to-content interaction employs objectivist learning principles such as content modules or units, learning objects, and learning activities which are designed by the teacher. Thus, teachers can monitor, build, and update courses and activities. However, teacher-to-teacher interaction is also important as it encourages professional development and support. Content-to-content interaction is the final interaction described and is unique to this model. It is a developing mode of interaction in which programs link content with other information sources to automatically and continuously update and acquire new information (Anderson, 2008).

Interestingly enough, Anderson (2008) states "various forms of student interaction can be substituted for each other, depending upon costs, content, learning objectives, convenience, technology used, and time availability. The substitutions do not decrease the quality of learning that results" (p. 66). Anderson continues by saying meaningful learning can occur as long as at least one of the three forms of interaction (student-toteacher, student-to-student, or student-to-content) is utilized at very high levels. He states the other two areas may even be eliminated (Anderson, 2008). This seems to present a dichotomy, as the model presented supports the use of all types of interaction, yet its developer supports the use of as little as one type of interaction within Web-based teaching and learning.

Framework for developing new e-learning programs.

An article by Booth, Carroll, Papaioannou, Sutton, and Wong (2009) presents a framework for developing new e-learning programs. The framework was developed following the review of 29 studies in which each explored workplace-based e-learning

(WBEL) in the United Kingdom. Review of the studies yielded several themes and subthemes of WBEL, including: presentation and design, flexibility, peer communication, support, and knowledge validation. The sub-themes included learning control, applicability, attractiveness, usability, offline working, asynchronous engagement, learning interaction, peer support, moderated learning, formal support, and assessment (Booth et al., 2009). Each of these sub-themes will be explored in greater detail.

Flexibility is of paramount importance to e-learning (Booth et al., 2009). Learner control should be allowed; one option is self-regulated work pace with extra weeks built into courses as make-up time missed due to life events. Group-supported synchronous learning is another appropriate option. The second sub-theme, applicability, is providing a realistic learning context. The use of guest speakers, problem-based scenarios, and case studies are suggested to increase the applicability of e-learning. The third sub-theme, attractiveness, relates to the e-learning environment itself, as learners often resent superficial attempts at designing e-learning courses. Simply placing textual materials online does not constitute an online course. Rather, courses should be made less dependent on simple text by including PowerPoint presentations, podcast lectures, or other visually stimulating formats. The fourth area, usability, refers to ease of using elearning technologies. Even low-tech applications with minimal difficulty can be challenging for novice online learners, so course developers should remove impediments and decrease the impact of technical difficulties in online courses. One specific suggestion regarding written directions is that they be very clear and easily accessible. The fifth area, offline working, refers to the paradox that online learners still often desire to have course materials in a portable format. This includes being able to access courses

from home, work, and other locations. Booth et al. (2009) suggest offering print materials, MP3 lecture files, and CD-ROM or USB stick course materials to support offline working. The sixth sub-theme, asynchronous engagement, refers to the ability to complete course work at any time during a course. Offering asynchronous options and having facilitators who are accessible via email for questions is advised. The seventh sub-theme, learner interaction, is an area of tension, as learners often desire shared learning experiences and feel pressure based on deadlines and limited time. Self-directed learning is proposed as an alternative. Furthermore, the eighth area, peer support, is imperative in conjunction with formal facilitator support and interaction, which is the ninth area. Formal support is also a key factor for student experiences, as prompt responses, support which is available outside of normal working hours, and technological support are all of mammoth importance. The final area, assessment, includes both formative and summative evaluation, as both are essential to keeping learners motivated. Two assessment methods are specifically recommended: quizzes and having learners apply principles to their own work setting (Booth et al., 2009).

The instructional strategy framework for online learning environments.

The instructional strategy framework for online learning environments, or the ISFOLE, by Johnson and Aragon (2003) supports social interaction in online education. The framework assumes learning is a complex process which can't be explained by a single learning theory. Rather, quality online learning environments should be built upon principles derived from behavioral, cognitive, and social learning theories (Johnson & Aragon, 2003). Within the model, powerful environments for on-line learning: "(1) address individual differences, (2) motivate the student, (3) avoid information overload,

(4) create a real-life context, (5) encourage social interaction, (6) provide hands-on activities, and (7) encourage student reflection" (Johnson & Aragon, 2003, p. 34). This model was utilized in this study and is further described in Chapter 3.

Benefits of Online Education

There are many positive aspects to online education. First and foremost, online education is effective (Billings, 2000; Cobb, 2004; Halstead & Billings, 2009; MacDonald & Walton, 2007; Pullen, 2006; Shachar & Neumann, 2003; Southernwood, 2008). And to this point, online education increases knowledge (Dunet, Reyes, Grossniklaus, Volansky, & Blanck, 2008; Huckstadt & Hayes, 2005; Schneiderman & Corbridge, 2009), enhances clinical practice (Pullen, 2006), increases the use of research in practice (Atack, 2003), and increases assessment skills related to questioning (Atack, 2003) and physical examination (Dunet et al., 2008). Additionally, students (Billings, 2000) and practicing nurses completing online CE increase their computer skills (Huckstadt & Hayes, 2005; Wilkinson, Forbes, Bloomfield, & Gee, 2004).

Many advantages of online education relate to its freedom from time or location limits. Online education is convenient (Ali, Hodson-Carlton, & Ryan, 2004; Atack, 2003; Billings, 2000; Huckstadt & Hayes, 2005), flexible (Ali et al., 2004; Southernwood, 2008), and easy to access from anywhere at any time (Huckstadt &Hayes, 2005; Ramsey & Clark, 2009). In fact, online education exposes learners to many viewpoints (Ramsey & Clark, 2009) and resources (Ali et al., 2004), facilitates life-long learning (Ramsey & Clark, 2009; Southernwood, 2008), and is not limited by space or location (Billings, 2000; Ramsey & Clark, 2009). Web-based education has also been found to be satisfying (Billings, 2000; Cobb, 2004; Halstead & Billings, 2009), time saving (Huckstadt & Hayes, 2005), and economical or cost effective (Billings, 2000; Huckstadt & Hayes, 2005; Southernwood, 2008; Zhang, Zhao, Zhou, & Nunamaker, 2004). Furthermore, online environments foster autonomy, allow time for reflection and thoughtful writing (Ali et al., 2004; Song, Singleton, Hill, & Koh, 2004), provide a less threatening environment than traditional classrooms (Southernwood, 2008), and decrease bias because gender, race, and physical characteristics are not revealed (Pallof & Pratt, as cited in Yang & Cornelious, 2005). Finally, most providers who have completed online CE prefer and recommend it (Bernhardt, Runyan, Bou-Saada, & Felter, 2003). Clearly online education has many advantages.

Challenges of Online Education

While there are many advantages to online education, it is certainly not without limits or challenges. Many of the most common limitations relate to the use of computers and the Internet. For example, online learners must have at least basic computer proficiency (Cobb, 2004; Muilenburg & Berge, 2005; Ramsey & Clark, 2009; Shade & Barber, 2004; Song et al., 2004) and access to a computer in a private location (Atack, 2003; Wilkinson et al., 2004). High-speed Internet access is preferred (Muilenburg & Berge, 2005; Shade & Barber, 2004) and it is important to have access to technical support services because technical difficulties are common (Ali et al., 2004; Cobb, 2004; Ramsey & Clark, 2009; Song et al., 2004).

Online learners are most successful when they have certain characteristics. Successful online learners are self-motivated (Muilenburg & Berge, 2005; Ramsey &

Clark, 2009; Song et al., 2004), self-disciplined (Atack, 2003; Ramsey & Clark, 2009; Song et al., 2004), persistent (Morris & Finnegan, 2008), self-reliant (Morris & Finnegan, 2008), and can manage their own time effectively (Song et al., 2004). Likewise they are able to seek assistance when needed (Billings, 2000).

There are also disadvantages or risks for the teacher/facilitator and their institution. For example, online education must be designed appropriately, requires modification of traditional teaching style, and has a high initial set-up cost (Ramsey & Clark, 2009). It is also time consuming to set-up and run (Ramsey & Clark, 2009; Zhang et al., 2004). Similarly, faculty must regulate their time to ensure they do not become overcommitted to one course and new workload policies may need to be employed (Ramsey & Clark, 2009).

The final limitation of online learning is that learners often feel closed off from others. Many students report feeling socially isolated at times (Ali et al., 2004; Muilenburg & Berge, 2005; Southernwood, 2008) and perceiving a lack of community (Song et al., 2004). In fact, some online learners report the desire for face-to-face contact (Pullen, 2006) or at least more interaction with others (Wilkinson et al., 2004) and more feedback (Song et al., 2004; Zhang et al., 2004). Obviously social interaction is important in online education.

Social Interaction in Online Education

Review of the literature evidences the importance of social interaction within distance education. In fact, it has been said that social communication is essential for online CE to be successful (Guan, Tregonning, & Keenan, 2008). Specifically, a review of 14 studies of nursing distance education concluded learners perceive: (a) socialization

occurs via timely feedback and by interaction between students and faculty; (b) learning entails supporting the values and contributions of other learners; (c) faculty should be efficient at leading conversations and should provide support; and finally (d) communication is critical (Mancuso-Murphy, 2007). A meta-analysis of 51 studies, which included over 11,000 participants, explored factors which influence the outcomes of grades, student satisfaction, and student participation in distance education (Zhao et al., 2005). Findings indicated instructor involvement had the greatest impact on outcomes; when involvement is low, face-to-face education trumps distance education. However, when the instructor is moderately or highly-involved, the outcomes of distance education are better (Zhao et al., 2005). Another study found a positive correlation between students agreeing web courses were convenient and active learning, feedback, faculty interaction, interaction with peers, satisfaction, practicality, socialization, and connectedness (Billings, Connors, & Skiba, 2001). There is a significant positive correlation between classroom community, connectedness, and learning (Rovai, 2002). Students who receive individualized feedback perform better than those who do not and are more satisfied with the course in general (Gallien & Oomen-Early, 2008). Current research supports social interaction in distance education.

Best Design Practices for Online Education

Literature review reveals a plethora of information regarding best practices in online teaching and learning. Numerous topics, areas, and suggestions will be discussed in the following section. Specifically, page layout, content organization, text/fonts, color, and use of graphics will be described in further detail.

Page layout.

Page layout is an important aspect of learner usability. One part of page layout which greatly impacts usability is course navigation. Material should be linked from the course home page (Dunet et al., 2008; O'Neil, 2009) and learners should be able to access everything within the course using no more than three mouse clicks (O'Neil, 2009). Learners should be able to navigate pages and return to previously viewed pages with little effort (Kordel, 2008). Navigation is facilitated by giving each page a specific title (O'Neil, 2009), using gridlines to assist in formatting (Gillani, 2003), and ensuring course buttons and navigation tools work as expected (Kordel, 2008).

The actual layout of each page included in the course is also important. Ample white space, or empty space, should be included (Dunet et al., 2008; Gillani, 2003) and scale, contrast, and hierarchy used to make pages visually appealing and easy to follow (Gillani, 2003). Learners' attention should be directed to information and concepts which are critical or confusing (Dunet et al., 2008). This can be accomplished by using a bulleted format, which should be utilized whenever possible (Dunet et al., 2008), and by putting the most important information first (Buhmann & Johnson, 2009). Additionally, page length should be regulated, as Webpage length should fit the screen without scrolling (Buhmann & Johnson, 2009). Finally, consistency is key; the same things should be in the same place throughout the course site including: colors, fonts, headings, text, navigation, and etc. (Gillani, 2003; O'Neil, 2009).

Content organization.

Content organization is also of utmost importance to course design. Each course should have an introduction to familiarize learners with the course (Blood-Siegfried et al.,

2008). Course content should be presented as modules (Dunet et al., 2008; Gillani, 2003) which have separate introductions (Wilkinson et al., 2004), clearly defined objectives and expectations (Blood-Siegfried et al., 2008; Dykman & Davis, 2008), and are around one week in length (Dykman & Davis, 2008). Each module, or unit, typically begins with a written or recorded lecture which summarizes the concepts presented in the module. Other audiovisual materials are also commonly utilized to present information. Discussion questions or other teaching and learning activities are then completed by the learners. Specific timeframes, transparent assignments, and clear due dates are critical in online education, as they foster a sense of control in learners which allows them to feel more comfortable and increases their motivation (Dykman & Davis, 2008). Furthermore, it is suggested that courses start slow, with a lighter workload early in the course to allow students to become comfortable with online learning and the structure of the course

(Dykman & Davis, 2008).

Guidelines are also provided for structuring modules. Instructional modules should be divided into subunits which have headings and subheadings for clarity (Blood-Siegfried et al., 2008). Material within modules should be chunked into small sections (O'Neil, 2009) which contain between five and nine bits of information (Cercone, 2008). It is important to limit main text, to the extent possible, only to key points or ideas, assimilating these for review in a format which can be easily downloaded or printed by learners (Dunet et al., 2008).

Text and fonts.

While discussing text style and font size may seem quite tedious, specific suggestions are given for their use in online courses. Sans serif fonts, such as Helvetica,

New York, Arial, or Veranda should be used for main text (Buhmann & Johnson, 2009; Gillani, 2003) while serif fonts such as New Century or Times New Roman should only be used, if used at all, for page titles and other large distinct text (Gillani, 2003). No more than three font styles should be utilized (O'Neil, 2009) and both upper and lower case letters should be utilized (Gillani, 2003). An adequately sized font, such as 10 or 12 point should be used for the body text, while a 12 or 14 point font is appropriate for headings and titles (Buhmann & Johnson, 2009). In general, bold should not be overused and, to avoid confusion with hyperlinks, text should not be blue (O'Neil, 2009). Furthermore, text should not be underlined since hyperlinks are commonly underlined (Blood-Siegfried et al., 2008; Buhmann & Johnson, 2009) and use of all uppercase letters should be avoided because it is commonly construed as yelling at the learner (Blood-Siegfried et al., 2008). Above all, consistency is key; each page within the course should have similar font and text styles (Gillani, 2003; O'Neil, 2009).

Use of color.

Color use within online courses is important because of visual appeal and the feelings different colors inspire. In general, no more than five plus or minus two colors should be used within a course and both dark and light colors should be utilized. Black or other dark text on a white or light background is easiest to read and pale blue or purple are commonly preferred background colors. Blue inspires calmness and suggests confidence and comfort. Conversely, red suggests danger or alertness while orange calls attention. Green suggests action or cleanliness and yellow suggests fun. White indicates purity and black suggests mystery (Gillani, 2003). Certainly color use impacts the feel, attitude, and atmosphere of an online course.

Graphics.

Graphics enhance online courses and add visual appeal to the course. However, consideration must be given when selecting graphics for use. Graphics should not be used solely for artistic or aesthetic appeal; rather, they should convey information or support concepts (Gillani, 2003). Cluttering course pages with extraneous graphics or images should be avoided (Dunet et al., 2008). Icons are appropriate and commonly used for navigation, but they should be simple, clear, and used consistently throughout a course (Gillani, 2003). Graphics Interchange Formatted (GIF) images are preferred for Webpage use and images often need to be compressed or reduced in size to allow for easy downloading (Gillani, 2003). Note it is important to be conscious of copyright law when selecting and utilizing graphics (Blood-Siegfried et al., 2008; Buhmann & Johnson, 2009).

Additional recommendations.

Four additional recommendations are evident in current literature. First, class size should be limited to allow for adequate time, student feedback, and to facilitate class discussions. Ideally, when new courses are being offered or new course instructors are utilized, class size should be limited to 15 to 20 students at most (Dykman & Davis, 2008). Experienced online educators teaching established online courses may be able to effectively teach a class of 25 or even 30 students (Dykman & Davis, 2008). Next, online courses should have an area to share personal information and have questions answered (Blood-Siegfried et al., 2008). In other words, discussion boards should have areas for interaction which do not focus on specific course objectives or content. Additionally, courses should include information about being an online learner (Blood-

Siegfried et al., 2008) and technical support should be available to learners. Ideally tech support is available 24 hours a day, seven days a week, as online learning occurs around the clock.

Teaching and Learning Strategies for Online Education

Many teaching and learning strategies are referenced in online educational literature. Several of these strategies will be discussed in detail. It is important to remember multi-modal strategies which include visual, textual, auditory, and/or interactive activities support the various learning styles and should therefore be utilized (Blood-Siegfried et al., 2008).

Case studies.

Case studies have been highly recommended for employment online (Conrad & Donaldson, 2004; Dunet et al., 2008; Johnson & Aragon, 2003; Ramsey & Clark, 2009). In fact, a survey of nursing faculty indicated 90% of educators believe case studies are the most effective online teaching approach (Christianson, Tiene, & Luft, 2002), Huckstadt and Hayes (2005) found 73 graduate level nurses participating in online CE enjoyed the case study approach, and Cobb (2003) found nearly 70% of nurses and over 60% of physicians believe CE should include case studies.

Case studies are authentic learning activities (Conrad & Donaldson, 2004) in which in-depth analyses of pragmatic situations are used to apply course content to reallife settings (Rowles & Russo, 2009). Case studies stimulate critical thinking and are highly appropriate for adult learners (Rowles & Russo, 2009). Johnson and Aragon (2003) purport student learning increases as the number of real-life examples and case studies utilized increases. In fact, Dunet et al. (2008) conveys physicians and nurses enjoyed using case studies in online CE and found them to be engaging.

Conrad and Donaldson (2004) provide an example of a case study and give guidelines for developing strong case studies. Well-designed case studies have specific objectives and instructions and can be completed individually or in collaboration with peers (Conrad & Donaldson, 2004). It is helpful to tie up all of the loose ends at the end of a case study assignment by pooling the main points of the discussion and clarifying any areas of confusion (Conrad & Donaldson, 2004). Unequivocally case studies are appropriate for online education.

Reflective journaling.

Reflective journaling is another strategy which has many benefits and is highly referenced in the literature. Reflective journaling can either be a private journal in which students only share their thoughts with the facilitator or it can be used to stimulate discussion in a group setting (Bender, 2003). Many types of reflective journaling are suggested; in fact, no method is inherently better than others, because they each serve different purposes (Phipps, 2005). Types of reflective journaling frequently implemented in online education include: spiritual journals; diaries; memoirs; professional logs, or logs which record growth within a field of study; theory logs, or those which examine theoretical concepts and points; technical reflection of performance based upon evaluative criteria; reflection in action, which refers to thinking of better implementation strategies for what has been previously done or observed; deliberative reflection, or examining and comparing one's own opinions to the opinions of others to determine the

best possible solution within a situation; personal reflection of motives, worldview, ethics, and beliefs; and critical reflection (Phipps, 2005).

Journaling is an ideal activity for online education because it allows assimilation of multiple ideas, draws on student experiences, and applies course material to real-life situations (Bender, 2003). Critical incidence analysis, often incorporated into reflective journaling, asks students to provide a description and analysis of a particular situation they have experienced (Rowles & Russo, 2009). Finally, reflection prompts affective change, as it allows one's beliefs, actions, and understanding to be challenged. For many, reflection leads to changes in values and assumptions (Johnson & Aragon, 2003).

Multiple suggestions are given for the use of reflective journaling. First, the objectives and expectations for the journaling exercise should be made clear. This can help decrease student perception that reflective journaling is simply busy work. Second, thoughtful feedback should be given to guide development and may come from either faculty or from group discussion and peer feedback. Finally, reflective journals are typically graded as pass/fail (Rowles & Russo, 2009).

Games.

Games are an effective teaching strategy and learning tool in the online environment which increases student motivation (Conrad & Donaldson, 2004; Johnson & Aragon, 2003). In fact, games allow skills and knowledge to be gained and enhanced in a fun way (Conrad & Donaldson, 2004; Rowles & Russo, 2009) while leading to longer retention of material than by lecture alone (Rowles & Russo, 2009). Uniquely engaging, games are especially appropriate for adult learners who tend to take responsibility for their learning and often appreciate receiving prompt feedback (Rowles & Russo, 2009).

Designing games appropriately is crucial in online education. It is suggested that games should involve the student directly in decision making and allow them to learn from the outcomes. Specifically, games should include tasks which engage students and encourage active learning. As such, the game must provide an environment which is safe for exploration and allows for kind feedback (Conrad & Donaldson, 2004).

Virtual field trips.

Virtual field trips are an interactive strategy which is tailored to online learning needs. In fact, virtual field trips have been compared to simulation and are a beneficial active learning technique (Bender, 2003). Virtual field trips simulate visits to real locations and facilitate interactions with different places, geography, and persons (Conrad & Donaldson, 2004). Although they are often set-up by the course leader or facilitator, students can be encouraged to discover additional sites on their own (Conrad & Donaldson, 2004). Following site exploration, students can respond to various prompts including what they learned from their visit, whether or not they would suggest the site be used for future virtual fieldtrips, and what they would like to explore in person (Conrad & Donaldson, 2004).

Discussion.

Discussion has been used extensively in online courses and is supported by numerous sources (Bender, 2003; Halstead & Billings, 2009; O'Neil, 2009). Discussion is well suited for open-ended questions which reflect on the course content (O'Neil, 2009). Discussion also works well for facilitating peer-to-peer feedback and interaction when using case studies, problem-based learning, or reflective journaling. In fact, it has been suggested that effective online discussion facilitation encourages active learning,

collaborative learning, critical thinking, and reflection (Halstead & Billings, 2009). Simply stated, effective discussion feedback is honest, timely, respectful, relevant, empowering, and open (Halstead & Billings, 2009).

Lecture.

While traditional lecture has long been a staple of face-to-face classroom education, it has not been used extensively in online coursework. However, that is not to say it does not have a place in Web-based teaching and learning. In fact, lecture allows content to be presented in multiple formats online. For example, lectures can be written and posted as documents, audio streamed and synchronized with presentations such as PowerPoint, or presented as actual video footage, thus facilitating learning for visual and auditory learners (Johnson & Aragon, 2003). A more recent technology, podcasting, is another way to present lecture material online (Ramsey & Clark, 2009; Sadera & Fisher, 2009). While not often center stage, lecture has a place in online courses.

Summary.

Review of pertinent literature presents numerous teaching and learning strategies which are applicable to online education. Many of the strategies described support active learning, which promotes deeper understanding (Bangert, 2008). Different strategies may work better for some students than for others. It is therefore essential that online courses utilize more than one. Each strategy has a place within online teaching and learning.

Students' Evaluation of Online Education

Online course evaluation should include both objective areas, such as student completion, success, and grades as well as more subjective areas such as student

satisfaction and evaluation of the course itself. Five instruments have been commonly cited in the literature which measure learners' perceptions of teaching effectiveness, satisfaction, and learning environment. Each of these instruments will be described in detail.

Online Course Satisfaction Survey.

The Online Course Satisfaction Survey (OCSS) was developed in 2004 by Bolliger and Martindale to measure learner satisfaction in online courses. The Telecourse Evaluation Questionnaire (TEQ), constructed by Biner, was modified by Bolliger and Martindale to make it applicable for online courses (Bolliger & Martindale, 2004). A review of literature related to student satisfaction and course evaluation also impacted item development. The original OCSS contained 60 items: 42 used a Likerttype scale to address instructor, technology, course management, course Website, interactivity, and general issues; the other 18 items addressed demographic and general information related to the course. Three hundred and thirty graduate students participated in the pilot study. Confirmatory factor analysis, using varimax rotation, was performed. Four areas had eigenvalues greater than one; however, the scree plot indicated only three components existed. The three factors identified, named instructor, technology, and interactivity, explained 72% of the variance. Internal consistency reliability was found to be .99 for the entire scale, .98 for the instructor subscale, .93 for the technology subscale, .94 for the course management subscale, .96 for the course Web site subscale, .88 for the general issues subscale, and .83 for the interactivity subscale (Bolliger & Martindale, 2004). It is noteworthy that internal consistency reliability was calculated for six separate subscales after the factor analysis confirmed only three factors.

Distance Education Learning Environments Survey.

The Distance Education Learning Environments Survey (DELES) was developed by Walker and Fraser (2005) to measure the psychosocial aspects of learning environments in post-secondary distance education (Walker & Fraser, 2005). Review of existing instruments and current psychosocial learning environment literature guided the development of the DELES' 34 items. The original 48 items were piloted with 680 distance education students. Factor analysis was conducted and six factors emerged: (a) instructor support, (b) student interaction and collaboration, (c) personal relevance, (d) authentic learning, (e) active learning, and (f) student autonomy. These factors, referred to as learning environments, had 34 items with factor loading values of at least .54 which were retained. The reliability of the subscales was .87, .94, .92, .89, .75, and .79 respectively; scale internal consistency reliability was not reported (Walker & Fraser, 2005).

Distance Course Evaluation Instrument.

The Distance Course Evaluation Instrument (DCEI) was designed to address unique aspects of distance education while remaining consistent with instruments frequently used to evaluate on-campus college level courses (Roberts, Irani, Telg, & Lundy, 2005). Four hundred randomly selected distance education students were asked to list factors which they believed affected the quality of distance education. The data was then analyzed and 85 independent concepts or items were identified. Nine dimensions were identified and categorized as learner-instructor interaction, learnercontent interaction, learner-learner interaction, course organization, facilitator, instructor, technical support, support services, and delivery methods. A panel of 15 experts then
reviewed the 85 items, divided into the nine dimensions, and ranked them on a 5-point Likert-type scale of not necessary to essential. Thirty-eight items having mean scores greater than 4.25 were included in the instrument which was then piloted with 194 students, refined, and tested on 112 students. Internal consistency reliability was found to be .95; subscale reliability was not reported (Roberts et al., 2005).

Online Learning Environment Survey.

The Online Learning Environment Survey (OLES) was developed by Trinidad, Aldridge, and Fraser (Trinidad, Aldridge, & Fraser, 2005). The tool was developed to obtain feedback on students' perceptions of online learning environments. The tool is unique because it evaluates both students' experienced and preferred learning environments. Four existing instruments were reviewed during development of the OLES and concepts or scales from each were included in the OLES. The original OLES version had 62 items with the two response scales of actual and preferred. A 5-point Likert-type scale of almost always to almost never was utilized and the tool was piloted on 325 students enrolled in one of 11 e-learning classes. Principle factor analysis, using oblique rotation, was conducted and 52 items had loading values of at least .30 on the eight scales which emerged (computer usage, teacher support, student interaction and collaboration, personal relevance, authentic learning, student autonomy, equity, asynchronicity, and enjoyment) were retained. Internal consistency reliability of the actual and preferred scales was found to be .89 and .90, .93 and .96, .93 and .94, .86 and .93, .89 and .95, .90 and .95, .96 and .97, and .87 and .89 for each of the subscales, respectively (Trinidad et al., 2005).

Student Evaluation of Online Teaching Effectiveness.

The Student Evaluation of Online Teaching Effectiveness (SEOTE) was developed by Bangert in 2004 to assess constructivist online teaching practices (Bangert, 2008). The SEOTE uses a six-point Likert-type scale which ranges from strongly agree to strongly disagree. Open-ended questions are also included to evaluate student perception of online teaching effectiveness. The original scale included 35 items designed to assess the constructivist teaching principles supported by Chickering and Gamson's Seven Principles of Effective Teaching. Content validity was established by online instructors and the 35 items were piloted on 24 graduate students. Internal consistency reliability was .94. Factor analysis was then conducted with 489 college students, using principle component analysis with oblique rotation. Four factors emerged: student-faculty interaction, active learning, time on task, and cooperation among students; 26 items loaded at least .40 on one factor. A second confirmatory factor analysis was conducted with 807 graduate and undergraduate college students enrolled in online courses. Twenty-three items were retained which loaded at least .40 on one factor. A final study of 403 students yielded internal consistency reliabilities of .94 for the student-faculty interaction subscale, .86 for the cooperation among students subscale, .82 for the time on task subscale, and .85 for the active learning subscale; total instrument internal consistency reliability was not reported (Bangert, 2008).

Summary

Literature review reveals numerous cultural competence theories and instruments which have been designed to measure cultural competence or its components. Research has shown face-to-face cultural competence CE for healthcare providers is effective, but

such research has been commonly limited by using a convenience sample from one geographic area, not using a control group, an educational theory, nor a cultural competence theory to guide intervention development. Research has shown online CE for healthcare providers to be effective. However, socially interactive online cultural competence CE has only been studied with graduate students and little research has been done on socially isolated online cultural competence CE. This is unfortunate because online education has many advantages. Online education must be designed following best design practices and should utilize teaching and learning strategies which are recommended for online use. One limitation of many online CE courses is not using an educational theory to guide course development. As social interaction has been proven to be important, online educational theories and frameworks supporting interaction were reviewed. Due to these findings, this study was designed to evaluate the effect of two different online CE interventions (socially interactive versus socially isolated) on the cultural competence level, as measured by the CCA, of women's health nurses providing care for childbearing women or newborns in the US.

CHAPTER 3

THEORETICAL FRAMEWORKS

Chapter 2 included a review of cultural competence theories, models, and frameworks. It also reviewed several frameworks or models of online education. This chapter will discuss the two theoretical foundations which underpin the study in greater detail.

3-Dimensional Puzzle Model of Culturally Congruent Care

The Schim and Miller cultural competence model (SMCCM), originally described in 1999, defined cultural competence as the integration of diversity experiences (or fact), awareness (or knowledge), and sensitivity (or attitude) into practice (Schim et al., 2003). The model had a stepwise progression, meaning individuals had to first have diverse experiences which then allowed them to gain knowledge and awareness of differences and similarities amongst groups and individuals. This, in turn, allowed sensitivity to develop. Finally, cultural competence behaviors could occur (Schim et al., 2007).

In 2004 the SMCCM was re-conceptualized as a puzzle with four pieces. The four pieces were the traits or competencies needed for a provider to be culturally competent (cultural sensitivity, diversity, awareness, and competence behaviors). This change demonstrated the interconnectedness of the four parts and reinforced that cultural competence is an ongoing process (Doorenbos & Schim, 2004), not an end point (Schim et al., 2005).

The model is currently described as a 3-dimensional model in which culturally congruent care is the result of nurses and clients working together with cultural respect and humility (Schim et al., 2006a). This current 3-D model includes two levels: a

provider level in which the cultural competencies a provider must have to participate in culturally congruent care continue to be represented by the puzzle model developed in 2004, and a client level. Hence, the 3-D model moves beyond sole focus on the provider to encompass the client as well. Unfortunately, at this time the constructs of the client level have yet to be delineated and therefore will not be discussed (Schim et al., 2007). See Figure 1 for a visual representation of the provider level of the 3-D model.



Figure 1. Provider level of the 3-Dimensional puzzle model of culturally congruent care. Adapted from "Culturally Congruent Care: Putting the Puzzle Together," by S. Schim, A. Doorenbos, R. Benkert, and J. Miller, 2007, *Journal of Transcultural Nursing, 18*, p. 105. Copyright 2007 by SAGE Publications.

Cultural Diversity

The first puzzle piece, cultural diversity, is a fact of life (Schim et al., 2007). The

World is essentially becoming smaller as travel, the Internet, and technology improve.

However, cultural diversity experience is not equal for all; it varies in quantity and quality. Exposure to cultural realities, the magnitude of people encountered, and the strength of cross-cultural encounters varies amongst individuals (Schim et al., 2007). Ultimately, cultural diversity experience and the recognition of diverse groups with unique beliefs, values and customs forms a foundation upon which to build cultural competence (Doorenbos & Schim, 2004).

Cultural Awareness

The second part of the puzzle, cultural awareness, is a cognitive concept which implies there is not only a reality to contemplate, but also the ability to process knowledge (Schim et al., 2007). In healthcare, cultural awareness is having knowledge of the cultural expressions in which groups tend to be both similar and different (Doorenbos et al., 2005). Consequently, cultural awareness allows one to think globally about how cultural groups may be similar or different and is therefore defined as knowledge of areas in which large between-group differences frequently occur (Schim et al., 2007).

Awareness of variation in lifestyles, values, practices, and beliefs both within and between groups is essential to nursing assessment and intervention (Schim et al., 2006a). However, it is impossible to have knowledge of everyone. Simply memorizing facts about common differences between cultural groups is not advised because the variation within any group is likely to be large; hence, formulating general ideas often leads to stereotyping. Furthermore, formulating general ideas about culture based on knowledge gained from a few individuals or a small sample of individuals who share a common culture also often leads to stereotyping. Therefore, awareness of difference patterns is

highly valuable because awareness allows for assessment and dialogue between client and provider (Schim et al., 2007).

Cultural Sensitivity

The third piece of the puzzle, cultural sensitivity, is an affective construct which includes values, beliefs, and attitudes (Schim et al., 2007). Central to the concept of cultural sensitivity are an individual's attitudes about themselves and others, as well as their openness to new experiences and exposures. Cultural sensitivity's focus is approaching the client with humility and, rather than taking a position of knowledge, being a learner (Schim et al., 2007).

Unfortunately, much of Western culture is egocentric, placing high value on treating others as we ourselves would want to be treated. According to Hsieh (as cited in Schim et al., 2007), this is a cultural fallacy, as an individual's culture is used as the standard of comparison for all other cultures. "This admonishable perspective is even called 'the golden rule' in Western Christian cultures. When we approach others with an attitude of cultural humility, we can demonstrate cultural sensitivity by 'treating others as they would like to be treated' (Lester, 1998)" (Schim et al., 2007, p. 107).

Cultural Competence Behaviors

The fourth and final puzzle piece on the provider layer of the 3-D model is cultural competence behaviors, a behavioral construct of the actions taken in response to cultural diversity, awareness, and sensitivity. Cultural competency is an individual's ability to incorporate certain behaviors into practice (Schim et al., 2007). Cultural competence adapts and grows over time based upon diversity experiences, gaining sensitivity and awareness, acquiring new skills, and evolving abilities. Provider cultural

competence is having and using the cognitive, affective, and psychomotor skills necessary to facilitate cultural congruence between provider and client.

Following the 3-D model, a culturally competent women's health nurse would possess many attributes. For example, the nurse caring for a laboring woman from a different culture would be experiencing cultural diversity by interacting with the woman. The nurse, aware that pain and its management is a common area of cultural difference, could be culturally sensitive by realizing their view of pain may or may not be consistent with the client's view. The nurse would demonstrate cultural competence behaviors by completing a thorough assessment of cultural practices related to pain in childbearing and adapting their nursing care as needed, thus providing culturally competent care.

Instructional Strategy Framework for Online Learning Environments

Johnson and Aragon (2003) state online education frequently involves the simple conversion of face-to-face courses to an online setting, using traditional instructional methods such as recorded lectures, readings, and online tests. They refute this practice and have developed their own model for online learning entitled the ISFOLE. The ISFOLE assumes learning is a process which can't be explained by a singular learning theory; quality online learning should be built upon the principles of several learning theories.

Specifically, Johnson and Aragon (2003) purport online learning should utilize: behavioral learning theory elements such as positive reinforcement and repetition; cognitive learning elements such as assimilating new information with prior knowledge, limiting information volume, and addressing multiple senses; and elements of social learning such as group interaction, personal feedback, and peer assessment. Therefore,

powerful on-line learning environments: "(1) address individual differences, (2) motivate the student, (3) avoid information overload, (4) create a real-life context, (5) encourage social interaction, (6) provide hands-on activities, and (7) encourage student reflection" (Johnson & Aragon, 2003, p. 34). See Figure 2 for a visual representation of the model.



Figure 2. Copyright 2003 Wiley. Used with permission. From Johnson, S. D., and Aragon, S. R., An instructional strategy framework for online learning environments, *New Directions for Adult and Continuing Education, 100,* 31-43, Wiley Periodicals.

Individual Differences

According to Johnson and Aragon (2003), individual differences are student variations in skill, aptitude, and ability to process and apply information. Precisely,

differences exist in intelligence, preferred learning style, personality, prior knowledge, and cognitive style and control. Respect for individual differences is achieved by providing content in more than one format, encouraging active collaboration, and allowing students to have an individual locus of control such as accessing material in a different order or at a different time (Johnson & Aragon, 2003).

Student Motivation

Student motivation varies based on attention span, perceived relevance, confidence, and satisfaction. Therefore, facilitators must be able to acquire and hold student attention, so the environment should be engaging, participative, and relevant. Content should be pragmatic and relate to professional goals. Motivation can be enhanced by games, simulation, and multimedia use (Johnson & Aragon, 2003).

Avoidance of Memory or Information Overload

Memory overload occurs when one is exposed to too much information too quickly, thus limiting learning. Most individuals can handle around seven pieces of information at once; providing more information overloads short-term memory. For this reason content should be chunked into small segments for students to learn before accessing new information. Specifically, information overload can be avoided by curtailing the number of course activities, limiting the amount of content, using short lecture clips, organizing instruction by learning cycles, and providing a graphic course organizer (Johnson & Aragon, 2003).

Real Life Context of Learning

Learning is contextual and should therefore be pragmatic and resemble real life situations. Knowledge is produced and impacted by learning activities, the context in

which learning occurs, and culture. The three major premises of context are: learning is a social activity because it creates a group context, situational tools guide an individual's ability to learn, and thinking is augmented by environmental interaction. Contextual learning can be enhanced through the creation of virtual learning teams, case studies, and collaborative projects within the context of student work environments (Johnson & Aragon, 2003).

Social Interaction

Online learning should be socially interactive. Social learning theory includes elements of both behavioral and cognitive learning theory, suggesting individuals learn when interacting with others by imitation and reinforcement. Cognitive learning theory supports observation, as people can regulate their own behaviors by being cognizant of consequences. Therefore, social learning is manifested via socialization, mentoring, and social roles in which instructors and peers serve as role models. Social learning is supported by personal connections, utilizing peer feedback, interaction, and discussion (Johnson & Aragon, 2003).

Active Learning or Hands-on Activities

Active learning, referred to in the model as hands-on activities, is also imperative in the online learning environment. This is because cognitive learning theory asserts learning activities cannot be separated from cognition. Active learning in online learning environments commonly involves creativity, decision making, and problem solving. Specifically, active learning may be enhanced by the use of problem-based learning, synchronous sessions, projects, and applying skills (Johnson & Aragon, 2003).

Reflective Learning

The final principle of learning supported by the model is reflective learning. Reflective learning involves making new interpretations or revising existing beliefs related to the meaning of an experience. The reflective process allows for the review and challenge of one's ideas, understandings, and beliefs. This in turn can lead to a change in values and assumptions. Reflective learning may be facilitated by providing extensive feedback, one-minute papers, and reflective journaling (Johnson & Aragon, 2003).

Conceptual Definitions

The following conceptual definitions were used in this study: (a) socially isolated online education is education conducted via the Internet and a LMS which does not allow any interaction between course participants or with the course facilitator; (b) socially interactive online education is education conducted via the Internet and a LMS which encourages interaction between course participants and with the course facilitator; (c) CE is the additional education practicing licensed professionals need to acquire the new skills, knowledge, and attitudes needed to keep up with changes in nursing (Sadera & Fisher, 2009); (d) women's health nurses are nurses who specialize in caring for childbearing women and their families; and (e) cultural competence is having and using the cognitive, affective, and psychomotor skills necessary for facilitating culturally congruent nursing care.

Operational Definitions

The following operational definitions were used in this study: (a) socially interactive online courses are online courses in which all seven ISFOLE constructs (reflection, real-life context, motivation, individual differences, active learning,

avoidance of memory overload, and social interaction) are evident; (b) socially isolated online courses are online courses in which only six of the ISFOLE constructs are utilized; social interaction is not permitted; (c) CE is education licensed professionals receive CEUs for completion; (d) women's health nurses are RNs who work with childbearing women and/or newborns in the US; and (e) cultural competence is operationally defined by measurement with the Cultural Competence Assessment (CCA) instrument by Doorenbos et al. (2003).

Summary

In summation, provider cultural competence is having and using the knowledge, skills, and attitudes which are needed to facilitate culturally congruent care. Cultural competence can be increased by educational interventions; nurses learn from both socially isolated and socially interactive online learning. Furthermore, the cultural competency of nurses enrolled in graduate studies can be enhanced through socially interactive online education. This is supported by the ISFOLE which posits online education should promote social interaction. Hence, while both interventions should be effective, a socially interactive online CE intervention should increase the cultural competence level of women's health nurses more than a socially isolated online CE intervention.

Hypotheses

 There will be a change in the cultural competence of women's health nurses in this study. Specifically, there will be a greater increase in the CCA scores of nurses completing the socially interactive intervention compared to those completing the

socially isolated interventions; both groups will have CCA score increases which differ significantly from the control group.

- 2. Among women's health nurses, CCA scores are related to demographic variables such as highest level of education completed and previous cultural diversity training.
- CCA scores are positively correlated with the MCSDS scores of women's health nurses.

Assumptions

For the purpose of the study, the following assumptions were made:

- 1. Women's health nurses care for culturally diverse clients.
- 2. Women's health nurses are not culturally competent.
- 3. Participants respond truthfully to all questionnaires.
- 4. There is a relationship between the self-perceived cultural competence level and ability to provide culturally competent nursing care.
- Lack of culturally competent nursing care has a negative impact on the health of childbearing women and their families, while increasing nursing cultural competence decreases health disparities.

CHAPTER 4

METHODOLOGY

This chapter describes the study methodology. Specifically, the design, sample, ethical considerations, procedures, and data collection are addressed. Planned data analysis is discussed and the variables, including the online courses, described in detail.

Design

The study utilized an experimental pre- and post-test design. Three randomly assigned groups were included: two experimental groups (socially isolated versus socially interactive online CE course) and one control group. This design was chosen because experimental studies offer greater control of extraneous influences, making causal inferences appropriate (Polit & Beck, 2008).

As in any study, validity was a primary concern. The internal validity threat of testing effect was decreased by using a control group, the threat of history minimized by using a simultaneous period between the pre- and post-test for all groups, and selection bias minimized by random assignment. Additionally, the short study period should have decreased the risk of maturation effects, mortality, and attrition (Polit & Beck, 2008). As an incentive, subjects completing either course were granted CEUs and those in the control group were given the chance to complete the socially isolated online cultural competence course for CEUs after the study's conclusion. External validity was enhanced by using random assignment and using power analysis to determine appropriate study sample size to increase the representativeness of the sample.

The original study design was to utilize a pilot study for the purpose of: determining sampling plan feasibility, intervention refinement, and to ensure the study

instruments translated well into an online environment. The planned pilot study was to utilize a random national sample of women's health nurses, accessed by purchasing a United States Postal Service mailing list of nurses belonging to AWHONN, and sending them a postcard invitation to participate in the study. However, due to a poor response rate of approximately 3%, the study design was altered to include a non-probability sample of randomly assigned women's health nurses.

Population and Sample

The target population for the study was all women's health nurses in the US. The accessible population consisted of approximately 22,600 (InFocus Marketing, 2011) members of AWHONN who care for childbearing women and/or their newborns in the US. The population was accessible through the state (section) leaders of AWHONN.

The study had many inclusion criteria. Only RNs who care for childbearing women and/or their newborns in the US met the inclusion criteria. Other inclusion criteria included having: an email address; access to a computer with speakers and highspeed Internet, or an Internet connection which is not dial-up; and the basic computer skills of turning on a computer, accessing the Internet, sending email, using a computer mouse, and typing.

A sample of 180 (60 per group) was planned for the study as determined via G*Power 3 (Faul, Erdfelder, Lang, & Buchner, 2010) using the repeated measures multivariate analysis of variance (MANOVA) with a within-between interaction statistical test, a medium *f* effect size of .25 (Faul et al., 2010), alpha of .05, and a power of .8 as suggested by Cohen (1988), Polit and Beck (2008), and C. Cross (personal communication, June 30, 2010). The medium effect size estimate was appropriate

considering effect sizes from previous studies: Cohen's d = 0.7 (Jones, 2009) and Cohen's d = 0.4 (Schim et al., 2006b). While MANOVA was not to be the statistical test used for data analysis, it yields a more conservative estimate of need and thus ensures adequate sample size and enhances study power (C. Cross, personal communication, June 30, 2010). An additional 15% was added to the G*Power 3 estimate to obtain the given sample size, as attrition was expected to be less than 15%. Multiple previous studies of cultural competence interventions had no attrition (Brathwaite, 2005; Kelly & Papadopoulos, 2009; Jones, 2009), while Schim et al. (2006b) had approximately 15% attrition, Smith (1998) had 11% attrition, and Williams (2002) had only 7.7% attrition.

An e-mail was sent to the each state's AWHONN section leaders (see Appendix A); follow-up phone calls were made approximately one week later to the leaders whose phone numbers were publicly available (see Appendix B). The e-mail requested that the leaders distribute information regarding the study to the AWHONN members in their state. A PDF and JPEG invitation was attached for convenience (see Appendix C). A brief description of the study, a Website address, and a password for the Website was included. Interested individuals were instructed to visit the Website, built by the investigator, to obtain further information about the study and register if they desired to do so. The Website described the purpose of the study, the background and significance, what participation entailed, and that CEUs would be granted upon completion of one of the online courses. All who registered via the secure encrypted form on the Website were randomly assigned to one of the three groups and e-mailed a Web Survey link to the informed consent form. See Appendices D, E, and F for the full participant informed consent form for each group.

Ethical Considerations

There were no anticipated negative effects of participating in the study and there were no adverse effects of not participating or of withdrawing from the study at any time. Potential subjects were informed their information would be confidential and findings only reported by groups. Anonymity was fostered, but the investigator, the University of Nevada, Las Vegas (UNLV) CE coordinator, and the Nevada State Board of Nursing (NSBON) had access to each subject's true identity for the purpose of granting CEUs. However, subjects were permitted to choose to use a temporary e-mail address for course registration and had the option of choosing and using a pseudonym in WebCampus of which only the investigator had knowledge. Informed consent was obtained for the study following UNLV Institutional Review Board (IRB) approval. See Appendix G for original IRB approval notice and Appendix H for modification approval.

Procedures and Data Collection

The independent variable (between-groups variable) for hypothesis one was online cultural competence CE courses and had three levels: control, socially isolated, and socially interactive. Both the socially isolated and socially interactive online courses included four modules, offered one per week, over four consecutive weeks. Each module was two and a half hours long on average, totaling 10 hours for each course. Both courses covered the same content with the same learning objectives. See Table 1 for full details including a list of objectives with corresponding content, learning activities/assessments for the socially interactive group, learning activities/assessments for the socially isolated group, 3-D model (CCA) constructs, and ISFOLE constructs for each module. Table 1

Objectives, Learning Activities/Assessments, 3-D model (CCA) Constructs, and ISFOLE Constructs for Both Courses by Module

| Course Objective | 1 | 2 | | | | |
|-------------------------|------------------------------------|-------------------------------|--|--|--|--|
| Number | | | | | | |
| Actual Objective | describe the four components of | define words commonly | | | | |
| Subjects will: | the provider level of the 3-D | associated with culture and | | | | |
| | model. | cultural competence. | | | | |
| Content | - Introduction to the 3-D model | - Culture | | | | |
| | and its constructs | - Cultural competence | | | | |
| | - Visual representation of 3-D | - Associated words | | | | |
| | model | | | | | |
| Corresponding 3- | - Cultural Diversity | - Cultural Awareness | | | | |
| D model/CCA | - Cultural Awareness | | | | | |
| Construct(s) | - Cultural Sensitivity | | | | | |
| | - Cultural Competence | | | | | |
| | Behaviors | | | | | |
| Learning | LA: Multimedia presentation | LA: Crossword puzzle game, | | | | |
| Activities (LA) & | and/or review written text. Drag- | review written text, and/or | | | | |
| Assessment | and-drop game. | view multimedia presentation. | | | | |
| Techniques (AT)- | | | | | | |
| Socially Isolated | AT: Successful completion | AT: Successful completion of | | | | |
| | (defined as score of at least 80%) | matching quiz | | | | |
| | of fill-in-the-blank quiz | | | | | |
| Learning | LA: Multimedia presentation | LA: Crossword puzzle game, | | | | |
| Activities & | and/or review written text. Drag- | review written text, and/or | | | | |
| Assessment | and-drop game. Discussion. | view multimedia presentation. | | | | |
| Techniques- | | | | | | |
| Socially | AT: Discussion posting. | AT: Successful completion of | | | | |
| Interactive | | matching quiz. | | | | |
| Corresponding | - Motivation | - Motivation | | | | |
| ISFOLE | - Individual Differences | - Individual Differences | | | | |
| Construct(s) | - Social Learning for socially | | | | | |
| | interactive group | | | | | |

Module 1: Introduction to Culture and Cultural Competence

Module 2: Getting to Know Yourself

| Course Objective | 3 | 4 |
|-------------------------|---------------------------------|-------------------------------|
| Number | | |
| Actual Objective | reflect upon personal reactions | describe how a nurse's own |
| Subjects will: | to the views others have of | culture could impact the care |

| | American culture. | they provide to childbearing |
|-------------------------|------------------------------------|---------------------------------|
| | | women. |
| Content | - How American culture has | - Managing prejudices |
| | been viewed | - Ways in which prejudices |
| | - How it feels to be stereotyped | and personal culture/beliefs |
| | | impact nursing care |
| Corresponding 3- | - Cultural Sensitivity | - Cultural Sensitivity |
| D model/CCA | | _ |
| Construct(s) | | |
| Learning | LA: Review written text and/or | LA: View multimedia |
| Activities & | multimedia presentation (You | presentation (You Tube video) |
| Assessment | Tube video) on the Internet. | on managing personal |
| Techniques- | Reflective writing/journaling. | prejudices and/or review |
| Socially Isolated | | written text. Interpretations |
| | AT: Submission of reflective | exercise. |
| | writing/journal describing | |
| | personal reaction to the written | AT: Submission of |
| | text and/or multimedia | interpretations exercise. |
| | presentations viewed on the | |
| | Internet. | |
| Learning | LA: Review written text and/or | LA: View multimedia |
| Activities & | multimedia presentation (You | presentation (You Tube video) |
| Assessment | Tube video) on the Internet. | on managing personal |
| Techniques- | Reflective writing/journaling. | prejudices and/or review |
| Socially | | written text. Completion of |
| Interactive | AT: Discussion board posting | interpretations exercise and |
| | detailing personal reaction to the | discussion of exercise. |
| | written text and/or multimedia | |
| | presentations viewed on the | AT: Discussion posting |
| | Internet. | describing interpretations |
| | | exercise results and response |
| | | to one peer presenting at least |
| | | one other way in which part of |
| | | the interpretations exercise |
| | | could be viewed. |
| Corresponding | - Reflective Learning | - Reflective Learning |
| ISFOLE | - Social Learning for socially | - Social Learning for socially |
| Construct(s) | interactive group | interactive group |

| Course Objective | 5 | 6 |
|-------------------------|----------------------------------|---------------------------------|
| Number | | |
| Actual Objective | utilize resources to access both | recognize areas in which |
| Subjects will: | general information about | cultural groups often differ in |
| | different cultures and | general and with regard to |

| | information specific to | childbearing. |
|-------------------------|---|---------------------------------------|
| | childbearing. | |
| Content | - Review of some cultural | - Common areas of general |
| | groups' general | difference among cultural |
| | beliefs/practices. | groups. |
| | - Review of some cultural | - Common areas of difference |
| | groups' practices/beliefs | among cultural groups which |
| | which are unique to | are specific to childbearing. |
| | childbearing. | |
| Corresponding 3- | - Cultural Diversity | - Cultural awareness |
| D model/CCA | - Cultural Competence | |
| Construct(s) | Behaviors | |
| Learning | LA: Multimedia presentation | LA: Multimedia presentation, |
| Activities (LA) & | and Website review. | review of written text, and/or |
| Assessment | | Website review. |
| Techniques (AT)- | AT: Successful completion of a | |
| Socially Isolated | short quiz about material | AT: Submission of a written |
| | presented on the Websites | description of the three areas |
| | reviewed. | of difference, with examples, |
| | | which are most evident in your |
| | | individual nursing practice. |
| Learning | LA: Multimedia presentation | LA: Multimedia presentation, |
| Activities & | and Website search. Review of | review of written text, and/or |
| Assessment | peers' discussion postings. | Website search/review. |
| Techniques- | | Discussion. |
| Socially | AT: Find one additional | |
| Interactive | Website related to cultural | AT: Discussion board posting |
| | group(s) and post a description | describing the two areas of |
| | of the site which outlines its | difference, with examples, |
| | content. | which are most evident in your |
| | | individual nursing practice. |
| | | Substantial response to one |
| | | peer who discussed a difference |
| Companyording | Motivation | Motivation |
| Corresponding | - WOUVAUOII Social Learning for socially | - Mouvation Individual Differences |
| LOFULE Construct(s) | interactive group | - Individual Differences |
| | interactive group | - Social Learning for socially |
| | | interactive group |

Module 4: Taking Action

| Course Objective | 7 | 8 |
|-------------------------|--------------------------------|------------------------|
| Number | | |
| Actual Objective | gain experience performing a | analyze cross-cultural |
| Subjects will: | cultural assessment on someone | encounters focusing on |

| Content | of a different culture. General cultural assessment questions Cultural assessment questions specific to childbearing Existing cultural assessment instruments/tools | strengths and ways in which nursing practice could be improved when caring for childbearing women. - Cross-cultural nursing encounter analysis - Ways to improve nursing care in unique cross-cultural childbearing experiences |
|--|--|---|
| Corresponding 3- D model/CCA Construct(s) | - Cultural Competence Behaviors | - Cultural Awareness |
| Learning Activities (LA) & Assessment Techniques (AT)- Socially Isolated | LA: Application of skills through completing a cultural assessment of an individual (client, co-worker, etc.) who is of a different culture. AT: Submission of cultural assessment results. | LA: Review the two posted case study presentations about caring for a culturally diverse childbearing woman and reflect upon it. AT: Submission of a written description of what the nurse |
| | | in each scenario did well and how nursing care could be improved in each scenario |
| Learning Activities & Assessment Techniques- Socially Interactive | LA: Application of skills through completing a cultural assessment of an individual (client, co-worker, another learner in the course, etc.) who is of a different culture. | LA: Reflect upon a personal challenging experience working with a client from a different culture. Reflect upon a peer's experience caring for a client from a different culture. |
| | AT: Submission of cultural assessment results. | AT: Discussion board posting addressing what was done well and how care could be improved. Respond to at least one peer experience focusing on what was done well and how care could be improved |
| Corresponding ISFOLE Construct(s) | Active Learning Contextual Learning | Active Learning Reflective Learning Contextual Learning Social Learning for socially interactive group |

As shown in Table 1, the two courses used several varying learning activities and assessment techniques. For example, the socially isolated course met objective eight by reviewing two different case study presentations related to the care of a culturally diverse childbearing woman, submitting a written description of what each nurse did well, and how nursing care could be improved in each scenario through the assessment area of WebCampus. The socially interactive group met the same objective by posting a written description on the discussion board of a challenging experience they have had working with a client from a different culture, relating what they did well and how their care could be improved. They then read and responded to at least one peer experience focusing on what their peer did well and how care could be improved. Hence, both groups analyzed two case studies reflecting upon strengths and weaknesses.

Several measures were taken to maintain the socially isolated nature of the socially isolated group. First, the socially isolated course was designed such that subjects could not proceed to the subsequent module or the CCA post-test until: (a) the required activities/assessment techniques for the previous module were completed, and (b) it was the week in which the next module was scheduled to be available. This was done to ensure all required activities/assessments were completed without requiring the investigator to audit progression. Several WebCampus features were also disabled. Specifically, participants in the socially isolated group did not have access to the chat, discussion board, or e-mail functions. In addition, appropriate questions related to WebCampus were directed to the IT Help Desk. Individualized feedback was not provided by the investigator. Rather, feedback was automatically provided for quizzes via the assessment manager in WebCampus and for writing assignments via

announcement postings of general sample answers which opened at the end of the module. Therefore, participants in the socially isolated group were not in direct contact with each other and had minimal contact with the investigator.

Best design practices for online courses were employed for both courses. Each followed recommended design principles for page layout, content organization, text/fonts, color use, and graphics as presented in Chapter 2 (see Appendix I for table of recommended design principles). Both courses were reviewed by three on-line education experts, one of whom had extensive experience with cultural competence, and by an additional on-line educator who was a state AWHONN leader and certified in in-patient obstetrical nursing. Revisions were made to both courses based upon the feedback obtained from these experts. The courses were offered via UNLV's WebCampus and were approved by Campus Computing Services.

The dependent variable, level of cultural competence, was operationally defined by measurement with the CCA as designed by Schim et al. (2003). Permission to utilize the CCA was obtained (see Appendix J). The actual CCA was not modified, as it was designed to assess cultural competence as conceptually defined by the SMCCM which is equivalent to the provider level of the 3-D model. See Appendix K for the full list of CCA items.

A short form, including 13 items, of the MCSDS was added to the CCA after its initial development in response to criticism that self-report measures are not reliable due to social desirability. MCSDS scores have been said to not correlate with CCA scores (S. Schim, personal communication, July 10, 2010). However, as there was no specific published data, the MCSDS short form was used in the study to assess social desirability.

Random grouping ensured the groups were comparable at baseline, but MCSDS scores could have been used as a covariate if shown to significantly impact the dependent variable (C. Cross, personal communication, July 19, 2010). See Appendix K to review the 13 MCSDS items.

Demographic data was also collected to describe the sample. Specifically, age, region of US residence, years of nursing experience, nursing areas of work, years worked in specialty area, how majority of nursing time is spent, race/ethnicity, highest level of education completed, number of years of nursing practice, whether or not original nursing training was completed in the US, previous cultural diversity training and type, previously completed online education or training, comfort level using a computer, and the location of computer/internet access were included. Previous studies have indicated age, race, years of nursing experience, and ethnicity do not impact CCA scores, but previous cultural training and educational level do impact CCA (Doorenbos & Schim, 2004) as well as CAS and CCB scores (Schim et al., 2005). Additionally, online education and comfort level using a computer has been shown to impact completion of online CE (Shade & Barber, 2004) and nurses completing online CE have reported it is easier to learn at home than at work (Atack, 2003). While the primary purpose of collecting this data was to describe the sample, data could have served as covariates if found to significantly impact the dependent variable (C. Cross, personal communication, July 9, 2010). See Appendix K for a full list of demographic items.

Subjects were recruited and informed consent obtained as previously discussed. Subjects in all groups were asked to complete the CCA, MCSDS, and demographic questions at the beginning of the study period. The control group completed them via an

online survey and the two experimental groups via the assessment tool in WebCampus. Experimental and control group subjects were permitted to leave any or all items blank. All groups were asked to complete the CCA again at the end of the four week period during which the courses were active. Once again, subjects were permitted to leave any or all items blank. Results were maintained by the investigator and subjects were asked to enter the e-mail address with which they registered for the course to permit matching of pre- and post-test responses. Those completing a course received 10 CEUs from the UNLV School of Nursing. After the study concluded, control group subjects had the chance to complete the socially isolated online cultural competence CE course for CEUs.

Subjects in the experimental groups were asked to complete a course evaluation at the end of the course, via the survey assessment option in WebCampus, to provide their opinions and give feedback to guide future course revisions as appropriate. The course evaluation addressed all parts of the ISFOLE, course design, and satisfaction using a 5point Likert-type scale of strongly disagree to strongly agree. For example, "The course content was engaging" addressed motivation and "I would recommend this course to others" addressed satisfaction. Technical issues and suggestions were addressed by open response items such as "What was your favorite part of the course?"

The investigator developed the course evaluation tool and its items, as the literature review provided in Chapter 2 revealed no single existing, applicable tool. More specifically, no one existing tool was identified which addressed all of the above areas and was applicable to both socially interactive and socially isolated courses. Subjects were permitted to leave any or all course evaluation items blank. See Appendix L for a full listing of course evaluation items.

The investigator designed and conducted the courses and downloaded the CCA results, MCSDS results, demographic data, and course evaluations. Electronic data was maintained on an encrypted external computer drive by the investigator, who had sole access. The external drive will be secured in the office of Dr. Lori Candela and destroyed three years after the study's completion. Subjects' names, state of nursing licensure, and nursing license numbers were maintained by the investigator and submitted as requested to the UNLV CE Coordinator, who may in turn be asked to submit them to the NSBON, for the purpose of CEU records.

Data Analysis

Data analysis was completed using SPSS version 17.0. While data from Likert scales is truly ordinal, CCA and MCSDS data were treated as interval level for statistical purposes (Polit & Beck, 2008). Only subjects with pre- and post-test data were utilized in the analysis of hypothesis one.

Pre- and post-test designs are typically analyzed in one of three ways: one-way analysis of variance (ANOVA) on gain scores, analysis of covariance (ANCOVA) using pretest scores as the covariate, or via a mixed model repeated measures ANOVA (RM-ANOVA) with a between-groups variable (Dimitrov & Rumrill, 2003; Girden, 1992). RM-ANOVA with a between-groups variable was planned for use in this study because it is strongly suggested when there are repeated measures (Garson, 2009). The repeated measures test reduces error by controlling some of the variation between individuals, increasing power and decreasing the number of subjects needed (Munro, 2005) while providing better analysis because the trend from pre to post is analyzed (C. Cross, personal communication, June 30, 2010). ANCOVA using pretest scores as a covariate

was not planned for use due to potential power loss from the decreased degrees of freedom (Dimitrov & Rumrill, 2003), because the regression slopes of each pre- to posttest relationship must be homogenous, which is often not true with a control group (Gliner, Morgan, & Harmon, 2003), and because it would take either a large pre- to posttest change or a larger sample to get significant results (C. Cross, personal communication, June 30, 2010). The gain scores approach was not planned for use because it increases systematic bias which can lead to false results (C. Cross, personal communication, June 30, 2010).

Hypotheses two and three were analyzed using correlation statistics. Specifically, hypothesis two utilized Spearman's rho as highest level of education attained and previous cultural diversity training provide ordinal data. Hypothesis three was analyzed using Pearson's r, as both CCA and MCSDS scores are summed data which therefore approximate interval level data (Polit & Beck, 2008).

Descriptive statistics were included to depict the MCSDS, course evaluation, and demographic data. This basic data is included in the following chapter to provide information to the reader and increase study replication potential. Internal consistency reliability of the CCA scores for this sample was conducted using a Cronbach alpha reliability coefficient; in this study the CCA had a Cronbach alpha coefficient of .88, indicating good internal consistency reliability.

CHAPTER 5

FINDINGS OF THE STUDY

This chapter presents the findings of the study. Demographic information about the sample is presented. The results of the three hypotheses are also presented. The sixth and final chapter will provide a discussion of these results.

Response and Attrition Rates

The 249 individuals who registered for the study on the informational Website were randomly assigned to the three groups (socially interactive, socially isolated, and control) and sent a link to an online survey to complete the informed consent. Of those who registered for the study, 190 visited the Website and provided informed consent to participate in the study: 61 from group 1 (socially interactive online CE course group), 63 from group 2 (socially isolated online CE course group), and 66 from group 3 (control group). There were 133 individuals who completed the pre-test CCA items in the Week 1 Survey during the first week of the study; one did not complete the MCSDS or demographic items. In total, 77 participants from groups 1 and 2 completed the Week 1 Survey, the first task of both online courses. In spite of announcements for the start of each module, prompt e-mail or phone responses to questions or concerns, announcements regarding WebCampus maintenance, and granting requests for additional time to complete individual modules, only 21 individuals completed the socially interactive course and only 23 completed the socially isolated course. Thus, 93 participants completed the CCA items of both the Week 1 Survey pre-test and the Week 4 Survey post-test.

Demographic Data

As previously stated, 132 participants answered the demographic questions included in the Week 1 Survey. Twenty-three reported living in the northeast, 43 in the southeast, 13 in the midwest, 3 in the southwest, and 50 in the western region of the US. While the majority of participants were White/Caucasian/European American (82%), 5% reported being Black/African American/Negro, 5% reported being Asian, 4% Hispanic/Latino, and 2% American Indian/Alaska Native. The mean participant age was 47; participants reported having 21.5 years of nursing experience on average, and 17.4 years on average in their specialty area. Only three (2%) of participants completed their original nursing training outside of the US. The majority of participants reported their highest degree was a bachelor's degree (55%), 30% reported their highest degree was a graduate or professional degree, 14% held an associate's degree as their highest degree, and only 2% held a diploma as their highest degree. Participants reported working in many areas of the specialty, with most reporting working with in-patient intrapartal women (61%) or in-patient mothers and infants (46%). Most participants spend the majority of their work time providing direct patient care (67%), while 17% work mostly in education/academia, and 15% in a management role. The vast majority of participants (83%) had previous cultural diversity training with most having had one (32%) or two (21%) types of previous diversity training. Most participants (83%) reported feeling very comfortable when using a computer, 89% had undergone previous online education or training, and 67% had access to a computer with high-speed internet and speakers at both work and home. See Table 2 for full demographic and descriptive details of the 132 participants who completed the Week 1 Survey pre-test and the 93 participants who

finished the study by completing both the Week 1 Survey pre-test and the Week 4 Survey post-test.

Table 2

Demographic and Descriptive Information

| Characteristic | Completed Pre-test Only (N=132) | | | Com Pos | pleted Pre- and st-tests (N=93) | |
|---|------------------------------------|-------|-----------|------------|------------------------------------|-----------|
| | M | SD | Range | M | SD SD | Range |
| Age in Years | 47 | 1.14 | 24-72 | 47 | 1.18 | 24-72 |
| Years of Nursing Experience | 21.5 | 12.47 | 1-49 | 22.13 | 13.21 | 1-49 |
| Years in Specialty Area | 17.37 | 10.77 | 0.5-43 | 17 | 10.85 | 0.5-37 |
| Pre-Test CCA Score | 5.38 | 0.65 | 3.68-6.68 | 5.37 | 0.67 | 3.68-6.56 |
| Self-Reported Level of Cultural Competence | 4.11 | 0.78 | 1-5 | 4.02 | 0.85 | 1-5 |
| MCSDS Score | 7.87 | 2.35 | 2-12 | 7.88 | 2.21 | 2-11 |
| Number of Ethnic/Racial Groups Encountered | 5.17 | 1.38 | 1-8 | 5.1 | 1.42 | 1-8 |
| Number of Special Populations Encountered | 4.56 | 1.42 | 1-7 | 4.49 | 1.48 | 1-7 |

| | Completed Pre- | | Completed Pre | |
|---|-----------------------|------|----------------------|------------|
| | test Only | | and I | Post-tests |
| | п | % | п | % |
| Area of Current US Residence | | | | |
| Northeast | 23 | 17.4 | 17 | 18.3 |
| Midwest | 13 | 9.9 | 12 | 12.9 |
| Southeast | 43 | 32.6 | 32 | 34.4 |
| Southwest | 3 | 2.3 | 2 | 2.2 |
| West | 50 | 37.9 | 30 | 32.2 |
| Current Work Area Does/Does Not Include | | | | |
| Antepartal (in-patient) | | | | |
| Does | 38 | 28.8 | 22 | 23.7 |
| Does Not | 94 | 71.2 | 71 | 76.3 |
| Current Work Area Does/Does Not Include | | | | |
| Intrapartal/L&D (in-patient) | | | | |
| Does | 81 | 61.4 | 54 | 58.1 |
| Does Not | 51 | 38.6 | 39 | 41.9 |
| Current Work Area Does/Does Not Include | - | | | |
| Postpartal (in-patient) | | | | |
| Does | 43 | 32.6 | 30 | 32.3 |
| Does Not | 89 | 67.4 | 63 | 67.7 |
| | 07 | 57.1 | 00 | 0,., |

| Current Work Area Does/Does Not Include | | | | |
|---|---------------|------|----|------|
| Mother/Infant (in-patient) | 61 | 46.2 | 45 | 48.4 |
| Does | 71 | 53.8 | 48 | 51.6 |
| Does Not | | | | |
| Current Work Area Does/Does Not Include | | | | |
| Nursery/NICU (in-patient) | | | | |
| Does | 36 | 27.3 | 23 | 24.7 |
| Does Not | 96 | 72.7 | 70 | 75.3 |
| Current Work Area Does/Does Not Include | | | | |
| Out-patient/Ambulatory Care | | | | |
| Does | 9 | 6.8 | 4 | 4.3 |
| Does Not | 123 | 93.2 | 89 | 95.7 |
| Current Work Area Does/Does Not Include | | | | |
| Community Health | | | | |
| Does | 9 | 6.8 | 4 | 4.3 |
| Does Not | 123 | 93.2 | 89 | 95.7 |
| Current Work Area Does/Does Not Include | | | | |
| Other | | | | |
| Does | 22 | 167 | 16 | 17.2 |
| Does Not | 110 | 83.3 | 77 | 82.8 |
| How Majority of Work Time is Spent | 110 | 0010 | | 0210 |
| Direct Patient Care | 88 | 667 | 66 | 71 |
| Management | 20 | 15.2 | 13 | 14 |
| Education/Academia | 22 | 16.7 | 14 | 15.1 |
| Other | $\frac{-}{2}$ | 1.5 | 0 | 0 |
| Ethnicity | | | - | - |
| Hispanic/Latino | 5 | 3.8 | 3 | 3.2 |
| White/Caucasian/European American | 108 | 81.8 | 79 | 84.9 |
| Black/African American/Negro | 7 | 5.3 | 5 | 5.4 |
| American Indian/Alaska Native | 2 | 1.5 | 0 | 0 |
| Asian | 7 | 5.3 | 5 | 5.4 |
| Native Hawaiian/Pacific Islander | 0 | 0 | 0 | 0 |
| Arab American/Middle Eastern | 0 | 0 | 0 | 0 |
| Other | 3 | 2.3 | 1 | 1.1 |
| Highest Level of Education Completed | | | | |
| Diploma | 3 | 2.3 | 3 | 3.2 |
| Associate Degree | 18 | 13.6 | 13 | 14 |
| Bachelor's Degree | 72 | 54.5 | 50 | 53.8 |
| Graduate or Professional Degree | 39 | 29.5 | 27 | 29 |
| Previously Participated in Cultural Diversity | | | | |
| Training | | | | |
| Yes | 109 | 82.6 | 77 | 82.8 |
| No | 23 | 17.4 | 16 | 17.2 |
| Number of Types of Previous Diversity | | | | |
| Training | | | | |
| 0 | 23 | 17.4 | 16 | 17.2 |
| 1 | 42 | 31.8 | 26 | 28 |
| 2 | 28 | 21.2 | 23 | 24.7 |
| 3 | 19 | 14.4 | 16 | 17.2 |

| Λ | 10 | 0.1 | 0 | 86 | |
|---|-----|------|----|------|--|
| 4 | 12 | 9.1 | 0 | 8.0 | |
| 5 | 6 | 4.5 | 3 | 3.2 | |
| 6 | 1 | 0.8 | 1 | 1.1 | |
| 7 | 1 | 0.8 | 0 | 0 | |
| Previously Completed Any Education or | | | | | |
| Training Online | | | | | |
| No | 15 | 11.4 | 11 | 11.8 | |
| Yes | 117 | 88.6 | 82 | 88.2 | |
| Comfort Level When Using a Computer | | | | | |
| Not At All | 0 | 0 | 0 | 0 | |
| Somewhat | 23 | 17.4 | 14 | 15.1 | |
| Very | 109 | 82.6 | 79 | 84.9 | |
| Location of Access to Computer With High- | | | | | |
| speed Internet and Speakers | | | | | |
| Home Only | 28 | 21.2 | 22 | 23.7 | |
| Home and Work | 89 | 67.4 | 61 | 65.6 | |
| Home and Other | 0 | 0 | 0 | 0 | |
| Home, Work, and Other | 4 | 3 | 2 | 2.2 | |
| Other Only | 1 | 0.8 | 0 | 0 | |
| Work Only | 8 | 6.1 | 6 | 6.5 | |
| Work and Other | 1 | 0.8 | 1 | 1.1 | |

* = *p* < .05

Hypothesis One

The first hypothesis addressed the effect of the two different online cultural competence continuing education courses on the cultural competence level of women's health nurses. CCA scores were calculated from the Week 1 Survey at the beginning of the study (pre-test) for the participants in the three groups and again at the end of the study from the Week 4 Survey (post-test). Unfortunately, the 93 participants who completed the study were not divided equally amongst the three study groups. Therefore, the 21 participants in group 1 and 22 participants in group 2 who completed the course/study were used for analysis. To balance the design, 22 participants were randomly selected from group 3 for inclusion in the analysis.

It was planned that a RM-ANOVA with a between-groups variable would be used to assess the impact of the two different interventions (socially interactive course, socially isolated course) on CCA scores. The assumptions of normal distribution, homogeneity of variances, and homogeneity of inter-correlations were met by group for the individuals who completed the study. The interaction effect, of primary interest, was not significant, Wilks Lambda = 0.91, F(2, 62) = 3.00, p = .06, observed power = .56. A main effect was noted for time, Wilks Lambda = 0.87, F(1, 62) = 9.34, p = .003, observed power = .85. This suggests a change in CCA scores from pre- to post-test. The main effect for group was not significant, F(2, 62) = 1.67, p = .2, observed power = .34. This insignificant finding suggests no difference in the effectiveness of the two online courses. However, non-significant results are commonly found due to insufficient power when group size is small (Pallant, 2007, Stevens, 2002).

Hypothesis one was therefore evaluated by ANCOVA on post-test CCA scores using pre-test CCA scores as a covariate. ANCOVA is more powerful than RM-ANOVA with a between-groups variable when there is a large effect size and homogeneity of regression slopes (Gliner et al., 2003). A one-way between groups ANCOVA was conducted to compare the effectiveness of the two interventions (socially isolated course, socially interactive course) at increasing cultural competence. Preliminary checks assured the assumptions of normality, linearity, homogeneity of variances, homogeneity of regression slopes, and reliable measurement of the covariates. After adjusting for pretest CCA scores, a significant difference was found between the three groups (socially interactive, socially isolated, and control) on post-test CCA scores, F(2, 61) = 5.20, p =.008, observed power = .81, partial $\eta^2 = .146$, indicating a large effect size. Pairwise

comparisons with a Bonferroni correction indicated that after controlling for pre-test CCA scores, the socially isolated group (M = 5.85) scored significantly higher (p = .006) on the post-test CCA than the control group (M = 5.36). No other significant comparisons were identified, indicating that after controlling for pre-test CCA scores the socially interactive group did not score significantly higher on the post-test than the control group and the socially interactive group did not score significantly higher than the socially isolated group.

Hypothesis Two

Hypothesis two related CCA scores to demographic data such as highest level of education completed and previous cultural diversity training obtained from the Week 1 Survey (n = 132). The relationship between cultural competence (as measured by the CCA) and highest level of education completed was investigated using Spearman rank order correlation. Preliminary analysis identified the three participants whose highest degree earned was a diploma to be outliers. However, as Spearman rank order correlation is non-parametric, it can be used when there are outliers (Polit, 2010). No significant correlation was found between CCA scores and highest level of education attained ($r_s = -.024$, n = 132, p = .786). A Kruskal-Wallis test confirmed this finding ($\chi^2 = .83$, p = .842).

Hypothesis two also explored the relationship between cultural competence (as measured by the CCA) and previous diversity training. Preliminary analysis revealed previous training as a yes/no question was skewed (skewness = -1.74). However, non-parametric statistics can be used when data is skewed (Polit, 2010). No relationship was found between CCA scores and previous diversity training as a yes/no question using

Spearman rank order correlation ($r_s = .021$, n = 132, p = .814). A Mann-Whitney U test confirmed this finding (U = 1214, p = .813). However, a small significant positive correlation was found between number of types of previous cultural diversity training and CCA scores ($r_s = .173$, n = 132, p = .047), indicating having more types of previous cultural diversity training was associated with higher CCA scores at baseline.

Hypothesis Three

The third and final hypothesis sought to support that cultural competence (as measured by the CCA) is positively correlated with social desirability (as measured by the MCSDS). The assumptions of related pairs, independent observations, normality, and linearity were met, so the Pearson product-moment correlation coefficient was used. A small positive correlation was noted between pre-test CCA and MCSDS scores (r = .219, n = 132, p = .012) with higher MCSDS scores being associated with higher CCA scores.

Attrition

As evidenced previously, both experimental groups (groups 1 and 2) had high attrition rates. Specifically, 42% of participants who completed the Week 1 Survey pretest to begin the socially interactive course did not complete the course/study and 44% of participants who completed the Week 1 Survey pre-test to begin the socially isolated course did not complete the course/study. To explore this attrition, Chi-square tests of independence were conducted to evaluate the relationship between course/study completion (yes/no) and: (a) areas of specialty in which the participants currently work, (b) how majority of work time is spent, (c) ethnicity, (d) level of educational attainment, (e) location of original nursing training, (f) previous diversity training (yes/no), (g) number of types of previous diversity training, (h) previous online education/training, (i)
comfort level using a computer, (j) location of computer with high-speed internet and speakers access, (k) self-reported cultural competence level, and (l) area of current US residence. The likelihood ratio (LR) was used when expected cell frequency was less than five. Independent t-tests were used to assess differences between those who completed the course/study (yes/no) for the following continuous variables: (a) age, (b) years of nursing experience, (c) years in specialty area of nursing, (d) pre-test CCA scores, (e) MCSDS scores, (f) number of ethnic/racial groups encountered at work in the past 12 months, and (g) the number of special population groups encountered at work in the past 12 months. Three significant relationships were found. While the relationship between course/study completion and area of current US residence violated the Chisquare assumption for minimum expected cell frequency, the LR (4, n = 77) = 12.41, p =.015, Cramer's V = .387, indicating a significant medium effect size. Another significant relationship was found between course/study completion and self-reported pre-test level of cultural competence for which the Chi-square test of independence violated the minimum expected cell frequency. However, the LR (4, n = 77) = 10.20, p = .037,Cramer's V = .314, indicating a medium effect size. The final significant relationship was found between course/study completion and ethnicity. While the minimum expected cell frequency assumption for the Chi-square test was violated, the LR (5, n = 77) =12.12, p = .033, Cramer's V = .346, indicating a medium effect size. See Table 3 for full details.

Table 3

| Comparison of Participant Completion and Non-Completion of Courses/Study | V |
|--|---|
|--|---|

| Characteristic | Non- | | Completers | | Statistic |
|--|------------|-------|------------|-------|-----------|
| | Completers | | | | |
| | M | SD | M | SD | |
| Age in Years | 45.91 | 1.02 | 47.23 | 1.11 | t = -0.54 |
| Years of Nursing Experience | 19.18 | 11.12 | 21.95 | 12.85 | t = -0.99 |
| Years in Specialty Area | 16.62 | 10.15 | 17.16 | 10.51 | t = -0.23 |
| Pre-test CCA Score | 5.67 | 0.61 | 5.4 | 0.7 | t = -0.20 |
| MCSDS Score | 7.62 | 2.65 | 8.12 | 2.08 | t = -0.92 |
| Number of Racial/Ethnic Groups Encountered | 5.38 | 1.39 | 5.02 | 1.3 | t = 1.17 |
| Number of Special Population Groups Encountered | 4.76 | 1.23 | 4.33 | 1.36 | t = 1.44 |

| Characteristic | Non- | | Completers | | Statistic |
|---------------------------------------|------|--------------|------------|------------|------------------|
| | Com | pleters | | | |
| | N | % | N | % | |
| Area of Current US Residence | | | | | |
| Northeast | 4 | 5.2 | 10 | 13 | LR=12.41* |
| Midwest | 1 | 1.3 | 7 | 9.1 | |
| Southeast | 8 | 10.4 | 16 | 20.8 | |
| Southwest | 1 | 1.3 | 0 | 0 | |
| West | 19 | 24.7 | 11 | 14.3 | |
| Self-Reported Level of Cultural | | | | | |
| Competence | | | | | |
| Very Incompetent | 0 | 0 | 1 | 1.3 | LR=10.20* |
| Somewhat Incompetent | Ő | Ő | 6 | 7.8 | 211 10.20 |
| Neither Competent Nor | 2 | 2.6 | 1 | 1.3 | |
| Incompetent | _ | | - | | |
| Somewhat Competent | 20 | 26 | 26 | 33.8 | |
| Very Competent | 12 | 15.6 | 9 | 11.7 | |
| Current Work Area Does/Does Not | | | | | |
| Include Antepartal (in-patient) | | | | | |
| Does | 12 | 15.6 | 10 | 13 | $x^2 = 1.72$ |
| Does Not | 21 | 27.3 | 34 | 13 44 2 | $\lambda = 1.72$ |
| Current Work Area Does/Does Not | 21 | 21.5 | 54 | <i>-</i> | |
| Include Intranartal/I &D (in-patient) | | | | | |
| Does | 21 | 77.2 | 22 | 20.0 | $u^2 = 0.00$ |
| Does Not | 21 | 27.3 15.6 | 25 | 29.9 | χ 0.99 |
| Comment World Arres Dates (Dates Not | 12 | 15.0 | 21 | 27.3 | |
| Current Work Area Does/Does Not | | | | | |
| Include Postpartal (in-patient) | | | | | |
| Does Not | 12 | 15.6 | 14 | 18.2 | $\chi^2 = 0.17$ |
| Does Not | 21 | 27.3 | 30 | 39 | |

| Current Work Area Does/Does Not Include Mother/Infant (in-patient) | | | | | |
|---|----|------|----|------|---------------------------|
| Does | 13 | 16.9 | 20 | 26 | $\gamma^2 = 0.28$ |
| Does Not | 20 | 26 | 24 | 31.2 | λ $\sim 10^{-10}$ |
| Current Work Area Does/Does Not | | | | | |
| Include Nursery/NICU (in-patient) | | | | | |
| Does | 11 | 14.3 | 13 | 16.9 | $\gamma^2 = 0.123$ |
| Does Not | 22 | 28.6 | 31 | 40.3 | <i>7</i> 0 |
| Current Work Area Does/Does Not | | | | | |
| Include Out-patient/Ambulatory Care | | | | | |
| Does | 4 | 5.2 | 2 | 2.6 | LR = 1.5 |
| Does Not | 29 | 37.7 | 42 | 54.5 | |
| Current Work Area Does/Does Not | | | | | |
| Include Community Health | | | | | |
| Does | 5 | 6.5 | 2 | 2.6 | LR = 2.57 |
| Does Not | 28 | 36.4 | 42 | 54.5 | |
| Current Work Area Does/Does Not | | | | | |
| Include Other | | | | | |
| Does | 5 | 6.5 | 10 | 13 | $\gamma^2 = 0.69$ |
| Does Not | 28 | 36.4 | 34 | 44.2 | λ |
| How Majority of Work Time is Spent | - | | - | | |
| Direct Patient Care | 19 | 24.7 | 28 | 36.4 | |
| Education/Academia | 6 | 7.8 | 9 | 11.7 | LR = 2.15 |
| Management | 7 | 9.1 | 7 | 9.1 | 2 |
| Other | 1 | 1.3 | 0 | 0 | |
| Ethnicity | | | | | |
| Hispanic/Latino | 2 | 2.6 | 1 | 1.3 | LR=12.12* |
| White/Caucasian/European | 26 | 33.8 | 38 | 49.4 | |
| American | | | | | |
| Black/African American/Negro | 0 | 0 | 4 | 5.2 | |
| American Indian/Alaska Native | 2 | 2.6 | 0 | 0 | |
| Asian | 2 | 2.6 | 0 | 0 | |
| Native Hawaiian/Pacific Islander | 0 | 0 | 0 | 0 | |
| Arab American/Middle Eastern | 0 | 0 | 0 | 0 | |
| Other | 1 | 1.3 | 1 | 1.3 | |
| Highest Level of Education | | | | | |
| Completed | | | | | |
| Diploma | 0 | 0 | 1 | 1.3 | LR = 1.93 |
| Associate Degree | 5 | 6.5 | 6 | 7.8 | |
| Bachelor Degree | 21 | 27.3 | 24 | 31.2 | |
| Graduate or Professional Degree | 7 | 9.1 | 13 | 16.9 | |
| Previously Participated in Cultural Diversity Training | | | | | |
| Yes | 29 | 37.7 | 38 | 49.4 | LR = 0.04 |
| No | 4 | 5.2 | 6 | 7.8 | |
| Number of Types of Previous Diversity Training | | | | | |
| 0 | 4 | 5.2 | 6 | 7.8 | LR = 6.18 |
| 1 | 16 | 20.8 | 14 | 18.2 | |

| 2 | 4 | 5.2 | 12 | 15.6 | |
|------------------------------------|----|------|----|------|-----------------|
| 3 | 3 | 3.9 | 7 | 9.1 | |
| 4 | 4 | 5.2 | 4 | 5.2 | |
| 5 | 1 | 1.3 | 1 | 1.3 | |
| 6 | 0 | 0 | 0 | 0 | |
| 7 | 1 | 1.3 | 0 | 0 | |
| Previously Completed Any Education | | | | | |
| or Training Online | | | | | |
| No | 3 | 3.9 | 7 | 9.1 | LR = 0.80 |
| Yes | 30 | 39 | 37 | 48.1 | |
| Comfort Level When Using a | | | | | |
| Computer | | | | | |
| Not At All | 0 | 0 | 0 | 0 | $\chi^2 = 1.43$ |
| Somewhat | 8 | 10.4 | 6 | 7.83 | <i>,</i> , |
| Very | 25 | 32.5 | 38 | 49.4 | |
| Location of Computer With High- | | | | | |
| speed Internet and Speakers Access | | | | | |
| Home Only | 6 | 7.8 | 11 | 14.3 | LR = 5.72 |
| Home and Work | 22 | 28.6 | 29 | 37.7 | |
| Home and Other | 0 | 0 | 0 | 0 | |
| Home, Work, and Other | 2 | 2.6 | 0 | 0 | |
| Other Only | 1 | 1.3 | 0 | 0 | |
| Work Only | 2 | 2.6 | 4 | 5.2 | |
| Work and Other | 0 | 0 | 0 | 0 | |
| | | | | | |

* = *p* < .05

In addition to these findings, 11 participants (33% of those who withdrew after completing the Week 1 Survey and beginning the study/course) e-mailed the investigator and offered explanations as to why they were withdrawing from the course/study. All 11 participants indicated they were withdrawing due to time, technology/computer trouble, or both. Specifically, five cited only time as a reason, three cited only computer/technology issues, one indicated both time constraints and not wanting to change computer settings, one cited time and up-coming vacation, and one cited poor technology skills which resulted in the course taking too much time. While this information was not solicited, it provides insight into reasons why participants withdrew.

Summary

This fifth chapter presented the findings for the study. Demographic and descriptive information was given for the 132 participants who completed the Week 1 Survey pre-test and for the 93 participants who completed both the Week 1 Survey pretest and the Week 4 Survey post-test. Data analysis results were detailed for each of the three study hypotheses. Hypothesis one was partially supported using ANCOVA with pre-test CCA scores as a covariate. Specifically, after controlling for pre-test scores, the post-test CCA scores for group 2 (socially isolated intervention) were found to be significantly different from group 3 (control); no other significant differences were identified. Hypothesis two was partially supported and partially rejected, as a significant correlation was found between CCA scores and number of types of previous diversity training, but not between CCA scores and highest level of education completed. Hypothesis three was supported by statistically significant findings. Finally, the attrition rate was explored and relationships were found between completion of the course/study (yes/no) and: (a) area of residence in the US, (b) ethnicity, and (c) self-reported cultural competence level. The sixth and final chapter will further explore these findings and their implications.

CHAPTER 6

SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS

This sixth and final chapter includes an overview of the study. Study findings and limitations are discussed. Implications for nursing and nursing education are also highlighted and recommendations for future research provided.

Brief Overview of Study

The purpose of this study was to evaluate the effect of two online CE interventions on the cultural competence of nurses who care for childbearing women and/or newborns. It was anticipated there would be a greater increase in the cultural competence of the women's health nurses in the socially interactive online CE course compared to those in the socially isolated course, and both groups would have significant increases in cultural competence compared to the control group. It was also anticipated educational level and previous cultural diversity training would be related to cultural competence, and that social desirability would be positively correlated with cultural competence.

The framework for the study included two existing models: the 3-dimensional puzzle model of culturally congruent care and the ISFOLE. Review of the literature revealed numerous theories and tools which have been used to describe and measure cultural competence, and face-to-face CE interventions have successfully increased the cultural competence of healthcare professionals. As face-to-face CE isn't always practical, online CE has been used with various topics for healthcare providers, but few online cultural competence CE sources exist for nurses. Two socially interactive graduate courses and one self-study course were identified. Two socially isolated

publically accessible courses were also noted (Gerace & Salimbene, n.d.; Office of Minority Health, 2007); unfortunately, neither uses an educational theory, only one employs a cultural competence theory, the material is mainly factual, and no conclusive evidence supports their effectiveness. A comprehensive review of the literature revealed no studies which have compared socially interactive online CE to socially isolated.

This study used an experimental pre- and post-test design with a control group and two different four-week, 10 hour, online cultural competence interventions: a socially interactive CE course and a socially isolated one. Demographic data was collected, the MCSDS used to measure social desirability, and the CCA used to measure cultural competence. Course evaluation questions were asked of those who completed a course.

Discussion of the Findings

The findings of the study are interpreted in this section. First, participants' demographic information is compared to nurses and AWHONN members in the US. Next, the three hypotheses are discussed individually with conclusions and possible explanations provided. Finally, attrition and course evaluation data is briefly described and discussed.

Demographic Information

Study participants were asked to complete a demographic questionnaire as part of the Week 1 Survey pre-test. See Appendix K for a full list of demographic questions and Table 2 for detailed demographic findings. This section will compare the demographic information for this study's sample to the entire population of AWHONN members as reported by InFocus Marketing (2011), and to all nurses in the US as described by the US Department of Health and Human Services, Health Resources and Services

Administration (Health Resources and Services Administration [HRSA], 2010), to the extent possible.

The US has approximately 3,063,000 licensed RNs (HRSA, 2010). Approximately 22,600 of these nurses are members of AWHONN (InFocus Marketing, 2011), indicating they specialize in women's health, obstetric, and neonatal nursing in the US. About 4,700 AWHONN members (21%) live in the northeast section of the US, 4,700 (21%) in the midwest, 5,500 (24%) in the southeast, 3,300 (15%) in the southwest, and 4,400 (19%) in the west. Individuals in this study's sample who completed the Week 1 Survey pre-test (n = 132) lived primarily in the west (38%), southeast (33%), northeast (17%), midwest (10%), and southwest (2%). Of those who completed the study (n = 93), about 34% lived in the southeast, 32% in the west, 18% in the northeast, 13% in the midwest, and 2% in the southwest. Therefore, while the study represented the entire US, in general the west and southeast were overrepresented, the northeast slightly underrepresented, and the midwest and southwest underrepresented. One potential explanation for this discrepancy is related to the sampling method. Each state's AWHONN leaders were contacted and asked to distribute information about the study to members in their state. Unfortunately, there is no way to know how many state leaders acted upon this request, as a response to the request to distribute information about the study was never received from many states. However, leaders from seven states responded to the investigator via e-mail or phone that they would not distribute the information to members; reasons offered included not enough time to do it, being out on extended leave for illness, and members in the state do not like to receive e-mail communications. Therefore, members in all states were not afforded the opportunity to

participate, impacting the distribution of participants across the US. For example, two of the four states in the southwest region, the least represented area in the study population, stated they would not be distributing the information to members in their state.

Participants were also asked about their age and ethnicity. The average age of an RN in the US is 47 years old (HRSA, 2010). The average age of participants in this study was also 47 years old, matching the national average. In the US, approximately 5% of RNs self-identify as being Black or African American; 6% as Asian, Native Hawaiian, or Pacific Islander; 4% Hispanic or Latino; and only 0.3% as American Indian or Alaska Native (HRSA). While this demographic information is not available for AWHONN members, the pre-test completers (n = 132) in this study aligned well with national averages: 5% self-reported being Black or African American, 5% Asian, 4% Hispanic, 2% American Indian or Alaska Native, and 82% White or Caucasian. Those who completed the study (n = 93) also represented the US RN demographics well, with the exception that the American Indian or Native Alaskan population was not represented. The ethnic composition or average age of women's health nurses in the US is not known.

There are two other demographic items for which this study's sample can be compared to the population of US nurses. The first is highest level of education completed. According to the latest national HRSA survey of RNs (HRSA, 2010), the majority of RNs in the US hold bachelor degrees (37%), followed by associate degrees (36%), diplomas (14%), and master's degrees or higher (13%). Nurses with graduate or bachelor's degrees were overrepresented in this study's sample, while diploma and associate degree nurses were underrepresented. There could be several reasons for this. First, nurses with higher nursing degrees complete more CE per year than those with

lower nursing degrees (Kubsch, Henniges, Lorenzoni, Eckardt, & Oleniczak, 2003) and there is a significant relationship between highest academic degree and participation in CE (Beatty, 2001). Advanced practice nurses often need CE to maintain their certification or for licensure (Nalle, Wyatt, & Myers, 2010). Furthermore, professional association membership, such as membership in AWHONN, has a significant relationship with CE; Smith (2004) found nurses belonging to one association accumulate on average 10 hours more of CE per year than those not belonging to an association. Finally, nurses with higher degrees may have a greater understanding of research and place more emphasis on it than those with less formal education. Therefore, associate degree and diploma nurses may have been underrepresented in this study because of the accessible population (AWHONN members) or because the study explored CE.

The other area which can be compared is the country of initial nursing training. In the US, approximately 5% of RNs completed their initial nursing training abroad (HRSA, 2010). In this study 2% of participants completed their initial training abroad. Unfortunately, data about location of initial nursing training does not exist for women's health nurses in the US.

The final demographic item for which participants in this study can be compared to the entire AWHONN membership is how the majority of work time is spent. Individuals in this study's sample who completed the Week 1 Survey pre-test (n = 132) primarily spent their work time in direct patient care (67%), followed by education/academia (17%), management (15%), and in other areas (2%). Of those who completed the study (n = 93), the majority primarily spent their work time in direct patient care (71%), followed by education/academia (15%), and management (14%). These percentages are fairly consistent with AWHONN members nationally. The majority of AWHONN members in the US primarily spend their work time in direct patient care (61%), followed by 19% in management, 18% in education/academia, and only 2% in other areas (InFocus Marketing, 2011).

Hypothesis One

Hypothesis one stated there would be a change in the cultural competence of the women's health nurses included in the study. Specifically, it was thought there would be a greater increase in the CCA scores of nurses completing the socially interactive intervention compared to those completing the socially isolated intervention and that both groups would have CCA score increases which differed significantly from the control group. ANCOVA analysis using pre-test CCA scores as a covariate revealed a significant difference between the three groups (socially interactive, socially isolated, and control) on post-test CCA scores. Pairwise comparisons with a Bonferroni correction indicated that after controlling for pre-test CCA scores, the socially isolated group scored significantly higher on the post-test CCA than the control group. No other significant comparisons were identified, indicating that after controlling for pre-test CCA scores the socially interactive group did not score significantly higher than the control group and the socially interactive group did not score significantly higher than the socially isolated group.

In accord with the findings of this study, several previous studies of healthcare professionals have revealed significant changes from socially isolated online CE courses. Many isolated courses have been shown to increase knowledge about various topics other than cultural competence. Specifically, significant knowledge increases have been noted

in staff nurses (Schneiderman & Corbridge, 2009), healthcare providers (MacDonald & Walton, 2007), and physicians (Casebeer et al., 2004; Pelayo, Cebrian, Areosa, Agra, Izquierdo, & Buendia, 2011). Research has also shown socially isolated online CE courses can impact the practice (as evidenced by self-report of change or intent to change) of nurses (Arnella, Yox, Eckstein, & Ousley, 2010), doctors and nurses (Dunet et al., 2008), and physicians (Casebeer et al., 2004; Casebeer et al., 2008). Finally, MacDonald & Walton (2007) found socially isolated online CE increased the skills of healthcare providers. This study confirms socially isolated online CE can increase cultural competence and therefore, cultural awareness (knowledge), sensitivity (attitude), and competence behaviors (skills and actions).

Contrary to hypothesis one, the socially interactive online CE course in this study did not result in significantly greater increases in cultural competence than the socially isolated online CE course. The socially interactive course also did not result in a significantly greater cultural competence increase than the absence of an intervention (control group). These findings indicate socially interactive online CE courses are not more effective than socially isolated courses. In fact, the socially interactive course was not significantly more effective than not completing a course at all. There could be several explanations for these findings.

Social interaction has been shown to be important in online education. A review of 14 studies of nursing distance education concluded learning entails values and contributions of other learners, communication, interaction, and faculty support (Mancuso-Murphy, 2007). A meta-analysis of 51 studies revealed instructor involvement has a great impact on student outcomes (Zhao et al., 2005). There has also been a

positive correlation between classroom community, connectedness, and learning (Rovai, 2002). Finally, students who receive personal feedback perform better and are more satisfied than those who receive collective feedback (Gallien & Oomen-Early, 2008). While each of these studies supports social interaction, each of them explored the impact of social interaction in formal college for-credit coursework. Therefore, social interaction may not be as important in online CE as it is in formal online college coursework. In fact, this study lends support to Anderson's (2008) stance that the forms of student interaction within an online course (student-content, student-student, and student-instructor) can be substituted for one another, or even eliminated, as long as one of the three is included at a very high level.

Another explanation for the socially interactive group's findings is related to expectations about online CE. Few studies have been conducted of socially interactive online CE for healthcare providers (Huckstadt & Hayes, 2005; Luconi, 2008; Pullen, 2006; Spice, Palacios, Biondo, & Hagen, 2011) and review of the literature revealed no socially interactive online CE cultural competence courses which were socially interactive. The course evaluation comments for this study and from the socially interactive study Spice et al. (2011) conducted indicated the course was not what some participants expected. Therefore, participant expectations of online CE and attitude toward interaction in online CE may have impacted the outcome of the two courses.

A third potential explanation for the findings of the socially interactive group in this study is related to course length. The formal college courses previously described were all significantly longer than the 10 hour course utilized in this study. Other studies have also found longer socially interactive interventions increase knowledge. Huckstadt

& Hayes (2005) found a socially interactive course which could be taken for one to three college credit hours or for CEUs led to a significant change in the knowledge level of advanced practice nurses and graduate nursing students about back pain and dermatology. Ferguson and DeFelice (2010) conducted a study of graduate students and found those who took a semester-long course perceived they learned more than students who took an equivalent five-week summer course. Luconi (2008) found a nine month CE course increased the knowledge of eight doctors. Furthermore, Spice et al. (2011) found the 10 physician residents who took a course which took an estimated 48-72 hours to complete felt their knowledge and motivation had increased. Finally, Pena-Shaff, Altman, and Stephenson (2005) found students in interactive online courses are cautious of and anxious about what they post early in a course, not becoming comfortable with online discussions until late in the semester. It is therefore possible that only socially interactive CE courses that are longer in length lead to significant changes in participant knowledge, skills, and attitudes.

A final potential explanation for this study's findings is related to autonomy. Autonomy in nursing includes having the freedom to make decisions, control over professional practice, self-regulation, independence, and accountability (Gagnon, Bakker, Montgomery, & Palkovits, 2010). Similarly, time management, self-motivation (Beaudoin, Kurtz, & Eden, 2009; Stanford-Bowers, 2008), self-discipline (Stanford-Bowers, 2008), self-reliance, and high internal locus of control (Morris & Finnegan, 2008) have all been shown to impact online course success. As obstetrical nurses are highly autonomous (Kramer et al., 2007), and belonging to professional nursing organizations increases autonomy (Guerrieri, 2010), it is not surprising that the mean

CCA score of the nurses in both the socially interactive and isolated online courses increased (from M = 5.38 to M = 5.62 and from M = 5.42 to M = 5.87 respectively) while the mean score of the random sample of nurses from the control group did not (M= 5.34 both times). Furthermore, the group in the socially isolated course, which requires more autonomous learning than the socially interactive course, increased their cultural competence level significantly when compared to the control group, but the socially interactive group did not. Hence, the findings of hypothesis one could be related to autonomy.

Hypothesis Two

Hypothesis two stated that among women's health nurses, CCA scores are related to demographic variables such as highest level of education completed or previous cultural diversity training. Highest level of education completed and previous cultural diversity training as a yes/no question were not correlated with CCA scores. However, number of types of previous cultural diversity training was correlated with CCA scores.

Two previous studies have suggested CCA scores are related to highest level of education completed (Doorenbos & Schim, 2004; Schim et al., 2005). Additionally, one study has indicated a significant relationship between education and CAS scores (Schim et al., 2006a). This study did not confirm this relationship. This is likely due to differences in the samples evaluated. Schim et al. (2006a) studied hospice nurses and their sample differed in level of educational completion from the current study; none of their nurses held a nursing diploma as their highest degree, their sample had many more associate degree prepared nurses compared to this study (49% versus 14% respectively), fewer bachelor's prepared nurses (29% versus 54% respectively), and fewer graduate

degrees (21% versus 29%). Furthermore, in the Doorenbos and Schim (2004) study of hospice providers, the sample was only 40% nurses, and nearly 20% of their sample had not completed any education after high school. Schim et al. (2005) studied hospital providers, mainly nurses; but clerical workers, nutritionists, and therapists were also included. Once again, a substantial portion of the sample (17%) held only a high school degree. These differences in level of educational attainment and profession likely explain the divergent findings in this study.

Previous studies have also suggested a relationship between previous cultural diversity training (yes versus no) and CCA scores (Doorenbos & Schim, 2004; Doorenbos et al., 2005; Schim et al., 2005). This relationship was not confirmed in the current study. However, 83% of the participants in the current study reported having completed previous cultural diversity training compared to 73% and 41% in the studies by Doorenbos and Schim (2004), and by Schim et al. (2005), respectively. As cultural competence training becomes more common and more professionals have been exposed to some previous training, the simple yes/no question may become obsolete. This study found a significant positive correlation between number of types of previous cultural diversity training and CCA scores, indicating having exposure to more types of training leads to greater cultural competence. Unfortunately, these findings cannot be compared because previous studies have not reported the number of types of previous diversity training completed. However, existing studies have indicated that having any previous training is directly related to having greater cultural competence. Thus it is logical to presume having more types of previous diversity training, and therefore having had more

repetition and opportunities to assimilate new information with old (Vandeveer, 2009), leads to greater cultural competence scores.

Hypothesis Three

Hypothesis three stated MCSDS scores would be positively correlated with the CCA scores of women's health nurses. Although it has been stated CCA scores do not correlate with MCSDS scores (S. Schim, personal communication, July 10, 2010), no specific published data supports this claim. In this study, MCSDS scores had a small positive correlation with CCA scores. This finding aligns with one of the most common criticisms about self-report scales: their validity and accuracy is often threatened by social desirability, especially when true responses could reveal socially unacceptable or embarrassing data (Polit & Beck, 2008). Interestingly enough, in this study no significant correlation was found between post-test CCA scores and MCSDS scores or between MCSDS scores and change in CCA scores from pre- to post-test.

Attrition Findings

This study had substantial attrition rates of 44% and 42% after having begun the course from the socially isolated and socially interactive online CE courses respectively. Only one previous study was identified which explored the relationship between demographic variables and attrition from online nursing CE courses. Atack and Rankin (2002), who reported 16% attrition from an online course, found no significant difference between nurses who completed the course and nurses who withdrew in respect to: age, education, marital status, hours worked, shifts worked, or previous online education. In this study, only the region of US residence, ethnicity, and self-reported level of cultural competence had significant relationships with course/study completion. Unsolicited

explanations from about one-third of the participants who withdrew from this study revealed time and technology issues were common reasons for withdrawing. This is consistent with the study by Atack and Rankin, in which qualitative findings revealed technical problems, workload, and missing traditional face-to-face learning were common reasons for nurses to withdraw from an online CE course.

Course Evaluation Findings

Participants in group 1 (socially interactive course) and group 2 (socially isolated course) were asked to complete course evaluation questions designed by the researcher. See Appendix L for a full list of questions and Appendix M for responses. Questions for which there was a discrepancy of greater than 10% of participants for the two courses answering "agree" or "strongly agree" will be discussed here.

The first area of discrepancy between the responses of participants in the two courses was the amount of interaction with others enrolled in the course. While 81% of participants in the socially interactive course agreed or strongly agreed the amount of interaction with others in the course was appropriate, only 23% of participants in the socially isolated course felt the same. Social interaction within coursework is supported by the ISFOLE (Johnson & Aragon, 2003) and by social learning theory. Similarly, Rovai (2002) found students who feel less socially isolated in an online course have greater satisfaction. Billings et al. (2001) found social isolation from faculty and students is negatively correlated with student satisfaction, indicating online learners prefer to interact with other online learners during coursework.

The second area of discrepancy in the responses of the two groups was the amount of instructor involvement. While 90% of the socially interactive group agreed or

strongly agreed the amount of instructor involvement was appropriate, only 59% of those in the socially isolated course felt the same. This is consistent with the findings of several previous studies which have suggested instructor interaction is important to online learners. Billings et al. (2001) found social isolation from faculty and peers is negatively correlated with satisfaction. Bollinger and Martindale (2004) purport the instructor is the largest predictor for student satisfaction in online courses and that satisfaction is positively correlated with instructor performance, availability, and response times. Finally, Gallien and Oomen-Early (2008) found students who receive personalized feedback (such as the participants in this study's socially interactive course) are more satisfied than students who are given only collective feedback (such as the socially isolated group participants in this study).

The final area of discrepancy in the responses of the two groups was with regard to the length of each module. While 100% of the socially isolated group agreed or strongly agreed the length of each module was appropriate, only 81% of the socially interactive group felt the same. While the WebCampus system tracking function shows participants in the socially isolated course spent on average 41 more minutes in the course than those in the socially interactive course, this data does not account for idle time or time participants were logged into the course but not actively working on course materials. Unfortunately, there are no previous studies against which to judge this finding, as no previous work has compared socially isolated to socially interactive online CE or coursework.

Recommendations for Nursing and Nursing Education

The findings of this study reveal several concepts important to nursing and nursing education. These findings have led to several recommendations which will be described in the following paragraphs.

The major finding of this study is that socially isolated online cultural competence CE for women's health nurses effectively increases cultural competence level. As socially interactive online cultural competence CE is not more effective than socially isolated, but likely more expensive due to instructor time requirements, CE providers should consider further utilizing socially isolated asynchronous delivery methods. However, many nurses do not have the technology skills needed to be successful in online education. Therefore, nurses often need to be taught computer skills, Internet skills, and how to use asynchronous online course (Sweeney, Saarmann, Flagg, & Seidman, 2008). In fact, marketing should include information about how to enroll in and use the course (Sweeney et al.).

This study also reveals the number of types of previous cultural diversity training to be positively correlated with cultural competence scores. Therefore, exposing nurses to more types of cultural diversity training is of utmost importance. Continuing to include cultural competence in nursing education as a separate course or as a thread through the curriculum is only a first step. Cultural competence education should be incorporated into face-to-face seminars, workshops, and in other formats of CE offerings such as journal articles and socially isolated on-line modules. Employers must also support cultural competence CE by sponsoring programs for employees and making cultural competence CE a priority, if not mandatory. Employers should offer work or

educational time to complete cultural competence CE. If online CE is available, employers have a duty to provide computers with the needed technology in a quiet location for employee use.

CCA findings and communication via the online CE courses also provide insight as to what employers and nursing areas can do to enhance cultural competence. The CCA item which addresses having resource books and other reference materials available to help learn about people from other cultures had a mean score of "sometimes" across all three groups. Therefore, employers should ensure all nursing areas have reference materials and resources available with content specific to culture and that particular area of nursing. For example, books such as *Caring for Patients from Different Cultures*, by Geri-Ann Galanti (2008), and nursing journals such as *The American Journal of Maternal/Child Nursing* or *Nursing for Women's Health* provide information about culture and obstetrical nursing. Many course participants also stated they had never completed a comprehensive cultural assessment or their workplace did not have or use a cultural assessment tool. Therefore, nursing units and facilities should adopt client cultural assessment instruments which are relevant to their client population.

The final significant finding is that MCSDS scores had a small positive correlation with the CCA scores of the participants in this study. While it is said this has not been true in previous studies (S. Schim, personal communication, July 10, 2010), it is suggested the short form of the MCSDS continue to be used when assessing cultural competence level with the CCA. If MCSDS scores are found to be correlated with CCA score within other populations, MCSDS scores could serve as a covariate for CCA score analysis.

Study Limitations

As with any research study, this study was not without limitations. Several limitations related to the sampling method were noted. Other limitations related to inclusion criteria, timing, and attrition were also noted. Each of these will be discussed.

One of the major limitations which impacts ability to generalize study findings stems from the use of a non-probability sample limited to AWHONN members. While no other accessible population of women's health nurses was identifiable, studying only AWHONN members inhibits generalizability to all women's health nurses. It is possible that women's health nurses in the US who belong to the national nursing specialty organization differ from women's health nurses who do not. As membership in professional organizations impact the amount of CE nurses complete (Smith, 2004), and part of AWHONN's mission includes research and education (AWHONN, 2011), it is conceivable AWHONN members differ from other women's health nurses in relation to learning, changing, and growing as professionals.

Another limitation was that the entire desired sample was not accessible to the researcher. While it is not known exactly how many state's AWHONN leaders distributed information about the study to the members in their state, it is known the leaders from seven states did not disseminate the information. This likely led to the discrepancy in number of participants from each of the five regions in the US.

The technology inclusion criteria are another limitation which inhibits generalizability. As the study was conducted solely in the on-line environment, participation was limited to those with existing computer skills and access to specific equipment. It is improbable that all women's health nurses have an e-mail address,

access to a computer with speakers and high-speed Internet, and basic computer skills. Therefore, these requirements for participation limit ability to generalize findings to all women's health nurses.

Yet another limitation is that the study was conducted during a set time period. The online CE courses ran from mid-April to mid-May, encompassing holidays such as Easter, Passover, and Mother's Day which many in the US observe. This time of year is also inclusive of spring break for many students and the end of the semester for many universities. While all causes of attrition are not known, time requirements and vacations were reported by some participants who chose to withdraw from the study. These are also reflected in the responses to the course evaluation question which addressed things about the course which participants would like to see changed.

Finally, the high attrition rate is of concern. While 190 individuals submitted informed consent to participate, only 132 completed the Week 1 Survey and 93 completed the entire study. While attrition was explored between those who completed the Week 1 Survey and completed the entire study, no information is available for the attrition between informed consent and completing the Week 1 Survey to begin the study. Furthermore, due to unequal group size following attrition, a random sample of participants from the control group was used in the analysis of hypothesis one.

Each of the above limitations could potentially decrease the representativeness of the sample, therefore limiting the ability to generalize study findings to all women's health nurses. These limitations have led to the following recommendations for future research.

Recommendations for Future Research

Based on the findings of this study, the following are suggestions for further research:

- Replicate the study using a larger probability sample size of women's health nurses in every state to enhance ability to generalize findings to all women's health nurses in the US.
- 2. Duplicate the study with women's health nurses who are not AWHONN members, thus enhancing generalizability to all women's health nurses.
- 3. Execute the study at a different time of year to explore the effect time of year has on attrition.
- 4. Alter the study to explore the effect of online cultural competence CE on the cultural competence level of nurses in other specialty areas.
- 5. Investigate the effect longer duration socially interactive online CE courses have on cultural competence or other subjects/topics.
- Compare socially isolated online CE to socially interactive online CE in subject areas other than cultural competence.
- 7. Modify the demographic question regarding highest level of education completed to include doctorate level education, thus better aligning with nursing degrees. This could change the findings about the relationship between highest level of education completed and cultural competence.

Summary

This final chapter presented a summary of the study and its findings. Major findings demonstrated socially isolated online cultural competence CE for women's

health nurses is significantly more effective than not having online cultural competence CE. However, socially interactive online cultural competence CE is not significantly more effective than not having any cultural competence online CE; and socially interactive online cultural competence CE is not more effective than having socially isolated online cultural competence CE. Other major findings indicate both MCSDS scores and number of previous types of cultural diversity training have small positive correlations with pre-test CCA scores. Based on these findings suggestions were given for nursing and nursing education, study limitations were described, and recommendations for future research provided.

In light of the significant health disparities minority women experience, enhancing the cultural competence of nurses who care for childbearing women and/or their newborns is of great importance. Enhancing the cultural competence of women's health nurses may allow diverse women and newborns to receive improved care, thus lessening health disparities. Therefore, cultural competence CE should be completed by all women's health nurses. In light of its advantages, socially isolated online cultural competence CE should be utilized.

APPENDIX A

E-MAIL TO AWHONN STATE LEADERS

Dear State AWHONN Leader,

I am an AWHONN member currently pursuing my PhD in Nursing. I am writing to request that you send the following information to the AWHONN members in your state on my behalf.

I am conducting a study of on-line cultural competence continuing education courses for nurses who work with childbearing women and their families. Nurses who complete the research study will be granted ten free continuing education units (CEUs). For more information, please visit the following Website: http://complabs.nevada.edu/~heitzler . To register, submit the form on the Website by March 28, 2011. Use the following password to enter the Website: 10FreeCEUs

As you are aware, the importance of cultural competence continues to grow. In fact, AWHONN's Mission, Vision, & Values include promoting the cultural competence of nurses who provide care to women and newborns. I would greatly appreciate your assistance distributing information about my study to AWHONN members in your state. While passing along this e-mail is perhaps the easiest way to distribute the information, I am attaching both a PDF file and a JPEG file for your convenience. Nurses who would like to participate must visit the above Website and register by March 28, 2011.

Thank you,

Ella Heitzler, PhD(c), WHNP, RNC-OB, Student Investigator Doctoral Student, University of Nevada, Las Vegas AWHONN Member

Lori Candela, EdD, RN, Principal Investigator University of Nevada, Las Vegas

APPENDIX B

FOLLOW-UP PHONE CALL TO AWHONN STATE LEADERS

Hello [insert name],

My name is Ella Heitzler. I am a student investigator calling on behalf of myself and Dr. Lori Candela, principal investigator, from the University of Nevada, Las Vegas. I am calling to follow-up on the e-mail sent to you last week. The e-mail asked you to send information about a cultural competence online continuing education study to the AWHONN members in your state. I wanted to be sure you received the e-mail and give you an opportunity to ask any questions you may have.

APPENDIX C

PDF AND JPEG INVITATION

Participate in a Research Study and Earn Free Continuing Education Units!

You are invited to participate in a free online cultural competence course. The course is part of a research study about online cultural competence continuing education for nurses who work with childbearing women and their families. Nurses completing the study will receive 10 free continuing education units (CEUs).

For more information, or to register, visit the following Website by March 28, 2011: http://complabs.nevada.edu/~heitzler

Use the following password to enter the Website: 10FREECEUS

Lori Candela, EdD, RN, Principal Investigator & Ella Heitzler, PhD(c), WHNP-BC, RNC-OB, Student Investigator University of Nevada, Las Vegas

APPENDIX D

INFORMED CONSENT FORM FOR GROUP 1



INFORMED CONSENT

Department of Nursing

TITLE OF STUDY: Enhancing the Cultural Competence of Women's Health

Nurses via Online Continuing Education

INVESTIGATOR(S): Lori Candela, Ed.D., R.N. (Principal Investigator) and Ella

Heitzler, Ph.D.(c), R.N.C.-O.B. (Doctoral Student Investigator)

CONTACT PHONE NUMBER: Lori Candela (702-895-2443); Ella Heitzler (804-627-5383)

Purpose of the Study

You are invited to participate in a research study. The purpose of the study is to evaluate the effect of two different online continuing education courses on the cultural competence level of nurses who care for childbearing women and/or their newborns in the United States. The relationship between social desirability and self-reported level of cultural competence will also be explored, as will the relationship between selected demographic variables and level of cultural competence.

Participants

You are being asked to participate in the study because you are a registered nurse who provides care to childbearing women and/or their newborns in the United States.

Procedures

You have been randomly assigned to group one. Your online cultural competence course will run from April 11, 2011 to May 8, 2011. You will be contacted via e-mail and given information about how to access the course. You will be asked to complete a survey which includes demographic information, questions related to cultural competence, and questions regarding how you think, feel, and act in general. You will begin the cultural competence online course during the same week in which you complete this survey. The course will consist of four modules and will be offered over four consecutive weeks. Each module will only be available during its assigned week and will take about 2-3

hours to complete; you may complete each module anytime during the assigned week at your convenience. During the fourth week, after finishing the fourth module, you will be asked to complete a survey similar to the one you completed the first week. You will also be asked to answer questions about your experience in the course for the purpose of guiding course revisions. You will not be required to respond to any item on the survey(s) to which you do not wish to respond. Upon completing the course and accessing the surveys you will receive a certificate via e-mail attachment for 10 continuing education units. You must complete all modules/activities and access the two surveys in order to receive the continuing education units; no partial credit or hours will be awarded.

Benefits of Participation

There may be direct benefits to you as a participant in this study. As a nurse, you may become more knowledgeable about culturally competent care of childbearing women. You may notice a change in your attitude toward culturally diverse clients and your nursing assessment and intervention skills may increase. You will gain exposure to learning in an online environment which could enhance your computer skills.

Overall, your participation might lead to the provision of more culturally competent adequate care, thus decreasing health disparities and improving outcomes for diverse childbearing women and their newborns. We hope to learn more about online continuing education methodologies and whether or not they can increase the cultural competence level of nurses who care for diverse childbearing women.

Risks of Participation

There are risks involved in all research studies. This study is thought to include only minimal risks. The study includes exploration of diverse cultures, personal attitudes/beliefs about diverse cultures, cultural competence behaviors, and activities such as cultural assessment. Therefore, you may be uncomfortable when answering some questions. However, you may choose to not answer any survey item(s). You may also be uncomfortable during course activities or be concerned your answers will be released to the public or your peers. Therefore, you may use any active e-mail account and may opt to use a pseudonym within the online course so only the investigator will have knowledge of your true identity. Please e-mail the investigator (Ella Heitzler) at <u>heitzler@unlv.nevada.edu</u> by April 3, 2011 if you wish to use a different e-mail account or a pseudonym for the online course.

Cost /Compensation

There is no financial cost to you to participate in this study. The study will take approximately 11 hours of your time. You will be compensated for your time by receiving 10 free continuing education units sponsored by the University of Nevada, Las Vegas School of Nursing upon your completion of one of the online cultural competence courses and accessing all surveys. The continuing education units certificate will be sent to you electronically via the e-mail address you provided when you registered for the study. You are not required to submit your nursing license number/state if you do not wish to receive continuing education units.

Contact Information

If you have any questions or concerns about the study, you may contact Lori Candela at **702-895-2443** or Ella Heitzler at **804-627-5383**. For questions regarding the rights of research subjects, any complaints or comments regarding the manner in which the study is being conducted you may contact **the UNLV Office of Research Integrity – Human Subjects at 702-895-2794 or toll free at 877-895-2794 or via email at IRB@unlv.edu.**

Voluntary Participation

Your participation in this study is voluntary. You may refuse to participate in this study or in any part of it. You may withdraw at any time without negative consequences. You are encouraged to ask any questions you may have before the study begins.

Confidentiality

The information gathered in this study will be kept completely confidential. Your name will not be included in any publications related to the study. Study results will only be reported for the group as a whole; no single individual will be referred to or identified in any presentation or written work. Should you choose to accept the continuing education units, the University of Nevada, Las Vegas continuing education coordinator and the Nevada State Board of Nursing will have access to your legal name and nursing license number/state. Electronic data will be maintained on an encrypted external computer drive to which only the investigator will have access. After three years, all data will be destroyed.

Participant Consent:

By entering your legal name and clicking "I consent" you acknowledge you have read the above information, consent to participate in the study, are at least 18 years of age, and are aware you can withdraw at any time without penalty. You may print a copy of this document for your records if you so choose.

APPENDIX E

INFORMED CONSENT FORM FOR GROUP 2



INFORMED CONSENT

Department of Nursing

TITLE OF STUDY: Enhancing the Cultural Competence of Women's Health

Nurses via Online Continuing Education

INVESTIGATOR(S): Lori Candela, Ed.D., R.N. (Principal Investigator) and Ella

Heitzler, Pd.D.(c), R.N.C.-O.B. (Doctoral Student Investigator)

CONTACT PHONE NUMBER: Lori Candela (702-895-2443); Ella Heitzler (804-627-5383)

Purpose of the Study

You are invited to participate in a research study. The purpose of the study is to evaluate the effect of two different online continuing education courses on the cultural competence level of nurses who care for childbearing women and/or their newborns in the United States. The relationship between social desirability and self-reported level of cultural competence will also be explored, as will the relationship between selected demographic variables and level of cultural competence.

Participants

You are being asked to participate in the study because you are a registered nurse who provides care to childbearing women and/or their newborns in the United States.

Procedures

You have been randomly assigned to group two. Your online cultural competence course will run from April 11, 2011 to May 8, 2011. You will be contacted via e-mail and given information about how to access the course. You will be asked to complete a survey which includes demographic information, questions related to cultural competence, and questions regarding how you think, feel, and act in general. You will begin the cultural competence online course during the same week in which you complete this survey. The course will consist of four modules and will be offered over four consecutive weeks. Each module will only be available during its assigned week and will take about 2-3

hours to complete; you may complete each module anytime during the assigned week at your convenience. During the fourth week, after finishing the fourth module, you will be asked to complete a survey similar to the one you completed the first week. You will also be asked to answer questions about your experience in the course for the purpose of guiding course revisions. You will not be required to respond to any item on the survey(s) to which you do not wish to respond. Upon completing the course and accessing the surveys you will receive a certificate via e-mail attachment for 10 continuing education units. You must complete all modules/activities and access the two surveys in order to receive the continuing education units; no partial credit or hours will be awarded.

Benefits of Participation

There may be direct benefits to you as a participant in this study. As a nurse, you may become more knowledgeable about culturally competent care of childbearing women. You may notice a change in your attitude toward culturally diverse clients and your nursing assessment and intervention skills may increase. You will gain exposure to learning in an online environment which could enhance your computer skills.

Overall, your participation might lead to the provision of more culturally competent adequate care, thus decreasing health disparities and improving outcomes for diverse childbearing women and their newborns. We hope to learn more about online continuing education methodologies and whether or not they can increase the cultural competence level of nurses who care for diverse childbearing women.

Risks of Participation

There are risks involved in all research studies. This study is thought to include only minimal risks. The study includes exploration of diverse cultures, personal attitudes/beliefs about diverse cultures, cultural competence behaviors, and activities such as cultural assessment. Therefore, you may be uncomfortable when answering some questions. However, you may choose to not answer any survey item(s). You may also be uncomfortable during course activities or be concerned your answers will be released to the public or your peers. Therefore, you may use any active e-mail account and may opt to use a pseudonym within the online course so only the investigator will have knowledge of your true identity. Please e-mail the investigator (Ella Heitzler) at <u>heitzler@unlv.nevada.edu</u> by April 3, 2011 if you wish to use a different e-mail account or a pseudonym for the online course.

Cost /Compensation

There is no financial cost to you to participate in this study. The study will take approximately 11 hours of your time. You will be compensated for your time by receiving 10 free continuing education units sponsored by the University of Nevada, Las Vegas School of Nursing upon your completion of one of the online cultural competence courses and accessing all surveys. The continuing education units certificate will be sent to you electronically via the e-mail address you provided when you registered for the study. You are not required to submit your nursing license number/state if you do not wish to receive continuing education units.

Contact Information

If you have any questions or concerns about the study, you may contact Lori Candela at **702-895-2443** or Ella Heitzler at **804-627-5383.** For questions regarding the rights of research subjects, any complaints or comments regarding the manner in which the study is being conducted you may contact **the UNLV Office of Research Integrity – Human Subjects at 702-895-2794 or toll free at 877-895-2794 or via email at IRB@unlv.edu.**

Voluntary Participation

Your participation in this study is voluntary. You may refuse to participate in this study or in any part of it. You may withdraw at any time without negative consequences. You are encouraged to ask any questions you may have before the study begins.

Confidentiality

The information gathered in this study will be kept completely confidential. Your name will not be included in any publications related to the study. Study results will only be reported for the group as a whole; no single individual will be referred to or identified in any presentation or written work. Should you choose to accept the continuing education units, the University of Nevada, Las Vegas continuing education coordinator and the Nevada State Board of Nursing will have access to your legal name and nursing license number/state. Electronic data will be maintained on an encrypted external computer drive to which only the investigator will have access. After three years, all data will be destroyed.

Participant Consent:

By entering your legal name and clicking "I consent" you acknowledge you have read the above information, consent to participate in the study, are at least 18 years of age, and are aware you can withdraw at any time without penalty. You may print a copy of this document for your records if you so choose.

APPENDIX F

INFORMED CONSENT FORM FOR GROUP 3



INFORMED CONSENT

Department of Nursing

TITLE OF STUDY: Enhancing the Cultural Competence of Women's Health

Nurses via Online Continuing Education

INVESTIGATOR(S): Lori Candela, Ed.D., R.N. (Principal Investigator) and Ella

Heitzler, Ph.D.(c), R.N.C.-O.B. (Doctoral Student Investigator)

CONTACT PHONE NUMBER: Lori Candela (702-895-2443); Ella Heitzler (804-627-5383)

Purpose of the Study

You are invited to participate in a research study. The purpose of the study is to evaluate the effect of two different online continuing education courses on the cultural competence level of nurses who care for childbearing women and/or their newborns in the United States. The relationship between social desirability and self-reported level of cultural competence will also be explored, as will the relationship between selected demographic variables and level of cultural competence.

Participants

You are being asked to participate in the study because you are a registered nurse who provides care to childbearing women and/or their newborns in the United States.

Procedures

You have been randomly assigned to group three, the control group. You will be e-mailed an online survey Web link and asked to complete two surveys including demographic information and questions related to cultural competence and how you think, feel, and act in general. One survey will be accessible during the week of April 11, 2011 and the other four weeks later during the week of May 2, 2011. You will not be required to respond to any item(s) on the surveys to which you do not wish to respond. You will not complete an online course immediately; rather, you will be contacted during the week of May 16, 2011 via the e-mail address you supplied and given the opportunity to begin an online course approximately 2 weeks after you complete the second survey. Upon completing the course you will be asked to complete a course evaluation survey. After accessing the course evaluation you will receive a certificate via e-mail attachment for 10 continuing education units. You must complete all modules/activities and access the surveys in order to receive the continuing education units; no partial credit or hours will be awarded.

Benefits of Participation

There may be direct benefits to you as a participant in this study. As a nurse, you may become more knowledgeable about culturally competent care of childbearing women. You may notice a change in your attitude toward culturally diverse clients and your nursing assessment and intervention skills may increase. You will also have the opportunity to gain exposure to learning in an online environment which could enhance your computer skills.

Overall, your participation might lead to the provision of more culturally competent adequate care, thus decreasing health disparities and improving outcomes for diverse childbearing women and their newborns. We hope to learn more about online continuing education methodologies and whether or not they can increase the cultural competence level of nurses who care for diverse childbearing women.

Risks of Participation

There are risks involved in all research studies. This study is thought to include only minimal risks. The study includes exploration of diverse cultures, personal attitudes/beliefs about diverse cultures, cultural competence behaviors, and activities such as cultural assessment. Therefore, you may be uncomfortable when answering some questions. However, you may choose to not answer any survey item(s). You may also be uncomfortable during course activities or be concerned your answers will be released to the public or your peers. Therefore, you may use any active e-mail account and may opt to use a pseudonym within the online course so only the investigator will have knowledge of your true identity. Please e-mail the investigator (Ella Heitzler) at <u>heitzler@unlv.nevada.edu</u> by May 8, 2011 if you wish to use a different e-mail account or a pseudonym for the online course.

Cost /Compensation

There is no financial cost to you to participate in this study. The study will take approximately 11 hours of your time. You will be compensated for your time by receiving 10 free continuing education units sponsored by the University of Nevada, Las Vegas School of Nursing upon your completion of one of the online cultural competence courses and accessing all surveys. The continuing education units certificate will be sent to you electronically via the e-mail address you provided when you registered for the study. You are not required to submit your nursing license number/state if you do not wish to receive continuing education units.

Contact Information
If you have any questions or concerns about the study, you may contact Lori Candela at **702-895-2443** or Ella Heitzler at **804-627-5383.** For questions regarding the rights of research subjects, any complaints or comments regarding the manner in which the study is being conducted you may contact the UNLV Office of Research Integrity – Human Subjects at 702-895-2794 or toll free at 877-895-2794 or via email at IRB@unlv.edu.

Voluntary Participation

Your participation in this study is voluntary. You may refuse to participate in this study or in any part of it. You may withdraw at any time without negative consequences. You are encouraged to ask any questions you may have before the study begins.

Confidentiality

The information gathered in this study will be kept completely confidential. Your name will not be included in any publications related to the study. Study results will only be reported for the group as a whole; no single individual will be referred to or identified in any presentation or written work. Should you choose to accept the continuing education units, the University of Nevada, Las Vegas continuing education coordinator and the Nevada State Board of Nursing will have access to your legal name and nursing license number/state. Electronic data will be maintained on an encrypted external computer drive to which only the investigator will have access. After three years, all data will be destroyed.

Participant Consent:

By entering your legal name and clicking "I consent" you acknowledge you have read the above information, consent to participate in the study, are at least 18 years of age, and are aware you can withdraw at any time without penalty. You may print a copy of this document for your records if you so choose.

APPENDIX G

ORIGINAL IRB APPROVAL



Biomedical IRB – Expedited Review Approval Notice

NOTICE TO ALL RESEARCHERS:

Please be aware that a protocol violation (e.g., failure to submit a modification for <u>any</u> change) of an IRB approved protocol may result in mandatory remedial education, additional audits, re-consenting subjects, researcher probation, suspension of any research protocol at issue, suspension of additional existing research protocols, invalidation of all research conducted under the research protocol at issue, and further appropriate consequences as determined by the IRB and the Institutional Officer.

| DATE: | December 8, 2010 |
|-------|--|
| TO: | Dr. Lori Candela, Physiological Nursing |
| FROM: | Office of Research Integrity - Human Subjects |
| RE: | Notification of IRB Action by /Charles Rasmussen/ Dr. Charles Rasmussen, Co-Chair Protocol Title: Enhancing the Cultural Competence of Wonmen's Health Nurses via Online Continuing Education Protocol #: 1011-3651 Expiration Date:December 7, 2011 |

This memorandum is notification that the project referenced above has been reviewed and approved by the UNLV Social/Behavioral Institutional Review Board (IRB) as indicated in Federal regulatory statutes 45 CFR 46 and UNLV Human Research Policies and Procedures.

The protocol is approved for a period of one year and expires December 7, 2011. If the above-referenced project has not been completed by this date you must request renewal by submitting a Continuing Review Request form 30 days before the expiration date.

PLEASE NOTE:

Upon approval, the research team is responsible for conducting the research as stated in the protocol most recently reviewed and approved by the IRB, which shall include using the most recently submitted Informed Consent/Assent forms and recruitment materials. The official versions of these forms are indicated by footer which contains approval and expiration dates.

Should there be *any* change to the protocol, it will be necessary to submit a **Modification Form** through ORI -Human Subjects. No changes may be made to the existing protocol until modifications have been approved by the IRB. Modified versions of protocol materials must be used upon review and approval. Unanticipated problems, deviations to protocols, and adverse events must be reported to the ORI – HS within 10 days of occurrence.

If you have questions or require any assistance, please contact the Office of Research Integrity - Human Subjects at IRB@unlv.edu or call 895-2794.

Office of Research Integrity - Human Subjects 4505 Maryland Parkway * Box 451047 * Las Vegas, Nevada 89154-1047 (702) 895-2794 * FAX: (702) 895-0805

APPENDIX H

IRB MODIFICATION APPROVAL



Biomedical IRB – Expedited Review Modification Approved

NOTICE TO ALL RESEARCHERS:

Please be aware that a protocol violation (e.g., failure to submit a modification for any change) of an IRB approved protocol may result in mandatory remedial education, additional audits, re-consenting subjects, researcher probation, suspension of any research protocol at issue, suspension of additional existing research protocols, invalidation of all research conducted under the research protocol at issue, and further appropriate consequences as determined by the IRB and the Institutional Officer.

DATE: February 14, 2011

TO: Dr. Lori Candela, Nursing

FROM:

Office of Research Integrity - Human Subjects Notification of IRB Action by /Charles Rasmussen/ Dr. Charles Rasmussen, Co-Chair RE: Protocol Title: Enhancing the Cultural Competence of Women's Health Nurses via Online **Continuing Education** Protocol #: 1011-3651 Expiration Date:December 7, 2011

The modification of the protocol named above has been reviewed and approved.

Modifications reviewed for this action include:

- > Change of recruitment method to sending an email to each state's AWHONN leader to ask them to send/distribute the study invitation to members in their state.
- Informed Consent revised to allow for the date change and participant changes.
- Population changed to any nurses who care for childbearing women and their newborns will be \geq included.
- > Pilot study was removed due to recruitment change.
- Number of subjects decreased.
- > Additional demographic questions added to the survey.

This IRB action will not reset your expiration date for this protocol. The current expiration date for this protocol is December 7, 2011.

PLEASE NOTE:

Upon approval, the research team is responsible for conducting the research as stated in the protocol most recently reviewed and approved by the IRB, which shall include using the most recently submitted Informed Consent/Assent forms and recruitment materials. The official versions of these forms are indicated by footer which contains approval and expiration dates.

Should there be any change to the protocol, it will be necessary to submit a Modification Form through ORI -Human Subjects. No changes may be made to the existing protocol until modifications have been approved by the IRB. Modified versions of protocol materials must be used upon review and approval. Unanticipated problems, deviations to protocols, and adverse events must be reported to the ORI - HS within 10 days of occurrence. Should the use of human subjects described in this protocol continue beyond December 7, 2011, it would be necessary to submit a **Continuing Review Request Form** 30 days before the expiration date. If you have questions or require any assistance, please contact the Office of Research Integrity - Human Subjects at IRB@unlv.edu or call 895-2794.

> Office of Research Integrity - Human Subjects 4505 Maryland Parkway · Box 451047 · Las Vegas, Nevada 89154-1047 (702) 895-2794 · FAX: (702) 895-0805

APPENDIX I

RECOMMENDED ONLINE COURSE DESIGN PRINCIPLES

| Design/Layout | Description of Recommendations/Specific Suggestions from the |
|---------------|---|
| Element | Literature |
| Page Layout | 1. Material linked from the home page (Dunet, Reyes, Grossniklaus, Volansky & Blanck 2008: O'Neil 2009) |
| | 2 Information "chunked" together for ease of navigation (O'Neil 2009) |
| | 3 Three click rule: able to access course content using no more than three |
| | clicks of the mouse (O'Neil, 2009). |
| | 4. The same things are in the same place throughout the course site |
| | including: colors, fonts, headings, text, navigation, and etc. Consistency is key (Gillani, 2003; O'Neil, 2009). |
| | 5. Each page has a specific title (O'Neil, 2009). |
| | 6. Scale, contrast, and hierarchy used to make pages attractive and easy to follow (Gillani, 2003). |
| | 7. Attention directed to information and concepts which are critical or may be confusing (Dunet et al., 2008). |
| | 8. Ample white space (empty space) (Dunet et al., 2008; Gillani, 2003). |
| | 9. Bulleted format used when possible (Dunet et al., 2008). |
| | 10. Page length regulated (Buhmann & Johnson, 2009). |
| | 11. Most important information is presented first (Buhmann & Johnson, |
| | 2009). |
| Content | 1. Course has introduction (Blood-Siegfried et al., 2008). |
| | 2. Material separated into small sections (Dunet et al., 2008). |
| | 3. Modular format used (Dunet et al., 2008; Gillani, 2003). |
| | 4. Each module has a separate introduction (Wilkinson, Forbes, |
| | Bloomfield, & Gee., 2004). |
| | 5. Each module has clearly defined objectives (Blood-Siegfried et al., 2008; Dykman & Davis, 2008). |
| | 6. Each module begins with a written or recorded "lecture" which |
| | summarizes module contents (Dykman & Davis, 2008). |
| | 7. Audiovisual materials used to present information (Dykman & Davis, 2008). |
| | 8. Key points and ideas reviewed in a format which can be downloaded or easily printed off (Dunet et al., 2008). |
| | 9. Main text limited to key points (Dunet et al., 2008). |
| | 10. Specific timeframes/due dates are given (Dykman & Davis, 2008). |
| | 11. Course starts slow with a lighter workload (Dykman & Davis, 2008). |
| Text/Fonts | 1. Consistent texts and fonts (Gillani, 2003; O'Neil, 2009). |
| | 2. Bold used sparingly (O'Neil, 2009). |
| | 3. Blue text not used; suggests a hyperlink (O'Neil, 2009). |
| | 4. Text not underlined; underlining looks like a hyperlink (Buhmann & Johnson, 2009). |
| | 5. Text aligns to the left while headings/titles may be centered (Buhmann & Johnson, 2009). |
| | 6. No more than three fonts (O'Neil, 2009). |

| | 7. | Serif fonts may be used for page titles. Sans serif fonts used for |
|-----------------|----|---|
| | | information/main text (Gillani, 2003). Veranda is a sans serif font |
| | | which was designed specifically for online viewing (Buhmann & |
| | | Johnson, 2009). Two types of fonts: serif and sans serif. Serif fonts |
| | | have flags or decorations at the end of each stroke (such as Times or |
| | | New Century) while sans serif do not (Helvetica, New York, Chicago, |
| | | Arial). |
| | 8. | Both upper and lowercase letters used (Gillani, 2003). |
| | 9. | 10 or 12 point font for the body text, 8 for footer, and 12 or 14 for |
| | | headings (Buhmann & Johnson, 2009). |
| Use of Color | 1. | No more than five plus or minus two colors (three to seven colors |
| | | maximum) used within the course (Gillani, 2003). |
| | 2. | Both dark and light colors used (Gillani, 2003). |
| | 3. | Black text preferred (Gillani, 2003). |
| | 4. | Background color is light blue or purple (Gillani, 2003). |
| | 5. | Color theory: blue inspires calmness and suggests confidence and |
| | | comfort. Red suggests danger or alertness while orange calls attention. |
| | | Green suggests actions or cleanliness, yellow suggests fun, white |
| | | suggests purity, and black suggests mystery (Gillani, 2003). |
| Graphics | 1. | Images compressed or reduced in size so easy to download (Gillani, |
| | | 2003). |
| | 2. | GIF formatted images are the most ideal (Gillani, 2003). |
| | 3. | All images/graphics should convey information instead of being art. |
| | | Avoid cluttering with extraneous graphics (Dunet et al., 2008, Gillani, |
| | | 2003). |
| | 4. | Simple, clear, consistent icons are used for navigation use (Gillani, |
| | | 2003). |
| | 5. | Be conscious of copyright law (Buhmann & Johnson, 2009). |
| Additional | 1. | Class size limited to 15 to 20 students for new educators and 25 to 30 |
| Recommendations | | for experienced (Dykman & Davis, 2008). |

APPENDIX J

PERMISSION TO USE CCA

Re: Request

Saturday, October 9, 2010 7:57 AM

From: "Stephanie Myers Schim" <s.schim@wayne.edu> To: "Ella Heitzler"

I responded to your request on October 7, but it appears you may have not gotten my message. Let me try again...

Dear Ella -

You formally have my permission to use the CCA instrument in you dissertation research. You may modify the demographic questions as needed to work with your population as needed.

I am attaching the most recent MS Word version of the instrument for your use. If you have additional questions as you procede with your work, please do not hesitate to contact me.

Stephanie Myers Schim, PhD, RN, PHCNS-BC Associate Professor Family, Community, and Mental Health Nursing Wayne State University 240 Cohn Building (313) 577-4034 <u>s.schim@wayne.edu</u>

From: "Ella Heitzler" To: "s schim" <s.schim@wayne.edu> Sent: Friday, October 8, 2010 3:51:53 PM Subject: Request

Hello Dr. Schim,

As you are aware, I am preparing for my dissertation and would like to use the CCA instrument in my work as well as the 3-D Puzzle Model of Culturally Congruent Care. While we have previously discussed my use of the instrument and model, I would like to formally ask for permission to use them for my dissertation about online cultural competence continuing education for nurses.

Also, when we spoke last in July, you were having computer difficulties and had just ordered a copy of the most recent version of the CCA from your library. Would you please forward it to me so I use the most recent version?

Thank you, Ella Heitzler

APPENDIX K

CCA, MCSDS, AND DEMOGRAPHIC ITEMS

- 1. In the past 12 months, which of the following racial/ethnic groups have you encountered among your clients and their families or within the health care environment or workplace? *Mark 'X' for all that apply.*
 - Hispanic/Latino (including Mexican, Mexican American, Chicano, Puerto Rican, Cuban, other Spanish)
 - U White/Caucasian/European American
 - □ Black/African American/Negro
 - American Indian/Alaska Native
 - Asian (Asian Indian, Chinese, Filipino, Japanese, Korean, Vietnamese, or other Asian)
 - □ Native Hawaiian/Pacific Islander
 - □ Arab American/Middle eastern
 - Other (specify)
- 2. In your current environment what percentage of the total population is made up of people from these racial/ethnic groups? *Write in percents to add to 100%*
 - _____ Hispanic/Latino (including Mexican, Mexican American, Chicano, Puerto Rican, Cuban, other Spanish)
 - White/Caucasian/European American
 - _____ Black/African American/Negro
 - _____ American Indian/Alaska Native
 - _____ Asian (Asian Indian, Chinese, Filipino, Japanese, Korean, Vietnamese, or other Asian)
 - _____ Native Hawaiian/Pacific Islander
 - ____ Arab American/Middle Eastern
 - ____ All other groups combined
 - 100 % = TOTAL
- 3. In the past 12 months which of the following special population groups have you encountered among your clients and their families or within the health care environment or workplace? *Mark 'X' for all that apply.*
 - Mentally or emotionally III
 - Physically Challenged/Disabled
 - □ Homeless/Housing Insecure
 - □ Substance Abusers/Alcoholics
 - Gay, Lesbian, Bisexual, or Transgender
 - Different religious/spiritual backgrounds
 - Other (specify)

- 4. In your current environment what percentage of the total population is made up of people from these special population groups? *Write in percents; may not total 100%*
 - ____ Mentally or emotionally III
 - _____ Physically Challenged/Disabled
 - _____ Homeless/Housing Insecure
 - _____ Substance Abusers/Alcoholics
 - _____ Gay, Lesbian, Bisexual, or Transgender
 - _____ Different religious/spiritual backgrounds
- 5. Overall, how competent do you feel working with people who are from cultures different than your own?

| Very competent | Somewhat competent | Neither competent nor incompetent | Somewhat Incompetent | Very Incompetent |
|----------------|--------------------|--------------------------------------|-------------------------|------------------|
| | | | | |

[CCA Items- CAS Subscale]

6. Race is the most important factor in determining a person's culture.

| Strongly | | Somewhat | | Somewhat | | Strongly | No |
|----------|-------|----------|---------|----------|----------|----------|---------|
| Agree | Agree | Agree | Neutral | Disagree | Disagree | Disagree | Opinion |
| | | | | | | | |

7. People with a common cultural background think and act alike.

| Strongly | | Somewhat | | Somewhat | | Strongly | No |
|----------|-------|----------|---------|----------|----------|----------|---------|
| Agree | Agree | Agree | Neutral | Disagree | Disagree | Disagree | Opinion |
| | | | | | | | |

8. Many aspects of culture influence health and health care.

| Strongly | | Somewhat | | Somewhat | | Strongly | No |
|----------|-------|----------|---------|----------|----------|----------|---------|
| Agree | Agree | Agree | Neutral | Disagree | Disagree | Disagree | Opinion |
| | | | | | | | |

9. Aspects of cultural diversity need to be assessed for each individual, group, and organization.

| Strongly | | Somewhat | | Somewhat | | Strongly | No |
|----------|-------|----------|---------|----------|----------|----------|---------|
| Agree | Agree | Agree | Neutral | Disagree | Disagree | Disagree | Opinion |
| | | | | | | | |

10. If I know about a person's culture, I don't need to assess their personal preferences for health services.

| Strongly | | Somewhat | | Somewhat | | Strongly | No |
|----------|-------|----------|---------|----------|----------|----------|---------|
| Agree | Agree | Agree | Neutral | Disagree | Disagree | Disagree | Opinion |

| 11. Spiritually and religious beliefs are important aspects of many cultural groups. | | | | | | | | | |
|--|--------------------------|-----------------------------------|-----------------------------|----------------------|---------------|----------------------|---------------|--|--|
| Strongly Agree | Agree | Somewhat Agree | Neutral | Somewhat Disagree | Disagree | Strongly Disagree | No Opinion | | |
| 12. Individu | ial people | e may identify | with more | than one cul | tural group. | | | | |
| Strongly Agree | Agree | Somewhat Agree | Neutral | Somewhat Disagree | Disagree | Strongly Disagree | No Opinion | | |
| 13. Langua | ge barrie | rs are the only | y difficultie | s for recent ir | nmigrants to | the United | I Sates. | | |
| Strongly Agree | Agree | Somewhat Agree | Neutral | Somewhat Disagree | Disagree | Strongly Disagree | No Opinion | | |
| 14. I believ heritage | e that eve e. | eryone should | be treated | d with respect | t no matter v | vhat their c | ultural | | |
| Strongly Agree | Agree | Somewhat Agree | Neutral | Somewhat Disagree | Disagree | Strongly Disagree | No Opinion | | |
| 15. I unders care" in | stand that different | t people from ways. | different c | ultures may c | lefine the co | ncept of "h | ealth | | |
| Strongly Agree | Agree | Somewhat Agree | Neutral | Somewhat Disagree | Disagree | Strongly Disagree | No Opinion | | |
| 16. I think t individu | hat knowi Ials, famil | ing about diffe ies, groups, a | erent cultur Ind organiz | al groups he | lps direct my | work with | | | |
| Strongly Agree | Agree | Somewhat Agree | Neutral | Somewhat Disagree | Disagree | Strongly Disagree | No Opinion | | |
| | | [CCA | Items- | CCB Subso | cale] | | | | |
| 17. l include | e cultural | assessment | when I do i | individual or o | organization | al evaluatic | ons. | | |
| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure | | |

18. I seek information on cultural needs when I identify new people in my work or school.

| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
|--------|---------------|-------------------|-------|-----------|-----------|-------|----------|
| | | | | | | | |

19. I have resource books and other materials available to help me learn about people from different cultures.

| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
|--------|---------------|-------------------|-------|-----------|-----------|-------|----------|
| | | | | | | | |

20. I use a variety of sources to learn about the cultural heritage of other people.

| | Very | Somewhat | | - · | | | |
|--------|-------|----------|-------|-----------|-----------|-------|----------|
| Always | Often | Often | Often | Sometimes | Few Times | Never | Not sure |
| | | | | | | | |

21. I ask people to tell me about their own explanations of health and illness.

| Alwavs | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
|--------|---------------|-------------------|-------|-----------|-----------|-------|----------|
| | | | | | | | |

22. I ask people to tell me about their expectations for health services.

| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
|--------|---------------|-------------------|-------|-----------|-----------|-------|----------|
| | | | | | | | |

23. I avoid using generalizations to stereotype groups of people.

| | Very | Somewhat | | | | | |
|--------|-------|----------|-------|-----------|-----------|-------|----------|
| Always | Often | Often | Often | Sometimes | Few Times | Never | Not sure |
| | | | | | | | |

24. I recognize potential barriers to service that might be encountered by different people.

| | Very | Somewhat | | | | | |
|--------|-------|----------|-------|-----------|-----------|-------|----------|
| Always | Often | Often | Often | Sometimes | Few Times | Never | Not sure |
| | | | | | | | |

25. I remove obstacles for people of different cultures when I identify barriers to services.

| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
|--------------------|---------------|-------------------|---------------|------------------|-----------------|--------------|-----------|
| 26. I remov me. | e obstacl | es for people o | of differen | t cultures whe | en people ide | entify barri | iers to |
| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
| 27. I welcor | ne feedba | ack from client | ts about h | ow I relate to | people from | different | cultures. |
| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
| 28. I find wa | ays to ada | apt my service | s to indivi | dual and grou | up cultural pro | eferences | |
| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |
| 29. I docum | nent cultu | ral assessmer | nts if I prov | vide direct clie | ent services. | | |
| Always | Very Often | Somewhat Often | Often | Sometimes | Few Times | Never | Not sure |

30. I document the adaptations I make with clients if I provide direct client services.

| | Very | Somewhat | | | | | |
|--------|-------|----------|-------|-----------|-----------|-------|----------|
| Always | Often | Often | Often | Sometimes | Few Times | Never | Not sure |
| | | | | | | | |

[MCSDS Items]

31. It is sometimes hard for me to go on with my work if I am not encouraged.

True False

32. I sometimes feel resentful when I don't get my way.

| True | False |
|------|-------|
| | |

33. On a few occasions, I have given up doing something because I thought too little of my ability.

True False

34. There have been times when I felt like rebelling against people in authority even though I knew they were right.

True False

35. No matter who I'm talking to, I'm always a good listener.

| True | False |
|------|-------|
| | |

36. There have been occasions when I took advantage of someone.

True False

37. I'm always willing to admit it when I make a mistake.

True False

38. I sometimes try to get even rather than forgive and forget.

True False

39. I am always courteous, even to people who are disagreeable.

True False

40. I have never been irked when people expressed ideas very different from my own.

True False

41. There have been times when I was quite jealous of the good fortune others.

True False

42. I am sometimes irritated by people who ask favors of me.

True False

43. I have never deliberately said something to hurt someone's feelings.

| True | False |
|------|-------|
| | |

[Demographic Items]

- 44. In what year were you born?
- 45. In what part of the US do you live?
 - □ Northeast (CT, DE, ME, MD, MA, NH, NJ, NY, PA, RI, VT)
 - □ Midwest (IL, IN, IA, KS, MI, MN, MO, NE, ND, OH, SD, WI)
 - Southeast (AL, AR, FL, GA, KY, LA, MS, NC, SC, TN, VA, WV)
 - □ Southwest (AZ, NM, TX, OK)
 - U West (AK, CA, CO, HI, ID, MT, NV, OR, UT, WA, WY)

46. For how many years have you been a practicing nurse?

47. In what area do you currently work? (Choose one or more)

- □ Antepartal (In-patient)
- □ Intrapartal/L&D (In-patient)
- Postpartal (In-patient)
- □ Mother/Infant (In-patient)
- □ Nursery/NICU (In-patient)
- Out-patient/Ambulatory Care
- Community Health
- Other

48. For how many years have you worked in your specialty area?



49. How do you spend the majority of your time?

- Direct patient care
- Management
- Education/Academia
- Other
- 50. Using the categories below, what do you consider yourself? (Choose one or more)
 - Hispanic/Latino (including Mexican, Mexican American, Chicano, Puerto Rican, Cuban, other Spanish)
 - □ White/Caucasian/European American

- Black/African American/Negro
- American Indian/Alaska Native
- Asian (Asian Indian, Chinese, Filipino, Japanese, Korean, Vietnamese, or other Asian)
- □ Native Hawaiian/Pacific Islander
- Arab American/Middle eastern
- Other (specify) ______
- 51. If you chose more than one category when answering number 7 above, which do you consider to be your main category?
 - Hispanic/Latino (including Mexican, Mexican American, Chicano, Puerto Rican, Cuban, other Spanish)
 - □ White/Caucasian/European American
 - □ Black/African American/Negro
 - □ American Indian/Alaska Native
 - Asian (Asian Indian, Chinese, Filipino, Japanese, Korean, Vietnamese, or other Asian)
 - □ Native Hawaiian/Pacific Islander
 - □ Arab American/Middle eastern
 - Other (specify) _____
- 52. What is your highest level of education completed?
 - Less than high school
 - Diploma
 - High school or GED
 - □ Associate degree
 - Bachelors degree
 - Graduate or professional degree
- 53. Was your original nursing training completed in the United States?
 - □ Yes
 - 🛛 No
- 54. Have you ever participated in cultural diversity training?
 - □ Yes
 - 🛛 No
- 55. If you have had prior diversity training, which option below best describes it? *(Choose one or more)*
 - □ Separate college course for credit
 - Content covered in a college course
 - □ Professional Conference or Seminar
 - □ Employer Sponsored Program
 - □ On-line (computer assisted) Education
 - Continuing Education Offering
 - Other diversity training types (Specify) ______
- 56. Have you ever completed any education or training online? □ Yes

🛛 No

- 57. How comfortable are you when using a computer?
 - Very
 - □ Somewhat
 - Not at all
- 58. Where do you have access to a computer with high-speed Internet and speakers? *(Choose one or more)*
 - □ Home
 - Work
 - Other

APPENDIX L

COURSE EVALUATION ITEMS

1. The course encouraged me to reflect upon my previous experiences.

strongly disagree disagree neither agree nor disagree agree strongly agree 2. The course content was relevant to my job.

strongly disagree disagree neither agree nor disagree agree strongly agree3. Concepts and material from the course will be incorporated into my nursing practice.

strongly disagree disagree neither agree nor disagree agree strongly agree 4. The content was interesting.

strongly disagree disagree neither agree nor disagree agree strongly agree 5. The learning activities fostered my growth.

strongly disagree disagree neither agree nor disagree agree strongly agree6. The content was offered in multiple ways which promoted my learning.

strongly disagree disagree neither agree nor disagree agree strongly agree 7. The amount of content in the course was appropriate.

strongly disagree disagree neither agree nor disagree agree strongly agree 8. The amount of interaction with others enrolled in the course was appropriate.

strongly disagree disagree neither agree nor disagree agree strongly agree9. The amount of instructor involvement and interaction was appropriate.

strongly disagree disagree neither agree nor disagree agree strongly agree 9. Material was presented in a visually appealing way.

strongly disagree disagree neither agree nor disagree agree strongly agree 10. Overall, the course was set up in a way that was easy to use.

strongly disagree disagree neither agree nor disagree agree strongly agree 11. The length of each module was appropriate.

strongly disagree disagree neither agree nor disagree agree strongly agree 12. The length of the total course was appropriate.

strongly disagree disagree neither agree nor disagree agree strongly agree 13. The course objectives were clear.

strongly disagree disagree neither agree nor disagree agree strongly agree 14. Overall, I was satisfied with the course.

strongly disagree disagree neither agree nor disagree agree strongly agree 15. I would recommend this course to others.

strongly disagree disagree neither agree nor disagree agree strongly agree 16. What was your favorite part of the course?

17. What aspects of the course would you like to see changed or revised?

18. What technical issues did you encounter? How were they resolved?

APPENDIX M

TABLE OF COURSE EVALUATION FINDINGS BY COURSE

| Course Evaluation Question | Soc | ially | Socially Isolated | |
|---|--------|------------|-------------------|------|
| | Inter | active | | |
| | % A or | mode | % A or | mode |
| | SA | | SA | |
| | 100 | G A | 100 | C A |
| Course encouraged reflection | 100 | SA | 100 | SA |
| Course content relevant | 100 | SA | 100 | SA |
| Content was interesting | 100 | SA | 91 | SA/A |
| Learning activities fostered growth | 90 | SA | 86 | SA |
| Content offered in multiple ways | 95 | А | 91 | SA |
| Amount of course content appropriate | 81 | SA | 86 | А |
| Amount of interaction with others appropriate | 81 | А | 23 | Ν |
| Amount of instructor interaction appropriate | 90 | А | 59 | А |
| Material visually appealing | 81 | А | 91 | А |
| Course easy to use | 90 | А | 95 | А |
| Module length appropriate | 81 | SA | 100 | А |
| Course length appropriate | 81 | SA | 91 | А |
| Course objectives clear | 95 | SA | 95 | SA |
| Overall satisfied with course | 95 | SA | 90 | А |
| Would recommend course to others | 81 | SA | 82 | А |
| Course concepts will be incorporated into | 100 | SA | 95 | SA |
| nursing practice | | | | |

Course Evaluation Responses by Course

Note. A = agree and SA = strongly agree, SA/A = equal number of agree and strongly agree responses, N = neither agree nor disagree.

APPENDIX N

PERMISSION TO USE ISFOLE

Request for Permission

2 messages

Ella Heitzler <heitzler@unlv.nevada.edu> To: sjohnson@illinois.edu Bcc: ella_thomasrn@yahoo.com Tue, Sep 28, 2010 at 3:07 PM

Dear Dr. Johnson,

I am a PhD student at the University of Nevada, Las Vegas, School of Nursing. I am preparing for my dissertation and would like to use the Instructional Strategy Framework for Online Learning Environments and its associated figure you published in New Directions for Adult and Continuing Education in 2003. I will be exploring online continuing education interventions for nurses. May I have your permission to use your framework in my dissertation? Thank you in advance for your consideration.

Sincerely, Ella Heitzler

Johnson, Scott D <sjohnson@illinois.edu> To: Ella Heitzler <heitzler@unlv.nevada.edu>

Certainly. Good luck with your research. Scott

Scott D. Johnson | CIO

Associate Dean and Director, International Programs College of Education

Professor, Human Resource Education

University of Illinois at Urbana-Champaign 142 Education Building 1310 South Sixth Street Champaign, IL 61820 Voice: (217) 244-7005 Fax: (217) 244-0390 E-mail: <u>sjohnson@illinois.edu</u> Tue, Sep 28, 2010 at 3:10 PM

APPENDIX O

PERMISSION TO USE 3-D MODEL FIGURE



https://s100.copyright.com/AppDispatchServlet

6/17/2011

APPENDIX P

PERMISSION TO USE ISFOLE FIGURE

Rightslink Printable License

Page 1 of 4

JOHN WILEY AND SONS LICENSE TERMS AND CONDITIONS

Jun 17, 2011

This is a License Agreement between Ella T Heitzler ("You") and John Wiley and Sons ("John Wiley and Sons") provided by Copyright Clearance Center ("CCC"). The license consists of your order details, the terms and conditions provided by John Wiley and Sons, and the payment terms and conditions.

All payments must be made in full to CCC. For payment instructions, please see information listed at the bottom of this form.

| License Number | 2691440623231 |
|---------------------------------------|--|
| License date | Jun 17, 2011 |
| Licensed content publisher | John Wiley and Sons |
| Licensed content publication | New Directions for Adult & Continuing Education |
| Licensed content title | An instructional strategy framework for online learning environments |
| Licensed content author | Scott D. Johnson, Steven R. Aragon |
| Licensed content date | Dec 1, 2003 |
| Start page | 31 |
| End page | 43 |
| Type of use | Dissertation/Thesis |
| Requestor type | University/Academic |
| Format | Print and electronic |
| Portion | Figure/table |
| Number of figures/tables | 1 |
| Number of extracts | |
| Original Wiley figure/table number(s) | Figure 3.1. |
| Will you be translating? | No |
| Order reference number | |
| Total | 0.00 USD |
| Terms and Conditions | |

TERMS AND CONDITIONS

This copyrighted material is owned by or exclusively licensed to John Wiley & Sons, Inc. or one of its group companies (each a "Wiley Company") or a society for whom a Wiley Company has exclusive publishing rights in relation to a particular journal (collectively "WILEY"). By clicking "accept" in connection with completing this licensing transaction, you agree that the following terms and conditions apply to this transaction (along with the billing and payment terms and conditions established by the Copyright Clearance Center Inc., ("CCC's Billing and Payment terms and conditions"), at the time that you opened your Rightslink account (these are available at any time at http://myaccount.copyright.com)

https://s100.copvright.com/App/PrintableLicenseFrame.isp?publisherID=140&licenseID=... 6/17/2011

Terms and Conditions

1. The materials you have requested permission to reproduce (the "Materials") are protected by copyright.

2. You are hereby granted a personal, non-exclusive, non-sublicensable, non-transferable, worldwide, limited license to reproduce the Materials for the purpose specified in the licensing process. This license is for a one-time use only with a maximum distribution equal to the number that you identified in the licensing process. Any form of republication granted by this licence must be completed within two years of the date of the grant of this licence (although copies prepared before may be distributed thereafter). The Materials shall not be used in any other manner or for any other purpose. Permission is granted subject to an appropriate acknowledgement given to the author, title of the material/book/journal and the publisher and on the understanding that nowhere in the text is a previously published source acknowledged for all or part of this Material. Any third party material is expressly excluded from this permission.

3. With respect to the Materials, all rights are reserved. Except as expressly granted by the terms of the license, no part of the Materials may be copied, modified, adapted (except for minor reformatting required by the new Publication), translated, reproduced, transferred or distributed, in any form or by any means, and no derivative works may be made based on the Materials without the prior permission of the respective copyright owner. You may not alter, remove or suppress in any manner any copyright, trademark or other notices displayed by the Materials. You may not license, rent, sell, loan, lease, pledge, offer as security, transfer or assign the Materials, or any of the rights granted to you hereunder to any other person.

4. The Materials and all of the intellectual property rights therein shall at all times remain the exclusive property of John Wiley & Sons Inc or one of its related companies (WILEY) or their respective licensors, and your interest therein is only that of having possession of and the right to reproduce the Materials pursuant to Section 2 herein during the continuance of this Agreement. You agree that you own no right, title or interest in or to the Materials or any of the intellectual property rights therein. You shall have no rights hereunder other than the license as provided for above in Section 2. No right, license or interest to any trademark, trade name, service mark or other branding ("Marks") of WILEY or its licensors is granted hereunder, and you agree that you shall not assert any such right, license or interest with respect thereto.

5. NEITHER WILEY NOR ITS LICENSORS MAKES ANY WARRANTY OR REPRESENTATION OF ANY KIND TO YOU OR ANY THIRD PARTY, EXPRESS, IMPLIED OR STATUTORY, WITH RESPECT TO THE MATERIALS OR THE ACCURACY OF ANY INFORMATION CONTAINED IN THE MATERIALS, INCLUDING, WITHOUT LIMITATION, ANY IMPLIED WARRANTY OF MERCHANTABILITY, ACCURACY, SATISFACTORY QUALITY, FITNESS FOR A PARTICULAR PURPOSE, USABILITY, INTEGRATION OR NON-INFRINGEMENT AND ALL SUCH WARRANTIES ARE HEREBY EXCLUDED BY WILEY AND ITS LICENSORS AND WAIVED BY YOU.

6. WILEY shall have the right to terminate this Agreement immediately upon breach of this Agreement by you.

7. You shall indemnify, defend and hold harmless WILEY, its Licensors and their respective directors, officers, agents and employees, from and against any actual or threatened claims, demands, causes of action or proceedings arising from any breach of this Agreement by you.

8. IN NO EVENT SHALL WILEY OR ITS LICENSORS BE LIABLE TO YOU OR ANY OTHER PARTY OR ANY OTHER PERSON OR ENTITY FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL, INDIRECT, EXEMPLARY OR PUNITIVE DAMAGES, HOWEVER CAUSED, ARISING OUT OF OR IN CONNECTION WITH THE DOWNLOADING, PROVISIONING, VIEWING OR USE OF THE MATERIALS REGARDLESS OF THE FORM OF ACTION, WHETHER FOR BREACH OF CONTRACT, BREACH OF WARRANTY, TORT, NEGLIGENCE, INFRINGEMENT OR OTHERWISE (INCLUDING, WITHOUT LIMITATION, DAMAGES BASED ON LOSS OF PROFITS, DATA, FILES, USE, BUSINESS OPPORTUNITY OR CLAIMS OF THIRD PARTIES), AND WHETHER OR NOT THE PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES. THIS LIMITATION SHALL APPLY NOTWITHSTANDING ANY FAILURE OF ESSENTIAL PURPOSE OF ANY LIMITED REMEDY PROVIDED HEREIN.

https://s100.copyright.com/App/PrintableLicenseFrame.jsp?publisherID=140&licenseID=... 6/17/2011

9. Should any provision of this Agreement be held by a court of competent jurisdiction to be illegal, invalid, or unenforceable, that provision shall be deemed amended to achieve as nearly as possible the same economic effect as the original provision, and the legality, validity and enforceability of the remaining provisions of this Agreement shall not be affected or impaired thereby.

10. The failure of either party to enforce any term or condition of this Agreement shall not constitute a waiver of either party's right to enforce each and every term and condition of this Agreement. No breach under this agreement shall be deemed waived or excused by either party unless such waiver or consent is in writing signed by the party granting such waiver or consent. The waiver by or consent of a party to a breach of any provision of this Agreement shall not operate or be construed as a waiver of or consent to any other or subsequent breach by such other party.

11. This Agreement may not be assigned (including by operation of law or otherwise) by you without WILEY's prior written consent.

12. Any fee required for this permission shall be non-refundable after thirty (30) days from receipt.

13. These terms and conditions together with CCC's Billing and Payment terms and conditions (which are incorporated herein) form the entire agreement between you and WILEY concerning this licensing transaction and (in the absence of fraud) supersedes all prior agreements and representations of the parties, oral or written. This Agreement may not be amended except in writing signed by both parties. This Agreement shall be binding upon and inure to the benefit of the parties' successors, legal representatives, and authorized assigns.

14. In the event of any conflict between your obligations established by these terms and conditions and those established by CCC's Billing and Payment terms and conditions, these terms and conditions shall prevail.

15. WILEY expressly reserves all rights not specifically granted in the combination of (i) the license details provided by you and accepted in the course of this licensing transaction, (ii) these terms and conditions and (iii) CCC's Billing and Payment terms and conditions.

16. This Agreement will be void if the Type of Use, Format, Circulation, or Requestor Type was misrepresented during the licensing process.

17. This Agreement shall be governed by and construed in accordance with the laws of the State of New York, USA, without regards to such state's conflict of law rules. Any legal action, suit or proceeding arising out of or relating to these Terms and Conditions or the breach thereof shall be instituted in a court of competent jurisdiction in New York County in the State of New York in the United States of America and each party hereby consents and submits to the personal jurisdiction of such court, waives any objection to venue in such court and consents to service of process by registred or certified mail, return receipt requested, at the last known address of such party. BY CLICKING ON THE "I ACCEPT" BUTTON, YOU ACKNOWLEDGE THAT YOU HAVE READ AND FULLY UNDERSTAND EACH OF THE SECTIONS OF AND PROVISIONS SET FORTH IN THIS AGREEMENT AND THAT YOU ARE IN AGREEMENT WITH AND ARE WILLING TO ACCEPT ALL OF YOUR OBLIGATIONS AS SET FORTH IN THIS AGREEMENT.

v1.4

Gratis licenses (referencing \$0 in the Total field) are free. Please retain this printable license for your reference. No payment is required.

If you would like to pay for this license now, please remit this license along with your payment made payable to "COPYRIGHT CLEARANCE CENTER" otherwise you will be invoiced within 48 hours of the license date. Payment should be in the form of a check or money order referencing your account number and this invoice number

https://s100.copyright.com/App/PrintableLicenseFrame.isp?publisherID=140&licenseID=... 6/17/2011

Rightslink Printable License

RLNK11005978.

Once you receive your invoice for this order, you may pay your invoice by credit card. Please follow instructions provided at that time.

Make Payment To: Copyright Clearance Center Dept 001 P.O. Box 843006 Boston, MA 02284-3006

For suggestions or comments regarding this order, contact Rightslink Customer Support: <u>customercare@copyright.com</u> or +1-877-622-5543 (toll free in the US) or +1-978-646-2777.

https://s100.copyright.com/App/PrintableLicenseFrame.jsp?publisherID=140&licenseID=... 6/17/2011

REFERENCES

- Agency for Healthcare Research and Quality, U.S. Department of Health and Human Services. (2007). *National healthcare disparities report, 2007* (AHRQ Publication No. 08-0041). Retrieved from http://www.ahrq.gov/qual/nhdr07 /nhdr07.pdf
- Alexander, G. R., & Kotelchuck, M. (2001). Assessing the role and effectiveness of prenatal care: History, challenges, and directions for future research. *Public Health Reports*, 116, 306-316.
- Ali, N. S., Hodson-Carlton, K., & Ryan, M. (2004). Students' perceptions of online learning: Implications for teaching. *Nurse Educator*, 29, 111-115.
- American Association of Colleges of Nursing. (2008). *Cultural competency in baccalaureate nursing education*. Retrieved from http://www.aacn.nche. edu/education/pdf/competency.pdf
- American Institutes for Research. (2002). *Teaching cultural competence in health care: A review of current concepts, policies and practices* (Contract No: 282-98-0029).
 Report prepared for the Office of Minority Health. Washington, DC: Author.
- Anderson, T. (2008). *The theory and practice of online learning* (2nd ed.). Edmonton, AB: AU Press.
- Arnella, C., Yox, S., & Eckstein, D. S., & Ousley, A. (2010). Expanding the reach of a cancer palliative care curriculum through web-based dissemination: A publicprivate collaboration. *Journal of Cancer Education*, 25, 418-421. doi:10.1007/s13187-010-0066-1
- Association of Women's Health, Obstetric, and Neonatal Nurses. (2011). *About us.* Retrieved from http://awhonn.org/awhonn/content.do;jsessionid=483395869578 EB9987A2714FCB2C0395?name=10_AboutUs/10_AboutUs_landing.htm
- Association of Women's Health, Obstetric, and Neonatal Nurses. (n.d.). Vision, mission, and values. Retrieved from www. Awhonn.org/awhonn/content.do?name=10_ AboutUs/10A-MissionVisionValues.htm
- Atack, L. (2003). Becoming a web-based learner: Registered nurses' experiences. *Journal of Advanced Nursing*, 44, 289-297.
- Atack, L., & Rankin, J. (2002). A descriptive study of registered nurses' experiences with web-based learning. *Journal of Advanced Nursing*, 40, 457-465.

- Bangert, A. W. (2008). The development and validation of the Student Evaluation of Online Teaching Effectiveness. *Computers in the Schools*, 25(1-2), 25-47. doi:10.1080/07380560802157717
- Beatty, R. M. (2001). Continuing professional education, organizational support, and professional competence: Dilemmas of rural nurses. *The Journal of Continuing Education in Nursing*, *32*, 203-209.
- Beaudoin, M. F., Kurtz, G., & Eden, S. (2009). Experiences and opinions of e-learners: What works, what are the challenges, and what competencies ensure successful online learning. *Interdisciplinary Journal of E-Learning and Learning Objects*, 5, 275-289.
- Bender, T. (2003). *Discussion-based online teaching to enhance student learning: Theory, practice and assessment.* Sterling, VA: Stylus Publishing.
- Bernal, H., & Froman, R. (1987). The confidence of community health nurses in caring for ethnically diverse populations. *IMAGE: Journal of Nursing Scholarship*, 19, 201-203.
- Bernal, H., & Froman, R. (1993). Influences on the cultural self-efficacy of community health nurses. *Journal of Transcultural Nursing*, 4, 24-31. doi:10.1177/104365 969300400205
- Bernhardt, J. M., Runyan, C. W., Bou-Saada, I., & Felter, E. M. (2003). Implementation and evaluation of a web-based continuing education course in injury prevention and control. *Health Promotion Practice*, 4(2), 120-128. doi:10.1177/15248399 02250758
- Billings, D. M. (2000). A framework for assessing outcomes and practices in Web-based courses in nursing. *Journal of Nursing Education*, *39*, 60-67.
- Billings, D. M., Connors, H. R., & Skiba, D. J. (2001). Benchmarking best practices in Web-based nursing courses. Advances in Nursing Science, 23(3), 41-52.
- Blood-Siegfried, J. E., Short, N. M., Rapp, C. G., Hill, E., Talbert, S., Skinner, J., ...Goodwin, L. (2008). A rubric for improving the quality of online courses. *International Journal of Nursing Education Scholarship*, 5(1), 1-13. Retrieved from http://www.bepress.com/ijnes/vol5/iss1/art34
- Bolliger, D. U., & Martindale, T. (2004). Key factors for determining student satisfaction in online courses. *International Journal on E-Learning*, *3*(1), 61-67.
- Bonaparte, B. H. (1979). Ego defensiveness, open-closed mindedness, and nurses; attitude toward culturally different patients. *Nursing Research*, 28, 166-172.

- Booth, A., Carroll, C., Papaioannou, D., Sutton, A., & Wong, R. (2009). Applying findings from a systematic review of workplace-based e-learning: Implications for health information professionals. *Health Information and Libraries Journal*, 26, 4-21. doi: 10.1111/j.1471-1842.2008.00834.x
- Bowers, P. (n.d.). *Cultural perspectives in childbearing*. Retrieved from http://ce.nurse. com/ CE263-60/
- Brathwaite, A. E. C. (2004). Evaluation of a cultural competence educational intervention (Doctoral dissertation). Retrieved from ProQuest. (978-0-494-22071-9)
- Brathwaite, A. E. C. (2005). Evaluation of a cultural competence course. *Journal of Transcultural Nursing*, *16*, 361-369. doi:10.1177/1043659605278941
- Buhmann, J., & Johnson, B. (2009). *Designing Web pages for Web courses*. Unpublished manuscript, School of Nursing, Indiana University, Indianapolis, Indiana.
- Bussema, E., & Nemec, P. (2006) Training to increase cultural competence. *Psychiatric Rehabilitation Journal*, *30*(1), 71-72.
- Callister, L. C. (2005). What has the literature taught us about culturally competent care of women and children? *The American Journal of Maternal/Child Nursing*, *30*, 380-388.
- Callister, L. C., Getmanenko, N., Garvrish, N., Eugenevna, M. O., Vladimirova, Z. N., Lassetter, J., & Turkina, N. (2007). Giving birth: The voices of Russian women. *The American Journal of Maternal/Child Nursing*, *32*, 18-24.
- Camarota, S. A. (2005). *Births to immigrants in America, 1970 to 2002*. Retrieved from Center for Immigration Studies Website: http://www.cis.org/articles/2005/ back805.pdf
- Camarota, S. A. (2007). *Immigrants in the United States, 2007: A profile of American's foreign-born population.* Retrieved from Center for Immigration Studies Website: www.cis.org/articles/2007/back1007.pdf
- Campinha-Bacote, J. (1999). A model and instrument for addressing cultural competence in health care. *Journal of Nursing Education, 38,* 203-207.
- Campinha-Bacote, J. (2003). Many faces: Addressing diversity in health care. *Online Journal of Issues in Nursing*, 8(1). Retrieved from www.cinahl.com/cgibin/refsvc?jid=1331&accno=2003153459
- Campinha-Bacote, J. (2006). Cultural competence in nursing curricula: How are we doing 20 years later? *Journal of Nursing Education*, 45, 243-244.

- Campinha-Bacote, J. (2007). *The process of cultural competence in the delivery of healthcare services: The journey continues*. Cincinnati, OH: Transcultural C.A.R.E. Associates.
- Capell, J., Veenstra, G., & Dean, E. (2007). Cultural competence in healthcare: Critical analysis of the construct, its assessment and implications. *The Journal of Theory Construction and Testing*, 11(1), 30-37.
- Casebeer, L., Engler, S., Bennett, N., Irvine, M., Sulkes, D., DesLauriers, M., & Zhang, S. (2008). A controlled trial of the effectiveness of internet continuing medical education. *BMC Medicine*, 6(37). doi:10.1186/1741-7015/6/37
- Casebeer, L., Kristofco, R. E., Strasser, S., Reilly, M., Krishnamoorthy, P., Rabin, A.,... Myers, L. (2004). Standardizing evaluation of online continuing medical education: Physician knowledge, attitudes, and reflection on practice. *The Journal* of Continuing Education in the Health Professions, 24, 68-75.
- Castro, A., & Ruiz, E. (2009). The effects of nurse practitioner cultural competence on Latina patient satisfaction. *Journal of the American Academy of Nurse Practitioners, 21,* 278-286. doi:10.1111/j.1745-7599.2009.00406.x
- Cercone, K. (2008). Characteristics of adult learners with implications for online learning design. *Association for the Advancement of Computing in Education Journal, 16,* 137-159.
- Cesario, S. K. (2001). Care of the Native American woman: Strategies for practice, education, and research. *Journal of Obstetric, Gynecologic, & Neonatal Nursing, 30*, 13-19.
- Christianson, L., Tiene, D., & Luft, P. (2002). Web-based teaching in undergraduate nursing programs. *Nurse Educator*, *6*, 276-282.
- Cobb, S. C. (2004). Internet continuing education for health care professionals: An integrative review. *The Journal of Continuing Education in the Health Professions, 24,* 171-180.
- Coffman, M. J. (2004). Cultural caring in nursing practice: A meta-synthesis of qualitative research. *Journal of Cultural Diversity*, *11*, 100-109.
- Cohen, J. (1988). *Statistical power analysis for the behavioral sciences* (2nd ed.). Hillsdale, NJ: Lawrence Erlbaum Associates, Inc.
- Conrad, R., & Donaldson, J. A. (2004). *Engaging the online learner: Activities and resources for creative instruction*. San Francisco, CA: John Wiley & Sons.

- Cooper, M., Grywalski, M., Lamp, J., Newhouse, L., & Studlien, R. (2007). Enhancing cultural competence: A model for nurses. *Nursing for Women's Health*, 11, 148-159. doi:10.1111/j1751-486x.2007.00130.x
- D'Andrea, M., Daniels, J., & Heck, R. (1991). Evaluating the impact of multicultural counseling training. *Journal of Counseling and Development*, 70, 143-150.
- Davis, R. E. (2001). The postpartum experience for southeast Asian women in the United States. *The American Journal of Maternal/Child Nursing*, *26*, 208-213.
- Davis, S. L., & Finney, S. J. (2006). A factor analytic study of the Cross-Cultural Adaptability Inventory. *Educational and Psychological Measurement*, 66, 318-330. doi:10.1177/0013164405278571
- Dean, R. A. K. (2010). Cultural competence: Nursing in a multicultural society. *Nursing for Women's Health*, *14*, 51-60. doi:10.1111/j.1751-486X.201001507.x
- Dimitrov, D. M., & Rumrill, P. D. (2003). Pretest-posttest designs and measurement of change. Work: A journal of prevention, assessment, and rehabilitation, 20, 159-165.
- Doorenbos, A. Z., & Schim, S. M. (2004). Assessing cultural competence in hospice. *American Journal of Hospice and Palliative Care, 21,* 28-32. doi:10.1177/104990910402100108
- Doorenbos, A. Z., Schim, S. M., Benkert, R., & Borse, N. N. (2005). Psychometric evaluation of the cultural competence assessment instrument among healthcare providers. *Nursing Research*, *54*, 324-331.
- Dunet, D. O., Reyes, M., Grossniklaus, D., Volansky, M., & Blanck, H. M. (2008). Using evaluation to guide successful development of an online training course for healthcare professionals. *Journal of Public Health Management Practice*, 14(1), 66-75.
- Dykman, C. A., & Davis, C. K. (2008). Online education forum- part three: A quality online educational experience. *Journal of Information Systems Education, 19*, 281-289.
- Edwards, K. (2003). Increasing cultural competence and decreasing disparities in health. *Journal of Cultural Diversity*, 10, 111-112.
- Faul, F., Erdfelder, E., Lang, A., & Buchner, A. (2010). G*Power 3, release 3.1.2 [software]. Retrieved from http://www.psycho.uni-duesseldorf.de/ abteilungen/aap/gpower3/ download-and-register

- Fortier, J. P., & Bishop, D. (2004). *Setting the agenda for research on cultural competence in health care* (AHRQ report No. 00T061242). Retrieved from http://www.ahrq.gov/research/cultural.htm
- Ferguson, J. M., & DeFelice, A. E. (2010). Length of online course and student satisfaction, perceived learning, and academic performance. *The International Review of Research in Open and Distance Learning*, 11(2), 73-84. Retrieved from http://www.irrodl.org/index.php/irrodl/article/view/772/1547
- Gagnon, L., Bakker, D., Montgomery, P., & Palkovits, J. (2010). Nurse autonomy in cancer care. *Cancer Nursing*, *33*(3), 21-28.
- Galanti, G. (2008). *Caring for patients from different cultures* (4th ed.). Philadelphia, PA: University of Pennsylvania Press.
- Gallien, T., & Oomen-Early, J. (2008). Personalized versus collective instructor feedback in the online classroom: Does type of feedback affect student satisfaction, academic performance and perceived connectedness with the instructor? *International Journal of E-Learning*, 7, 463-476.
- Gardner, D. (2010, March 11). America nears 'tipping point' where babies born to minority parents outnumber whites for first time. *Daily Mail*. Retrieved from www.dailymail.co.uk/new/worldnews/article-1257110
- Garson, G. D. (2009). *Univariate GLM, ANOVA, and ANCOVA*. Unpublished manuscript, College of Humanities and Social Sciences, North Carolina State University, Raleigh, North Carolina. Retrieved from http://faculty.chass.ncsu.edu /garson/PA765/anova.htm
- Gerace, L. M., & Salimbene, S. (n.d.). *Cultural competence for today's healthcare* professionals, part 1: Cultural and women's health. Retrieved from http://ce. nurse.com/AH398-60
- Giger, J. N., & Davidhizar, R. (2002). The Giger and Davidhizar transcultural assessment model. *Journal of Transcultural Nursing*, 13, 185-188.
- Giger, J. N., & Davidhizar, R. E. (2008). *Transcultural nursing: Assessment and intervention* (5th ed.). St. Louis, MO: Mosby.
- Gillani, B. B. (2003). *Learning theories and the design of e-learning environments*. Lanham, MD: University Press of America Inc.
- Girden, E. R. (1992). ANOVA repeated measures. Newbury Park, CA: Sage Publications Inc.

- Gliner, J. A., Morgan, G. A., & Harmon, R. J. (2003). Pretest-posttest comparison group designs: Analysis and interpretation. *Journal of the American Academy of Child* and Adolescent Psychiatry, 43, 500-503. doi:0.1097/01.CHI.0000046809 .95464.BE
- Guan, J., Tregonning, S., & Keenan, L. (2008). Social interaction and participation: Formative evaluation of online CME modules. *Journal of Continuing Education in the Health Professions, 28,* 172-179. doi:10.1002/chp
- Guerrieri, R. (2010). Learn, grow, and bloom by joining a professional association. *Nursing*, *40*(5), 47-48. doi:10.1097/01.Nurse.0000371128.38511.od
- Halstead, J. A., & Billings, D. M. (2009). Teaching and learning in online learning communities. In D. M. Billings & J. A. Halstead (Eds.), *Teaching in nursing: A* guide for faculty (3rd ed., pp. 369-390). St. Louis, MO: Saunders.
- Health Resources and Services Administration, U.S. Department of Health and Human Services. (2008). *The registered nurse population: Initial findings from the 2008 national sample survey of registered nurses*. Retrieved from http://bhpr.hrsa. gov/healthworkforce/rnsurveys/rnsurveyinitial2008.pdf
- Hernandez, I. F. (2006). Promoting exclusive breastfeeding for Hispanic women. *The American Journal of Maternal/Child Nursing*, *31*, 318-324.
- Higgins, P. G., & Learn, C. D. (1999). Health practices of adult Hispanic women. *Journal of Advanced Nursing*, 29, 1105-1112.
- Huckstadt, A., & Hayes, K. (2005). Evaluation of interactive online courses for advanced practice nurses. *Journal of the American Academy of Nurse Practitioners*, 17(3), 85-89.
- Hunter, J. L. (2008). Applying constructivism to nursing education in cultural competence: A course that bears repeating. *Journal of Transcultural Nursing*, *19*, 354-362. doi:10.1177/1043659608322421
- Hunter, J. L., & Krantz, S. (2010). Constructivism in cultural competence education. Journal of Nursing Education, 49, 207-214. doi:10.3928/01464834-20100115-06
- InFocus Marketing. (2011). Association of Women's Health, Obstetric, and Neonatal Nurses. Retrieved from http://www.infocusmarketing.com/datacards/datacards/dc. aspx?id=160&type=1
- Jackson, A. K. (2007). Cultural competence in health visiting practice: A baseline survey. *Community Practitioner*, 80(2), 17-22.

- Jeffreys, M. R., & Smodlaka, I. (1998). Exploring the factorial composition of the transcultural self-efficacy tool. *International Journal of Nursing Studies*, 35, 217-225.
- Jibaja-Rusth, M. L., Kingery, P. M., Holocomb, J. D., Buckner, W. P., & Pruitt, B. E. (1994). Development of a multicultural sensitivity scale. *Journal of Health Education*, 25, 350-357.
- Johnson, C. A. (2005). Understanding the cultural care practices of rural immigrant Mexican women (Doctoral dissertation). Retrieved from http://etd1.library.duq. edu/theses/available/etd-07272005-144536/
- Johnson, S. D., & Aragon, S. R. (2003). An instructional strategy framework for online learning environments. New Directions for Adult and Continuing Education, 100, 31-43.
- Jones, L. R. (2009). *Hospital based nurse cultural competence program evaluation: A mixed methodological study* (Doctoral dissertation). Retrieved from ProQuest. (3378815)
- Kelley, C., & Meyers, J. (2010). The Cross-Cultural Adaptability Inventory. Retrieved from http://ccaiassess.com/CCAI_Tools.html
- Kelly, F., & Papadopoulos, I. (2009). Enhancing the cultural competence of healthcare professionals through an online course. *Diversity in Health Care*, *6*, 77-84.
- Khanna, S. K., Cheyney, M., & Engle, M. (2009). Cultural competency in health care: Evaluating the outcomes of a cultural competency training among health care professionals. *Journal of the National Medical Association*, *101*, 888-892.
- Kim-Godwin, Y. S. (2003). Postpartum beliefs and practices among non-western cultures. *The American Journal of Maternal/Child Nursing*, 28, 74-80.
- Kordel, R. (2008). Information presentation for effective e-learning. *Educause Quarterly*, *4*, 10-12.
- Kramer, M., Maguire, P., Schmalenberg, C. E., Andrews, B., Burke, R., Chmielewski, L.,...Tachibana, C. (2007). Structures enabling clinical autonomy. *The Journal of Nursing Administration*, 37, 41-52.
- Kubsch, S., Henniges, A., Lorenzoni, N., Eckardt, S., & Oleniczak, S. (2003). Factors influencing accruement of contact hours for nurses. *The Journal of Continuing Education in Nursing*, 34, 205-212.
- Ladany, N., Inman, A. G., Constantine, M. G., & Hofheinz, E. W. (1997). Supervisee multicultural case conceptualization ability and self-reported multicultural

competence as functions of supervisee racial identity and supervisor focus. *Journal of Counseling Psychology*, 44, 284-293.

- Lal, A. (2010). *The impact of an educational offering on the cultural competence of acute care registered nurses in an urban Midwest hospital in the United States* Doctoral dissertation). Retrieved from ProQuest. (UMI No. 3416628)
- Lee, C. A., Anderson, M. A., & Hill, P. D. (2006). Cultural sensitivity education for nurses: A pilot study. *The Journal of Continuing Education in Nursing*, 37, 137-141.
- Leininger, M. (1997). Overview of the theory of culture care with the ethnonursing research method. *Journal of Transcultural Nursing*, *8*, 32-52. doi:10.1177/1043659697000800205
- Leininger, M. M., & McFarland, M. R. (2006). *Culture care diversity and universiality:* A worldwide nursing theory (2nd ed.). Sudbury, MA: Jones and Bartlett.
- Lipson, J. G., & Desantis, L. A. (2007). Current approaches to integrating elements of cultural competence in nursing education. *Journal of Transcultural Nursing*, 18, 10S-20S. doi: 10.1177/1043659606295498
- Luconi, F. (2008). Exploring rural family physicians' learning from a web-based continuing medical education program on Alzheimer's disease: A pilot study (Doctoral dissertation, McGill University, Montreal, Canada). Retrieved from http://digitool.library.mcgill.ca/webclient/StreamGate?folder_id=0&dvs=1307903 231454~965
- MacDonald, C. J., & Walton, R. (2007). E-learning education solutions for caregivers in long-term care (LTC) facilities: New possibilities. *Education for Health*, 20(3), 1-9. Retrieved from http://www.educationforhealth.net/
- Majumdar, B., Browne, G., Roberts, J., & Carpio, B. (2004). Effects of cultural sensitivity training on health care provider attitudes and patient outcomes. *Journal of Nursing Scholarship, 36*, 161-166.
- Mancuso-Murphy, J. (2007). Distance education in nursing: An integrated review of online nursing students' experiences with technology-delivered instruction. *Journal of Nursing Education*, *46*, 252-260.
- Mattson, S. (2000). Providing care for the changing face of the U.S. AWHONN Lifelines, 4(3), 49-52.
- McHenry, D. M. (2007). A growing challenge: Patient education in a diverse America. Journal for Nurses in Staff Development, 23, 83-88. doi:10.1097/01.NND. 0000266616.01187.75

- Moore, M. L., Moos, M., & Callister, L. C. (2010). Cultural competence: An essential journey for perinatal nurses. Retrieved from http://www.marchofdimes.com/nursing/modnemedia/othermedia/slides/cultural_comp2010.ppt
- Morris, L. V., & Finnegan, C. L. (2008). Best practices in predicting and encouraging student persistence and achievement online. *Journal of College Student Retention*, 10(1), 55-64. doi:10.2190/CS.10.1.e
- Muilenburg, L. Y., & Berge, Z. L. (2005). Student barriers to online learning: A factor analytic study. *Distance Education*, 26(1), 29-48.doi:10.1080/015879105000 81269
- Munro, B. H. (2005). *Statistical methods for health care research* (5th ed.). Philadelphia, PA: Lippincott Williams & Wilkins.
- Nalle, M. A., Wyatt, T. H., & Myers, C. R. (2010). Continuing education needs of nurses in a voluntary continuing nursing education state. *The Journal of Continuing Education in Nursing*, 41, 107-115. doi:10.3928/00220124-20100224-03
- Noble, L. M., Noble, A., Geiss, D. M., & Hand, I. L. (2006). Cultural competence of healthcare professionals caring for breastfeeding mothers in urban areas [Abstract]. *Breastfeeding Medicine; the Official Journal of the Academy of Breastfeeding Medicine, 1*(3), 189.
- Office of Disease Prevention and Health Promotion. (2009). *Healthy people 2020 framework*. Retrieved from www.healthypeople/gov/hp2020/Objectives/ framework.aspx
- Office of Minority Health (2001). *National standards for culturally and linguistically appropriate services in health care: Executive summary* (Contract No: 282-99-0039). Retrieved from http://minorityhealth.hhs.gov/assets/pdf/checked/executive .pdf
- Office of Minority Health. (2007). Culturally competent nursing care: A cornerstone for caring. Retrieved from https://ccnm.thinkculturalhealth.org
- Office of Minority Health. (2008). Initiatives. Retrieved from http://minorityhealth.hhs. gov/templates/browse.aspx?lvl=3&lvlid=17
- O'Neil, C. A. (2009). Designing the online learning environment. In C. A. O'Neil, C. A. Fisher, & S. K. Newbold (Eds.), *Developing online learning environments in nursing education* (2nd ed., pp. 83-98). New York, NY: Springer.
- Paez, K. A., Allen, J. K., Carson, K. A., & Cooper, L.A. (2008). Provider and clinical cultural competence in primary care settings. *Social Science and Medicine*, 66, 1204-1216. doi:10.1016/j.socscimed.2007.11.027.

- Pallant, J. (2007). SPSS survival manual: A step by step guide to data analysis using SPSS for Windows (3rd ed.). New York, NY: McGraw-Hill.
- Papadopoulos, I., Tilki, M., & Ayling, S. (2008). Cultural competence in action for CAMHS: Development of a cultural competence assessment tool and program. *Contemporary Nurse*, 28, 129-140.
- Papadopoulos, I., Tilki, M., & Lees, S. (2004). Promoting cultural competence in healthcare through a research-based intervention in the UK. *Diversity in Health* and Social Care, 1, 107-115.
- Pelayo, M., Cebrian, D., Areosa, A., Agra, Y., Izquierdo, J. V., & Buendia, F. (2011). Effects of online palliative care training on knowledge, attitude and satisfaction of primary care physicians. *BMC Family Practice*, 12(37). doi:10.1186/1471-2296-12-37
- Pena-Shaff, J., Altman, W., & Stephenson, H. (2005). Asynchronous online discussions as a tool for learning: Students' attitudes, expectations, and perceptions. *Journal* of Interactive Learning Research, 16, 409-430.
- Phipps, J. J. (2005). E-Journaling: Achieving interactive education online. *Educase Quarterly*, 1, 62-65.
- Polacek, G., & Martinez, R. (2009). Assessing cultural competence at a local hospital system in the United States. *The Health Care Manager*, 28, 98-110.
- Polit, D. F. (2010). *Statistics and data analysis for nursing research* (2nd ed.). Upper Saddle River, NJ: Pearson Education Inc.
- Polit, D. F., & Beck, C. T. (2008). *Nursing research: Generating and assessing evidence for nursing practice* (8th ed.). Philadelphia, PA: Lippincott Williams & Wilkins.
- Ponterotto, J. G., Gretchen, D., Utsey, S. O., Rieger, B. P., & Austin, R. (2002). A revision of the multicultural counseling awareness scale. *Journal of Multicultural Counseling and Development*, 30, 153-180.
- Pugh, P. A. (2008). Effective educational delivery methods: A meta-analysis of cultural competence education for health care professionals (Doctoral dissertation). Retrieved from ProQuest. (UMI No. 3290661)
- Pullen, D. L. (2006). An evaluative case study of online learning for healthcare professionals. *The Journal of Continuing Education in Nursing*, *37*, 225-232.
- Purnell, L. (2000). A description of the Purnell model for cultural competence. *Journal of Transcultural Nursing*, *11*, 40-46. doi:10.1177/104365960001100107

- Purnell, L. (2002). The Purnell model for cultural competence. *Journal of Transcultural Nursing*, 13, 193-196.
- Purnell, L. D. (2008). The Purnell model for cultural competence. In L. D. Purnell & B. J. Paulanka (Eds.), *Transcultural health care: A culturally competent approach* (3rd ed., pp. 19-55). Philadelphia, PA: F. A. Davis Company.
- Ramsey, R, W., & Clark, C. E. (2009). Teaching and learning at a distance. In D. M. Billings & J. A. Halstead (Eds.), *Teaching in nursing: A guide for faculty* (3rd ed., pp. 351-268). St. Louis, MO: Saunders.
- Rew, L., Becker, H., Cookston, J., Khosropour, S., & Martinez, S. (2003). Measuring cultural awareness in nursing students. *Journal of Nursing Education*, 42, 249-257.
- Reynolds, W. M. (1982). Development of reliable and valid short forms of the Marlowe-Crowne Social Desirability Scale. *Journal of Clinical Psychology*, *38*(1), 119-125.
- Riordan, J., & Gill-Hopple, K. (2001). Breastfeeding care in multicultural populations. Journal of Obstetric, Gynecologic, and Neonatal Nursing, 30, 216-223.
- Roberts, T. G., Irani, T. A., Telg, R. W., & Lundy. L. K. (2005). The development of an instrument to evaluate distance education courses using student attitudes. *The American Journal of Distance Education*, 19(1), 51-64.
- Rooda, L. A. (1993). Knowledge and attitudes of nurses toward culturally different patients: Implications for nursing education. *Journal of Nursing Education*, 32, 209-213.
- Rooda, L., & Gay, G. (1993). Staff development for culturally sensitive nursing care. *Journal of Nursing Staff Development*, 9, 262-265.
- Rovai, A. P. (2002). Sense of community, perceived cognitive learning, and persistence in asynchronous learning networks. *The Internet and Higher Education*, 5, 319-332.
- Rowles, C. J., & Russo, B. L. (2009). Strategies to promote critical thinking and active learning. In D. M. Billings & J. A. Halstead (Eds.), *Teaching in nursing: A guide for faculty* (3rd ed., pp. 238-261). St. Louis, MO: Saunders.
- Sadera, W. A., & Fisher, C. A. (2009). Distance and continuing medical education. In C. A. O'Neil, C. A. Fisher, & S. K. Newbold (Eds.), *Developing online learning environments in nursing education* (2nd ed., pp. 151-166). New York, NY: Springer.

- Sarto, G. (2005). Of disparities and diversity: Where are we? American Journal of Obstetrics and Gynecology, 192, 1188-1195. doi:10.1016/j.ajog.2004.12.065
- Schim, S. M., Doorenbos, A., Benkert, R., & Miller, J. (2007). Culturally congruent care: Putting the puzzle together. *Journal of Transcultural Nursing*, 18, 103-110. doi:10.1177/1043659606298613
- Schim, S. M., Doorenbos, A. Z., & Borse, N. N. (2005). Cultural competence among Ontario and Michigan healthcare providers. *Journal of Nursing Scholarship*, 37, 354-360.
- Schim, S. M., Doorenbos, A. Z., & Borse, N. N. (2006a). Cultural competence among hospice nurses. *Journal of Hospice and Palliative Nursing*, 8, 302-307.
- Schim, S. M., Doorenbos, A. Z., & Borse, N. N. (2006b). Enhancing cultural competence among hospice staff. *American Journal of Hospice & Palliative Medicine*, 23, 404-411. doi:10.1177/1049909103292246
- Schim, S. M., Doorenbos, A. Z., Miller, J., & Benkert, R. (2003). Development of a cultural competence assessment instrument. *Journal of Nursing Measurement*, 11(1), 29-40.
- Schneiderman, J., & Corbridge, S. (2009). Demonstrating the effectiveness of an online, computer-based learning module for arterial blood gas analysis. *Clinical Nurse Specialist, 23*, 151-155.
- Schultz, D. (2004). Cultural competence in psychosocial and psychiatric care: A critical perspective with reference to research and clinical experiences in California, US and in Germany. *Social Work in Health Care, 39*(3), 231-247. doi:10.1300/ J010v39n03.02
- Semenic, S. E., Callister, L. C., & Feldman, P. (2004). Giving birth: The voices of Orthodox Jewish women living in Canada. *Journal of Obstetric, Gynecologic,* and Neonatal Nursing, 33, 80-87. doi:10.1177/0884217503258352
- Shachar, M., & Neumann, Y. (2003). Differences between traditional and distance education academic performances: A meta-analytic approach. *International Review of Research in Open and Distance Learning*, 4(2), 1-20.
- Shade, S. D., & Barber, G. M. (2004). When and where you want it: Continuing education from a distance. *Gerontology and Geriatrics Education*, 24(4), 95-114. doi:10.1300/J021v24n04_07
- Shaffer, C. F. (2002). Factors influencing the access to prenatal care by Hispanic pregnant women. *Journal of the American Academy of Nurse Practitioners*, 14, 93-96.
- Showstack, J. A., Budetti, P. P., & Minkler, D. (1984). Factors associated with birthweight: An exploration of the roles of prenatal care and length of gestation. *American Journal of Public Health*, 74, 1003-1008.
- Smith, J. E. (2004). Exploring the efficacy of continuing education mandates. *JONA's Healthcare Law, Ethics, and Regulation, 6*(1), 22-31.
- Smith, L. S. (1998). Evaluation of an educational intervention to increase cultural competence among registered nurses (Doctoral dissertation). Retrieved from ProQuest. (9915541)
- Sodowsky, G. R., Taffe, R. C., Gutkin, T. B., & Wise, S. L. (1994). Development of a multicultural counseling inventory: A self-report measure of multicultural competencies. *Journal of Counseling Psychology*, 41, 137-148.
- Song, L., Singleton, E. S., Hill, J. R., & Koh, M. H. (2004). Improving online learning: Student perceptions of useful and challenging characteristics. *The Internet and Higher Education*, 7, 59-70. doi:10.1016/j.iheduc.2003.11.003
- Southernwood, J. (2008). Distance learning: The future of continuing professional development. *Community Practitioner*, 81(10), 21-23.
- Spice, R., Palacios, M., Biondo, P. D., & Hagen, N. A. (2011). Design and implementation of an online course on research methods in palliative care: Lessons learned. *Journal of Palliative Medicine*, 14, 413-419. doi:10.1089/jpm.2010.0374
- SRA International. (2009). CCNM Two-Year Evaluation Report: March 16, 2007 to March 16, 2009. Retrieved from https://www.thinkculturalhealth.hhs.gov/ccnm/ CCNM_EvaluationReport.asp
- Stanford-Bowers, D. E. (2008). Persistence in online classes: A study of perceptions among community college stakeholders. *Journal of Online Teaching and Learning*, 4(1), 37-50.
- Stevens, J. P. (2002). *Applied multivariate statistics for the social sciences* (4th ed.). Mahwah, NJ: Lawrence Erlbaum Associates, Inc.
- Sweeney, N. M., Saarmann, L., Flagg, J., & Seidman, R. (2008). The keys to successful online continuing education programs for nurses. *The Journal of Continuing Education in Nursing*, 39, 34-41.

- Tandon, S. D., Parillo, K. M., & Keefer, M. (2005). Hispanic women's perceptions of patient-centeredness during prenatal care: A mixed-method study. *BIRTH*, 32, 312-317.
- Transcultural C.A.R.E. Associates (2007, November). *Reported reliability & validity of the IAPCC-R*. Retrieved from http://www.transculturalcare.net/iapcc-r.htm
- Trinidad, S., Aldridge, J., & Fraser, B. (2005). Development, validation, and use of the Online Learning Environment Survey. *Australasian Journal of Educational Technology*, 21(1), 60-81.
- Vandeveer, M. (2009). From teaching to learning: Theoretical Foundations. In D. M. Billings & J. A. Halstead (Eds.), *Teaching in nursing: A guide for faculty* (3rd ed., pp. 189-226). St. Louis, MO: Saunders.
- Walker, S. L., & Fraser, B. J. (2005). Development and validation of an instrument for assessing distance education learning environments in higher education: The Distance Education Learning Environments Survey (DELES). *Learning Environments Research*, 8, 289-308. doi:10.1007/s10984-005-1568-3
- Weasenforth, D., Biesenbach-Lucas, S., & Meloni, C. (2002). Realizing constructivist objectives through collaborative technologies: Threaded discussions. *Language Learning and Technology*, 6(3), 58-86.
- Welch, A. Z. (2002). Madeleine Leininger: Culture care: Diversity and universality theory. In A. M. Tomey & M. R. Alligood (Eds.), *Nursing theorists and their* work (5th ed., pp. 501-526). St. Louis, MO: Mosby.
- Wilkinson, A., Forbes, A., Bloomfield, J., & Gee, C. F. (2004). An exploration of four web-based open and flexible learning modules in post-registration nurse education. *International Journal of Nursing Studies*, 41, 411-424. doi:10.1016 /j.ijnurstu.2003.11.001
- Williams, C. C. (2002). Evaluation of an educational intervention to increase cultural competence in social workers (Doctoral dissertation). Retrieved from ProQuest. (AAT NQ69254)
- Yang, Y., & Cornelious, L. F. (2005). Preparing instructors for quality online instruction. Online Journal of Distance Learning Administration, 8(1). Retrieved from http://www.westga.edu/~distance/ojdla/spring81/yang81.htm
- Zhang, D., Zhao, J. L., Zhou, L., & Nunamaker, J. F. (2004). Can e-learning replace classroom learning? *Communications of the Association for Computing Machinery*, 47(5), 75-79.

Zhao, Y., Lei, J., Lai, Y.B.C., & Tan, H. S. (2005). What makes the difference? A practical analysis of research on the effectiveness of distance education. *Teachers College Record*, 107, 1836-1884.

VITA

Graduate College University of Nevada, Las Vegas

Ella T. Heitzler

Degrees:

Bachelor of Science in Nursing, 2002 University of Virginia

Master of Science in Nursing, 2004 Virginia Commonwealth University

Post-Master's Certificate of Science in Nursing, 2005 Virginia Commonwealth University

Special Honors and Awards:

Yaffa Dahan Dissertation Award, University of Nevada, Las Vegas Sigma Theta Tau International Honor Society for Nursing Inductee

Dissertation Title: Enhancing the Cultural Competence of Women's Health Nurses via Online Continuing Education

Dissertation Examination Committee: Chairperson, Lori Candela, Ed.D. Committee Member, Cheryl Bowles, Ed.D. Committee Member, Patricia Alpert, Dr.PH. Graduate Faculty Representative, LeAnn Putney, Ph.D.